SEARCHING FOR SELF-DUALITY IN NON-MAXIMALLY SUPERSYMMETRIC BACKGROUNDS



A thesis submitted for the degree of Doctor of Philosophy in Applied Mathematics, Department of Mathematics and Applied Mathematics, University of Cape Town.

 $21 \ \mathrm{March} \ 2017$

The copyright of this thesis vests in the author. No quotation from it or information derived from it is to be published without full acknowledgement of the source. The thesis is to be used for private study or non-commercial research purposes only.

Published by the University of Cape Town (UCT) in terms of the non-exclusive license granted to UCT by the author.

Dedicated to my husband, for never tiring; always inspiring me and for sharing an unending enthusiasm for the progression of scientific knowledge and the philosophy of science. Thank you for your support. Thank you for teaching me.

Acknowledgements

I want to thank Associate Professor Jeff Murugan for accepting me as a doctoral student at the Laboratory for Quantum Strings and Gravity, his advice and knowledge have taken me a long way. Many thanks to Dr Jonathan Shock for his kindness, endless support and for answering my many questions. Thanks also to Michael Abbott for mentoring me. Additionally, I would like D Sorokin, S Penati, Per Sundin, Martin Wolf, Linus Wulff, and A Pittelli for accepting me as part of their collaboration. Despite many challenges, academic and personal, my supervisors offered assistance and understanding. I managed to learn a vast amount during this process. Their guidance really paved the way to completing this thesis.

I am very grateful for all the stimulating discussions held with the other students and post doctoral fellows at the Laboratory for Quantum Strings and Gravity. These colleagues answered many questions, often giving more than one explanation, for which I am thankful. Particularly, I want to thank Dr Emmanuel Malek for clarifying many points.

Also within the Department of Mathematics and Applied Mathematics, I would like to thank the wonderful ladies in room 319. These woman have been incredibly helpful throughout the three years. They are efficient, always able to answer any questions and have guided me through many administrative battles.

I would like to thank Professor Robert de Mello Koch who greatly inspired me early on in my career. He really did influence my choice to join the string community as a result of his outstanding enthusiasm and passion. These are qualities that have always encouraged me. He has taught me more than I ever imagined possible, with vigour and graceful explanations. This has not only taught me to be careful with my explanations, but to teach with passion and understanding.

Finally, and most importantly, I want to thank the Science Faculty at the University of Cape Town for offering me one of their generous PhD Fellowships. The Science Faculty PhD Fellowship programme started in 2014, the year I began my PhD, and has grown enormously since. This is a great project which will support many students over the years to come.

Abstract

Fermionic T-duality is the generalisation to superspace of bosonic T-duality (i.e. to include fermionic degrees of freedom). Originally, T-duality described the equivalence relation between two physical theories, each living on a different background. However, this thesis is concerned with fermionic T-duality and its role in self-duality. The goal is to determine whether AdS backgrounds with less than maximal supersymmetry are self-dual. A background is said to be self-dual if, after a specific sequence of bosonic and fermionic T-duality transformations, the original background is recovered. Self-dual backgrounds are of great interest due to their link to integrability. Fermionic T-duality has played a pivotal role in proving that the maximally supersymmetric background $AdS_5 \times S^5$ is self-dual. This background is also known to be integrable, therefore, when it was shown to be self-dual, the hypothesis that self-duality implied integrability, and vice-versa, was born. We investigate how far this hypothesis may be stretched for a number of AdS backgrounds, for which integrability has already been determined. The following backgrounds were considered: $AdS_2 \times S^2 \times T^6$ and $AdS_d \times S^d \times T^{(10-3d)}$ (d = 2, 3). This question of self-duality was approached in two ways. In the first approach we show that these less supersymmetric backgrounds are self-dual by working with the supergravity fields and using the fermionic Buscher procedure derived by Berkovits and Maldacena. In the second approach, we verify the self-duality of Green-Schwarz supercoset σ -models on $AdS_d \times S^d$ (d = 2, 3) backgrounds. Furthermore, we prove the self-duality of $AdS_5 \times S^5$ without gauge fixing κ -symmetry. We show that self-duality is a property which holds for the exceptional backgrounds, where the need to T-dualise along one of the spheres arises, again. Nature is not supersymmetric, therefore learning how to do physics in $AdS_5 \times S^5$ is not enough. In order to understand theories like Quantum Chromodynamics, we need to systematically break the supersymmetry present in our toy models. In this regard, it is easy to appreciate the significance of studying backgrounds with less than maximal supersymmetry.

Declaration

The work presented in this thesis is partly based on research in collaboration with Associate Professor Jeff Murugan and Dr Michael Abbott at The Laboratory for Quantum Gravity & Strings, University of Cape Town. Our other important collaborators include: Associate Professor Silvia Penati (University of Milan-Bicocca), Antonio Pittelli (University of Surrey), Professor Dmitri Sorokin (University of Padova), Dr Per Sundin (University of Milan-Bicocca), Professor Martin Wolf (University of Surrey) and Dr Linus Wulff (Imperial College). The list below identifies sections which are partly based on the listed publications:

- **Chapter 3:** M. C. Abbott, <u>J. Tarrant</u> & J. Murugan, Fermionic T-Duality of $AdS_n \times S^n(\times S^n) \times T^m$ using IIA Supergravity, Class. Quant. Grav. **33** (2016), 075008, arXiv:1509.07872 [hep-th]
- **Chapter 4:** M. C. Abbott, J. Murugan, S. Penati, A. Pittelli, D. Sorokin, P. Sundin, <u>J. Tarrant</u>, M. Wolf & L. Wulff, *T-Duality of Green-Schwarz Superstrings on* $AdS_d \times S^d \times M^{(10-2d)}$, JHEP **1512** (2015), 104, arXiv:1509.07678 [hep-th]

I hereby declare that this thesis has not been submitted, either in the same or different form, to this or any other university for a degree and that it represents my own work.

Signed by candidate

21 March 2017

Author

Contents

Ι	Introduction and Overview		
1	Introductory Bosonic T-duality		
	1.1	Introduction	8
	1.2	Abelian T-Duality	9 12 14
	1.3	Non-Abelian T-duality	17 19
	1.4	Summary and Conclusion	21
2	Intr	oductory Fermionic T-Duality	23
	2.1	Introduction	23
	2.2	The Fermionic Buscher Procedure	25
	2.3	A Recipe for Fermionic T-duality	29
	2.4	Self-duality of the $AdS_5 \times S^5$ Background	30
	2.5	Summary and Conclusion	31
II	Su	pergravity	33
3	Feri	mionic T-duality of $AdS_d \times S^d(\times S^d) \times M$	34
	3.1	Introduction	34
	3.2	Type IIA Supergravity	35
	3.3	$AdS_2 \times M$ Backgrounds $3.3.1$ The case $\alpha = 1$: $AdS_2 \times S^2 \times T^6$ $3.3.2$ $AdS_2 \times S^2 \times T^6$ with other RR fluxes $3.3.3$ The case $\alpha < 1$: $AdS_2 \times S^2 \times T^4$	37 40 42 43

	3.4	$AdS_3 \times M$ Backgrounds	47
	3.5	Combined Bosonic and Fermionic T-duality: In General	$52 \\ 52$
		3.5.2 Compensating Bosonic T-duality	57
	3.6	Summary	58
II	I G	Freen-Schwarz σ -model	59
4	\mathbf{Sup}	ercoset Models	63
	4.1	Introduction	63
	4.2	The Coset Geometry of S^3	63
	4.3	The General Setup	68 60
		4.3.1 Matter-Cartan Forms and \mathbb{Z}_4 -graded Coset Superspaces	09 72
		4.3.3 Schematic Form of the Superconformal Algebra	72
		4.3.4 Coset Representative and Associated Current	73
		4.3.5 The T-duality Procedure	75
	4.4	Example: $AdS_5 \times S^5$ Self-duality	76
		4.4.1 Supercoset Action	77
		4.4.2 T-duality Transformations	80
	4.5	Summary	84
5	Less	s Than Maximally Supersymmetric Coset Models	85
	5.1	Introduction	85
	5.2	Self-duality of $AdS_3 \times S^3 \times T^4$ Superstrings	86
		5.2.1 Self-duality of the Supercoset Model	88
		5.2.2 Non-supercoset Fermions	89
	5.3	Self-duality of $AdS_d \times S^d \times S^d \times T^{10-3d}$ Superstrings	90
		5.3.1 The Self-duality of $AdS_2 \times S^2 \times S^2 \dots \dots$	90
		5.3.2 Self-duality for $AdS_3 \times S^3 \times S^3$	95
	5.4	Combined bosonic-fermionic T-duality of the Ramond-Ramond $AdS_d \times S^d \times I_{0,0}$	
		M^{10-2a} backgrounds	96
		5.4.1 Rules for Fermionic T-duality	96
		5.4.2 Compensating Bosonic T-duality	99

	5.5	Summary	101
IV	⁷ S	ummary and Conclusion	102
v	A	ppendices	106
Α	A Conventions		
	A.1	Pauli Matrices	107
	A.2	Gamma Matrices	107
	A.3	The Killing Spinor Equations	108 108 109
	A.4	Hodge Duality	109
в	D(2,	$(1; \alpha)$ Supercoset Currents	111

List of Figures

1	The web of string dualities [8] where Type I string theory is related to type II		
	string theory via a bosonic T-duality	3	
3.1	A summary of the T-dualisation process for $AdS_2 \times S^2 \times T^6$	44	
3.2	A summary of the T-dualisation process for $AdS_2 \times S^2 \times S^2 \times T^4$	48	
3.3	A summary of the T-dualisation process for $AdS_3 \times S^3 \times S^3 \times T^1$	52	
3.4	Different superstring formulations require curved backgrounds to be on-shell [108].	61	
5.1	The idea of self-duality we study is that a sequence of bosonic and fermionic T-		
	dualities returns us to the same background. This is shown here for the case in		
	which we start with type IIB $AdS_2 \times S^2 \times T^6$ supported by an $F^{(5)}$ Ramond-		
	Ramond flux [36]	87	
5.2	Triangle of the relationships between various concepts that we have studied	103	

List of Tables

3.2	Table displaying the Γ	corresponding to each	background and their various fluxes.	54
-----	-------------------------------	-----------------------	--------------------------------------	----

- 4.2 The cosets corresponding to the various backgrounds for $0 \le \alpha < 1$. Here n_b and n_f are the number of bosonic and fermionic coordinates, respectively. 69
- 4.4 Symmetries and their number of associated generators, $N. \ldots \ldots \ldots \ldots 73$

Part I

Introduction and Overview

Introduction

String theory is the leading candidate for a theory which unifies gravity and quantum mechanics, the research efforts in this exciting field of physics have been remarkable. String theory has undergone enormous developments to arrive at the current paradigm - where symmetries play an even larger role in our understanding of string theory [1]. As a result of this intensive development, the field has exploded in a number of directions. Consequently, it has become virtually impossible to keep up to date with all the streams of research being explored. This is both very exciting and hugely daunting, especially for newcomers. That so many young scientists still enter this field, despite the steep learning curve, speaks volumes about the enticing problems that still need to be considered and solved.

During the mid-nineties string dualities suddenly formed the basis of serious study. Finally, there was a general understanding of the importance of these types of symmetries. Duality symmetries do not manifest themselves perturbatively in the weak coupling expansion of string theory, the context in which string theory has mostly been probed, instead they give us information about the exact string theory which is more complicated to study [1]. As a major driving force in string theory, the uncovering of non-perturbative duality symmetries, asks whether it is possible to study non-perturbative physics using perturbative techniques - a core tenet of the new paradigm. Hence, it is not too surprising that string dualities remained hidden for so long. After gaining access to strong coupling aspects through the use of mathematical tools associated with string duality, it became clear that the relationship between weak and strong coupling would be important. What could strong coupling teach us about weak coupling? Interest turned towards the idea of weak/strong duality. In string theory this means that as the string coupling q_s becomes large we are able to find new dual degrees of freedom whose fluctuations become small. That is, they are described by a new coupling $g_s' \sim 1/g_s$. String theory includes gauge theory, hence we need to consider a further point: weak/strong duality in the gauge theory is a necessary (but not sufficient) condition to ensure that weak/strong duality exists in the string theory [1]. Progress was made when Hull, Townsend and independently Witten [2–4] proposed an almost complete set of duals to all string theories possessing at least $\mathcal{N} = 4$ supersymmetry. Suddenly, the idea of duality had been elevated to a general principle which could be applied to all string theories in any number of dimensions where previously, the idea of duality was only an isolated set of conjectures. These results led to the increasing prominence of weak/strong coupling duality.

String dualities now play a central role in modern string theory research [2,4–6]. Since the term duality is used in many contexts throughout various fields in physics, it is necessary to clarify what we mean by it here. A duality is a map, which is often invertible¹, between two theories which sends states in one theory to states in the other theory. At the same time, interactions, amplitudes, and global symmetries are usually preserved [7]. The two theories that are found to be dual may be thought of as being physically equivalent. Dualities are particularly important as they are used to map difficult problems into simpler ones, which we may have more chance of solving. Initially, five consistent superstring theories in 10-dimensions were known. These were the type I, two type II, and two heterotic string theories. Additionally, there is also an 11-dimensional theory known as M-theory. The different string theories are distinguished from each other by the number (and kind) of supersymmetries and also by the various worldsheet topologies permitted (i.e. oriented versus unoriented). Although the five string theories appear to be very different when dealt with in the weak-coupling regime, we now know that they are all related to each other by one of the known string dualities. The connection between the different theories is known as the 'web of string dualities', as seen in Figure 1.



Figure 1: The web of string dualities [8] where Type I string theory is related to type II string theory via a bosonic T-duality.

In this thesis, our focus is on one particular string duality called T-duality and its many facets. In general, it relates two theories with different spacetime geometries. The two theories are considered to be equivalent in the sense that all observable quantities in one description are identified with observable quantities in the dual description. T-duality differs from other dualities in that it does not mention the exchange between weak and strong coupling of the two

¹As is the case for Abelian T-duality, however, this is not true for non-Abelian T-duality

dual theories. However, the AdS/CFT correspondence which was discovered two decades ago, is a duality which precisely deals with the weak/strong coupling concept described earlier. The AdS/CFT correspondence (or gauge/gravity du- ality) relates a quantum field theory and gravity. For example, this correspondence is well understood in the case of the equivalence between $\mathcal{N} = 4$ super Yang-Mills and type IIB strings on $AdS_5 \times S^5$ [6]. Put another way, the correspondence relates the quantum physics of strongly correlated many-body systems to the classical dynamics of gravity in one dimension higher [9].

In this context, type IIA and type IIB string theories are related by bosonic T-duality, the two being exchanged with each other under this duality transformation. The same is true for the two heterotic string theories, $E_8 \times E_8$ and SO(32). A central idea in string theory, concerning dualities, is that the strongly coupled limit of *any* string theory is equivalent to the weakly coupled limit of some other string theory [1]. All the connections between various string theories embody this principle. It must be noted that Figure 1 is an oversimplification because the different limits are characterized by more than just different string theories, but also by the different topologies of the various compact dimensions [1].

T-duality is one of the oldest and simplest examples of a string duality [10], being a manifestation of the extended nature of strings. T-duality takes a non-linear σ -model (which may be described by the Polyakov action) with a target space possessing an isometry and, through the application of the Buscher procedure [11–13], returns a T-dual σ -model, after gauging the isometry and integrating out the gauge fields. This worldsheet approach² to deriving T-duality in string theory, via the Buscher procedure, is now standard practice. T-duality was discovered as a symmetry of the effective potential in the compactification associated with radial inversion symmetry: $R \leftrightarrow \alpha'/R$, where R is the radius of compactification on the spacetime [17, 18]. The Buscher procedure was initially applied to Abelian isometries, but further studies made it possible to extend the application to non-Abelian isometries and fermionic isometries [19–22,84].

Indeed it is the fermionic generalisation that we are concerned with in this thesis. Motivation for fermionic T-duality arose when hidden symmetries, not manifest in the Lagrangian, were discovered in the scattering amplitudes of supersymmetric theories [23]. Soon after this, a remarkable connection between planar ³ scattering amplitudes and Wilson loops in $\mathcal{N} = 4$ super Yang-Mills was found [26–28], see [29] for a review. The most widely studied example of the AdS/CFT correspondence [6] is 4-dimensional $\mathcal{N} = 4$ super Yang-Mills and type IIB superstring theory on $AdS_5 \times S^5$. The planar limit of super Yang-Mills theory corresponds to the classical limit in string theory. The scattering amplitudes are described in terms of scalar one-loop integrals where, in the planar limit, contributing integrals reveal an interesting property. That is, when they are exchanged for their dual graphs they appear to exhibit a

²Originally worldsheet techniques [14, 15] had dominated the scene. Currently the trend is toward the spacetime approach taking over [16].

³The planar limit [24], for which the t'Hooft coupling $\lambda = g_{YM}^2 N$ is held fixed, is the case where only planar diagrams survive [25]. This occurs because the genus expansion in terms of Feynman diagrams corresponds to the 1/N expansion.

novel, non-trivial conformal symmetry, called dual conformal symmetry [30,31]. Originally, this symmetry showed up in the perturbative computations explored in [32], though soon afterwards dual conformal symmetry was found in next to next to maximally-felicity-violating (next to MHV) amplitudes [31], where it was then extended to full dual superconformal symmetry.

Ordinary conformal symmetry in the T-dual model is the dual conformal symmetry in the original model. To make sense of their relationship we note that dual conformal symmetry of scattering amplitudes in the gauge theory acts on momenta in much the same way as ordinary conformal symmetry acts on coordinates in the dual model. Simultaneously, it associates to each scattering amplitude a string worldsheet in a dual AdS space [33–35]. Berkovits and Maldacena showed that dual superconformal symmetry may be understood using a T-duality symmetry of the full superstring theory on $AdS_5 \times S^5$. The T-duality involves ordinary bosonic T-dualities (considered in [26]) and a novel set of fermionic T-dualities. Application of these symmetries returned a background equivalent to $AdS_5 \times S^5$. This resultant background was termed *self-dual*.

Fermionic T-duality generates new solutions via the identification of commuting fermionic isometry directions built from Killing spinors [22]. We may attribute to the existence of dual superconformal symmetry, the self-duality of the superstring σ -model under a sequence of T-duality transformations [36]. These transformations are of the bosonic and fermionic string modes on the worldsheet corresponding to some commuting isometries of $AdS_5 \times S^5$, see [21,22] for details. Simultaneously, self-duality arises as a result of the combined bosonic and fermionic T-dualities not changing the form or values of the background fields on $AdS_5 \times S^5$, such as the dilaton and the Ramond-Ramond flux [23]. The dilaton and metric tensor make up the NSNS (Neveau-Schwarz-Neveau-Schwarz) sector in the RNS (Ramond-Neveau-Schwarz) formalism of type II supergravity. The dilaton is a scalar field related to the string coupling g_s . The RR sector is made up of the Ramond-Ramond fields, which are differential forms, that act as the higher dimen- sional generalizations of the fields in Maxwells electromagnetism. Fermionic T-duality, dual superconformal symmetry and their relationship to each other have been broadly studied in the context of the maximally supersymmetric $AdS_5 \times S^5$ background [21–23]. Therefore, this example is well understood. On the other hand, the fermionic T-duality of σ -models for superstrings on integrable⁴ AdS backgrounds possessing less than maximal supersymmetry is much less understood. Therefore, we are concerned with extending the understanding of this idea to example possessing less supersymmetry.

The readers first question may well be: Why would one study the less than maximally supersymmetric backgrounds? Furthermore, what could we learn? What we know is that the AdS/CFTcorrespondence for AdS_5/CFT_4 with $AdS_5 \times S^5$ has been studied in great detail. Much has been learned as a result and shortly after AdS/CFT was discovered ideas started drifting toward unfamiliar territory with regards to this duality [42]. This created a platform for testing the generality of this idea. To test generality, a good question is: If we know the results for

⁴As mentioned in the footnote found in [36], the classical integrability of superstrings in AdS backgrounds has been studied in [37–41].

maximal supersymmetry, then what can we infer for less supersymmetry? Do we have reason to believe that what we have observed so far, will hold for less than maximally supersymmetric cases? Following this logic, it makes sense to choose a background with less supersymmetry, but perhaps with as little supersymmetry broken as possible. Such a next-to-maximally supersymmetric background would be $AdS_4 \times \mathbb{C}P^3$ associated with N = 6 Cherns-Simons (ABJM).

As for the case of $\mathcal{N} = 4$ super Yang-mills, integrability features in ABJM in the planar limit [42–44] and in string theory on $AdS_4 \times \mathbb{C}P^3$ [38, 45–47]. There is also an overwhelming amount of evidence provided by perturbative calculations, that amplitude/Wilson loop duality and dual superconformal symmetry are present for the AdS_4/CFT_3 partnership. It is remarkable that such similarity between AdS_5/CFT_4 and AdS_4/CFT_3 cases exist and it is for this reason that string theorists were led to hope that a self-dual mapping of the geometry $AdS_4 \times \mathbb{C}P^3$ under bosonic and fermionic T-duality would account for the observed perturbative symmetries of the ABJM theory. We have hope because $AdS_5 \times S^5$ is self-dual under these exact T-duality transformations. Unfortunately, all attempts to prove self-duality under bosonic and fermionic T-duality have failed. Failed, despite the important ingredients like integrability remaining intact. We would like to work out part of the puzzle that is: Why this failure? To this end we test the self-duality of backgrounds which are integrable (like $AdS_4 \times \mathbb{C}P^3$ and $AdS_5 \times S^5$) but possess less supersymmetry (like $AdS_4 \times \mathbb{C}P^3$).

Whether or not self-duality is present in the backgrounds considered herein, we will be one step closer to understanding the issues surrounding $AdS_4 \times \mathbb{C}P^3$. Furthermore, we know that for $AdS_5 \times S^5$, T-self-duality appears to be a predictor of integrability. This is a brilliantly useful relationship as self-duality is much simpler to demonstrate than integrability. Therein, we find more motivation to study $AdS_4 \times \mathbb{C}P^3$. Will this predictor of integrability be completely ruled out? An interesting question indeed! Therefore, we take it upon ourselves to see how far the relation (self-duality \Longrightarrow integrability) extends by considering backgrounds with less than maximal supersymmetry. In summary: Does less supersymmetry account for the failure of $AdS_4 \times \mathbb{C}P^3$ being self-dual?

Overview of Thesis

This thesis is divided into three Parts and further into Chapters. **Part I: Introduction and Overview** contains the general introduction to the whole thesis, as well as Chapter 1 and Chapter 2 which contain the essential mathematical framework required to understand the work which follows. **Part II: Fermionic T-duality of Non-maximally Supersymmetric Backgrounds** contains the body of original work created as part of two collaborations. The first team is based at the The Laboratory for Quantum Gravity & Strings at the University of Cape Town.

Chapter 1 provides an introduction to bosonic T-duality. We discuss some historical features to provide context and then go on to derive the Buscher rules. Two cases are examined: the original Abelian T-duality and its generalisation to non-Abelian isometries. We also consider generalised geometry and the O(D, D) group which encompasses Abelian T-duality transformations.

Chapter 2 introduces fermionic T-duality, the fermionic generalisation of ordinary T-duality. It is this idea which forms the basis of this thesis. We discuss its properties, its differences compared to ordinary T-duality and then go on to derive the Buscher procedure for fermionic isometries. A general prescription for carrying out this procedure is also given.

Chapter 3 considers the self-duality of the backgrounds $AdS_2 \times S^2 \times T^6$ and $AdS_d \times S^d_+ \times S^d_- \times T^{10-3d}$ for d = 2, 3. We use the Buscher procedure and work at the level of the supergravity fields to show self-duality. This approach is very useful for seeing how the Ramond-Ramond fields transform explicitly. After studying particular cases we end with a general argument which pulls the ideas in this chapter together. We start in type IIA string theory, although we can easily arrive at a type IIB theory through a bosonic T-duality.

Chapter 4 deals with supercoset σ -models. We verify the self-duality of such models on $AdS_d \times S^d$ backgrounds for d = 5 under combined bosonic and fermionic T-dualities without fixing κ -symmetry gauge.

Chapter 5 We prove that the exceptional backgrounds $AdS_d \times S^d_+ \times S^d_-$ for d = 2, 3 are self-dual. This approach is to be contrasted with the methods in Chapter 3.

Chapter 1

Introductory Bosonic T-duality

1.1 Introduction

The string theories that we are interested in studying are described by non-linear σ -models. These tell us how to embed a 2-dimensional manifold into a target spacetime, the goal being to embed the manifold into a spacetime of physical interest. This 2-dimensional manifold is called the worldsheet which is the natural generalization of the worldline for a particle. The worldsheet is the minimal surface interpolating between the given initial and final configurations of the string. The 1-dimensional extended nature of the string has some interesting consequences. In particular, we are interested in the unique duality symmetry called target space duality, or just T-duality¹. Duality symmetries provide a one-to-one correspondence (or map) between two different systems. This map furnishes a dictionary which allows us to translate between the two theories in a precise way. Often, through this dictionary we are able to map difficult problems into easy ones, and vice versa². This chapter deals with bosonic T-duality which is the original form of this duality symmetry. Interest in bosonic T-duality has been revived due to the discovery of its cousin, fermionic T-duality [48], to be explored in the following chapter. The bosonic Neveu-Schwarz (NS) sector is the massless sector common to all types of string theory. It contains a symmetric 2-form tensor field G called the metric, an antisymmetric 2-form tensor field B called the Kalb-Ramond field and the scalar field Φ called the dilaton. The B field is a higher rank tensor analogue of a gauge field. The dilaton is related to the string coupling g_s as follows

 $g_s = \exp(\langle \Phi \rangle)$

¹The 'T' also stands for toroidal. A compactification is toroidal if the compact space is a torus.

²Note that symmetries always operate within the same system, leading to conservation laws which simplify problems greatly, but they are quite different from dualities in that they do not map difficult problems into easy ones. Instead, symmetries simplify difficult problems by applying constraints to the possible forms of the answer.

where $\langle \Phi \rangle$ is the vacuum expectation value of Φ . The coupling is a dynamical variable in string theory. T-duality, for closed strings, relates small distance scales to large distance scales. It is also crucial to the understanding of D-branes, which arise in open string theory [49]. An important point to mention is that T-duality maps $R \to 1/R$ and consequently it is the winding of strings that is crucial ³. Point particles are 0-dimensional and thus cannot possess winding modes. Thus it is this extended nature of strings which exposes new properties such as T-duality. This chapter introduces bosonic T-duality in the Abelian and non-Abelian contexts. We place emphasis on closed string theory throughout this thesis.

1.2 Abelian T-Duality

Begin by considering a closed bosonic string ⁴ which moves in the background $S^1 \times R^{1,24}$. This means that the spacetime is compactified along one direction, i.e. the S^1 . This circle has a radius R. The coordinates X describe the embedding of the worldsheet into spacetime, that is, our coordinates are parametrised by the worldsheet coordinates (τ, σ) , much like the worldline is parametrized by proper time. Specifically, the circle compactification amounts to singling out one direction, say $X^1(\tau, \sigma)$, which is compact. Here τ is the time on the worldsheet and σ is the circular, periodic direction along the closed string. This compactification changes the string dynamics in two important ways [51]. Firstly, the spatial momentum, p^1 , may no longer assume any value. It is now quantized:

$$p^1 = \frac{n}{R}, \quad n \in \mathbb{Z}$$

Secondly, the boundary conditions for the mode expansion of X become more general after compactification. As a result, we are able to move around the string whilst relaxing the condition $X(\tau, \sigma + 2\pi) = X(\tau, \sigma)$ to $X^1(\tau, \sigma + 2\pi) = X^1(\tau, \sigma) + 2\pi mR$, where $m \in \mathbb{Z}$ and we have an isometry along the X^1 -th direction. Here the integer m tells you how many times the string winds around the compact direction, S^1 . Naturally, m is called the winding number, that is, the number of times X^1 (the compact direction) needs to be circled in order to return to the starting point. This winding term is a uniquely stringy phenomenon. To find the energy spectrum, consider the periodic field X^1 , which has the following mode expansion

$$X^{1}(\tau,\sigma) = x^{1} + \frac{\alpha' n}{R}\tau + mR\sigma + \text{oscillator modes},$$

incorporating both quantized momentum and the possibility of winding number. The centre of mass for the system is given by $x^1 + \alpha Rn\tau$. Start with a simple non-linear σ - model,

³Here 1/R is the radius of the dual background in terms of the radius R of the original background.

⁴The worldsheet approach, in a general framework, has been studied for closed bosonic strings in [50].

 $S(G = \delta, B = 0)$, whose dynamics is governed by a wave equation. Here δ is the Kronecker-delta and we have a constant dilaton. The wave equation is much simpler to solve if the coordinates are split into left- and right-moving modes:

$$X^1(\sigma,\tau) = X^1_L(\sigma^+) + X^1_R(\sigma^-)$$

where

$$\begin{aligned} X_L^1(\sigma^+) &= \frac{1}{2}x^1 + \frac{1}{2}\alpha' p_L \sigma^+ + i\sqrt{\frac{\alpha'}{2}} \sum_{n \neq 0} \frac{1}{n} \tilde{\alpha}_n^1 \exp(-in\sigma^+) \\ X_R^1(\sigma^-) &= \frac{1}{2}x^1 + \frac{1}{2}\alpha' p_R \sigma^- + i\sqrt{\frac{\alpha'}{2}} \sum_{n \neq 0} \frac{1}{n} \alpha_n^1 \exp(-in\sigma^-) \end{aligned}$$

and α' is the square of the fundamental string scale, l_s . The Fourier modes $\tilde{\alpha}_n^1$ and α_n^1 act as creation and annihilation operators respectively. The momenta are given by

$$p_L = \frac{n}{R} + \frac{mR}{\alpha'}, \quad p_R = \frac{n}{R} - \frac{mR}{\alpha'}$$

with $\sigma^{\pm} = \tau \pm \sigma$. Finally, consider the mass spectrum, calculated using

$$M^2 = \sum_{\mu=0}^{25} p_{\mu} p^{\mu}$$

Then the mass squared of the (N, \tilde{N}) excited state is given by [51]

$$M^{2} = \frac{n^{2}}{R^{2}} + \frac{m^{2}R^{2}}{\alpha'^{2}} + \frac{2}{\alpha'}(N + \tilde{N} - 2), \qquad (1.1)$$

where N and \tilde{N} are the number operators for left-moving and right-moving oscillation modes of the string, respectively. The first term in (1.1) tells us that a string with n > 0 units of momentum around the compact direction, X^1 , contributes n/R to its mass. The second term says that a string winding m > 0 times around the compact direction picks up a contribution of $2\pi mRT$ to its mass, where

$$T = \frac{1}{2\pi\alpha'}$$

Notice that the spectrum (1.1) is invariant under the following transformation

$$m \leftrightarrow n, \quad R \leftrightarrow \frac{\alpha'}{R}.$$

Thus, the strings compactified on circles with radii R and αR are T-dual, and have identical mass spectra. This is called radial inversion symmetry. The precise mapping is given by

$$X_R^1 \leftrightarrow -X_R^1, \quad X_L^1 \leftrightarrow X_L^1.$$

This extends the symmetry of the spectrum to a symmetry of the whole theory by reflecting ⁵ the

 $^{^{5}}$ We actually mean parity reflection here. In non-flat backgrounds possessing an isometry, T-duality does not reduce to a parity transformation any longer, which acts on right-moving (or left-moving) worldsheet variables alone. Instead, in the general case, T-duality acts as a canonical transformation affecting both the left- and

right-moving co-ordinate and leaving the left-moving part invariant ⁶. The above procedure also applies to toroidal compactifications [56], specifically to Narain compact-ifications [57], leading to the T-duality group O(D, D, Z). We can think of this group as follows, consider the group O(2D) which has parity and rotations of 2D coordinates. Then the group O(D, D) results from still maintaining parity, but letting D coordinates in the metric acquire negative signs. Then the object preserved is the metric with an equal number of negative coordinates as positive. This describes a double space where all D coordinates are doubled, mimicking a string theory with all dimensions being compact ⁷.

Transformation of Ramond-Ramond Fields

Up till this point, we have been discussing 26-dimensional bosonic string theory. Now we concentrate on the type II 10-dimensional superstring theories. In addition to the NS sector, superstring theory also contains a *Ramond-Ramond* or RR sector, which contains fields which are differential forms. These fields exist in the 10*D* spacetime of type II supergravity which is the classical limit of type II string theory. The RR *p-forms* generalize Maxwell's electromagnetism. The flux transformations for the RR fields can be found in [52]. We start by constructing a *poly-form* composed of RR forms contracted with Γ -matrices ⁸. For a IIA theory we have even fluxes:

After T-dualising, the left and right moving modes couple to two distinct sets of vielbeins. However, as a result of these vielbeins describing precisely the same geometries, we may relate them by a Lorentz transformation, Λ . This Λ is used to define an action on spinors parametrized by the matrix Ω satisfying [58]

$$\Omega^{-1}\Gamma^a\Omega = \Lambda^a_b\Gamma^b.$$

For SU(2) isometries this is solved by

$$\Omega = \frac{\alpha'}{\sqrt{G_{99}}} \Gamma_{11} \Gamma_9,$$

right-movers on the worldsheet [52–55].

⁶This reflection applies to fermionic co-ordinates for the superstring [56] as well. Particularly, reflection of the right-moving fermion in the compact direction causes a change in chirality resulting in type IIA and type IIB superstring theories being interchanged.

⁷See the box titled "T-duality in Field Theory" for more information.

⁸See Appendix A for conventions.

with Γ^{11} being the 10-dimensional chirality operator. Then we obtain the dual fluxes through the inversion of Ω . They are related to the original fluxes as follows

$$\tilde{P} = P \cdot \Omega^{-1}.$$

1.2.1 The Buscher Rules

String theory backgrounds are described by both their metric and Kalb-Ramond field, therefore in what follows we refer to (G, B) as the *background*. The action of T-duality re-lates the backgrounds (G, B) and (g, b), which are completely different, using the Buscher rules [12, 13]. Backgrounds related in this way are said to describe the same physics. That is, S(G, B) and S(g, b) give rise to identical theories, furthermore they give rise to the same quantum theory [59]. This may be illustrated by constructing an intermediate action obtained by gauging the isometries ⁹. That is, we make the global symmetry a local symmetry through the introduction of a gauge field and the replacement of ordinary derivatives by covariant derivatives. The σ model is described by the Polyakov action for a bosonic string [60] in conformal gauge. Moving to conformal gauge involves selecting coordinates for which the metric becomes diagonal. It is given by

$$S = \int d^2 z [G_{mn}(X) + B_{mn}(X)] \partial X^m \bar{\partial} X^n$$

and written in terms of the complex worldsheet co-ordinate $z = \frac{1}{\sqrt{2}}(\tau + i\sigma)$. From the 2dimensional worldsheet perspective the fields X^m are bosons while G and B are field dependent couplings. From the spacetime perspective, X^m are the co-ordinates of spacetime parametrized by the worldsheet co-ordinates. We choose co-ordinates $\{X^1, X^i\}$ for i > 1 in such a way that the symmetry acts by shifting along the X^1 -direction only. The X^1 -direction is special because it is compact and it is the direction in which our isometry acts. Then we may write

$$S' = \int d^2 z [G_{11}A\bar{A} + L_{1i}A\bar{\partial}X^i + L_{i1}\partial X^i\bar{A} + L_{ij}\partial X^i\bar{\partial}X^j + \tilde{X}^1(\partial\bar{A} - \bar{\partial}A)]$$

where $L_{mn} = G_{mn} + B_{mn}$, and the background fields are all independent of X^1 . Note the replacement

$$(\partial X^1, \bar{\partial} X^1) \to (A, \bar{A})$$
 (1.4)

where (A, \overline{A}) forms the auxiliary ¹⁰ gauge field on the worldsheet. We may view this replacement as gauging the shift symmetry of the original σ -model by a minimal coupling to the gauge field A [10]:

$$\partial X^i \to DX^i = \partial X^i + A$$

⁹Isometries occur whenever the action S(G, B) is invariant under spacetime diffeomorphisms in a direction which has a global symmetry from the 2-dimensional worldsheet perspective.

 $^{^{10}}$ An auxiliary field does not contain any dynamics and therefore does not contribute to the degrees of freedom present in the theory. For ordinary fields, when we solve the equations of motion, the degrees of freedom are the labels which parametrize our solution. That is, they are the initial conditions. However, when solving the equations of motion for the auxiliary field A, we find algebraic equations, which are not integrated, and therefore there are no initial conditions which need to be specified.

for i > 1. The last term in S' contains the Lagrange multiplier \tilde{X}^1 , whose purpose is to impose the constraint F = dA = 0. The result is a gauged non-linear σ -model $S(g, b; A, \tilde{X}^1)$. To return to the original model, the Lagrange multiplier \tilde{X}^1 must be integrated out, which takes us back to the original theory S(G, B). The result being the reversal of the arrow in (1.4), and recovering the initial σ -model. However, integrating out A using the gauge field equations:

$$A = G_{11}^{-1} (\partial \tilde{X}^1 - L_i^1 \partial X^i),$$

$$\bar{A} = -G_{11}^{-1} (\bar{\partial} \tilde{X}^1 - L_i^1 \bar{\partial} X^i),$$

returns a different theory. These solutions tell you that A and \overline{A} are determined as soon as you specify the X's. It is the σ -model S(G, B) written in terms of new background fields ¹¹. This is the *dual* theory with action

$$S'' = \int d^2 z [\tilde{g}_{mn}(X) + \tilde{b}_{mn}(X)] \partial y^m \bar{\partial} y^n$$

written in terms of the co-ordinates $y^m = {\tilde{X}^1, X^i}$. The Lagrange multiplier represents the dual co-ordinate, the direction for which the dual theory is isometric. The dual fields are related to the original fields, as derived by Buscher

$$\tilde{g}_{11} = (G_{11})^{-1}, \quad \tilde{g}_{1i} = (G_{11})^{-1} B_{1i}, \quad \tilde{b}_{1i} = -(G_{11})^{-1} G_{1i}$$

$$\tilde{g}_{ij} = G_{ij} - (G_{11})^{-1} (G_{i1} G_{1j} + B_{i1} B_{1j})$$

$$\tilde{b}_{ij} = B_{ij} - (G_{11})^{-1} (G_{i1} B_{1j} + B_{i1} G_{1j}).$$
(1.5)

This method works in the case of abelian isometries ¹² [56]. Through recent developments, many derivations of the transformation rules for Ramond-Ramond fields [52,63–67] exist. We will first consider such rules in the context of T-duality described by an O(D, D) symmetry group.

¹¹This can also be done using a Hamiltonian approach through canonical transformations [54]. ¹²Global issues were first raised in [59] and are subsequently discussed in [61] and [62].

As we have seen, in the presence of a compact dimension, closed strings exhibit fundamentally new states due to their extended nature: that is, closed strings are able to wrap around the compact dimension. This winding makes it impossible to contract closed strings to a point. Since open string endpoints on a space-filling D25-brane are allowed to move freely, open strings can indeed always be contracted to a point. Compared to closed strings, open strings do not exhibit any fundamentally new states in the presence of a compact dimension. Furthermore, the string spectrum for an open string on a space-filling D25-brane with radius of compactification R and a related open string on a D25-brane with radius of compactification $\tilde{R} = R/\alpha'$, do not coincide. This implies that open strings on D25-branes are able to distinguish between the two different compactifications. To preserve T-duality in the presence of open strings, we find that T-duality relates a spacetime with radius R and a D25-brane to a spacetime with radius $\hat{R} = R/\alpha'$ and a D24-brane. In the dual space, X^1 is the Dirichlet direction of the D24-brane. We set $X^1 = 0$ to be the position of the brane along the compact direction. All open string endpoints are then held attached to points with $X^1 = 0$ in the dual direction. Because the string endpoints are kept fixed, new open string configurations exist that prevent open strings from being contracted away.

Therefore, an open string stretching from $X^1 = 0$ to $X^1 = 2\pi R$ winds around the compact direction once. Open strings may wind around the compact direction any number of times just like closed strings, moreover, open strings may even resemble closed strings, however they are not closed since the open string endpoints do not need to coincide.

1.2.2 Generalised Geometry

The fundamental theory of relativity tells us that physics is the same for all observers, that is, physics is co-ordinate invariant. This idea is best formulated mathematically using differential geometry. One considers a tangent bundle ¹⁴ to the given spacetime manifold, equipped with a metric tensor. It is the metric tensor which acts as the dynamical object in the theory, 'encoding' the physical content. The Riemannian metric of general relativity provides a measure of distance in this theory. But what about theories requiring a metric and an antisymmetric 2-form to define their background? The answer is, we expect that these two fields should be treated equally and therefore combined to form the dynamical object. This is done by developing an associated geometrical theory, using the O(D, D) structure group. This is the *generalized tangent bundle*

¹³This information box is an excerpt from [?], Chapter 18.

 $^{^{14}\}mathrm{A}$ tangent bundle of a differential manifold M is a manifold TM , which assembles all the tangent vectors in M .

described in the context of generalized geometry [69–72]. The dynamical object is now given by the generalized metric, M:

$$M = \begin{bmatrix} G - BG^{-1}B & -BG^{-1} \\ G^{-1}B & G^{-1} \end{bmatrix}.$$

The generalized dilaton, d, is

$$e^{-2d} = e^{-2\Phi} \sqrt{\det G}.$$

The generalized metric provides the most efficient description of the action of T-duality on the background. In this set-up, T-duality is generated by O(D, D) transformations with the following action

$$M \to M' = \mathcal{T}^{-1}M\mathcal{T}$$

with d remaining invariant. Then, T-duality is implemented using the following matrix [73]

T-duality in the
$$k^{th}$$
 direction : $\mathcal{T}_{T(k)} = \begin{bmatrix} \mathbf{1} - 1_k & 1_k \\ 1_k & \mathbf{1} - 1_k \end{bmatrix}$

where 1_k is the $D \times D$ -matrix with 1 as the k^{th} diagonal entry. This vector ensures we are dualising along the chosen direction, the k^{th} direction. Dualising along several co-ordinate directions generalizes naturally to $\mathcal{T}_{v1}\mathcal{T}_{v2} = \mathcal{T}_{v1+v2}$. The dual metric G' is then read off from the ' (G^{-1}) ' term which corresponds to the matrix in the 4th quadrant of M'. The change in the dilaton is given by

$$\Delta \Phi = \frac{1}{4} \log \frac{\det G'}{\det G}.$$

T-duality in Field Theory

T-duality relies on the extended nature of strings to wrap around compact dimensions. The existence of winding modes resulting from such wrappings, and also the momentum modes underlie T-duality. The result of the action of T-duality being the creation of a connection between the physics (of strings) defined on geometrically distinct backgrounds. Double Field Theory¹⁵ [77–79] tries to incorporate T-duality as a symmetry of field theory. At first the idea does seem bizarre, given that T-duality is a unique symmetry of strings, and that field theory describes 0-dimensional particles which are unable to wrap around compact dimensions. Therefore it is quite natural to assume that if we were to incorporate a T-duality symmetry in field theory, we should take into account information about the winding modes, somehow. To do this, we need to assign more degrees of freedom to our particles, accounting for their inability to reproduce stringy-like winding properties [74]. Double Field Theory incorporates these new degrees of freedom by doubling the space of co-ordinates. Thus ordinary and winding co-ordinates are considered to be the co-ordinates of a doubled manifold¹⁶.

¹⁵For a review, consult [74-76].

¹⁶For any manifold M with a boundary ∂M , we may define its double to be the manifold obtained by gluing two copies of M together along their common boundary [80]. This does not mean that the double is always closed. In fact, this concept makes sense for any manifold. However, when discussing doubles, one is primarily referring to manifolds M which are compact (and thus closed) with non-empty boundaries ∂M .

Effects on Ramond-Ramond Fields

Hassan [53], and later Fukuma et al [81] described the changes made to the RR fields due to the action of T-duality by expressing the fields as a set of fermionic (or grassmannian) creation and annihilation operators, ψ^m and ψ_m respectively. These are combined to arrive at the potentials $C^{(p)}$. They are then given by

$$|C\rangle = \sum_{p} \frac{1}{p!} C^{(p)}_{mn...p} \psi^{m} \psi^{n} ... \psi^{p} |0\rangle,$$

with corresponding algebra

$$\{\psi_m, \psi^n\} = \delta_m^n, \quad \{\psi^m, \psi^n\} = \{\psi_m, \psi_n\} = 0,$$

The annihilation operators ψ_m annihilate the vacuum $|0\rangle$. T-duality acting along the m^{th} direction may be expressed as having the following operator

$$T_m = \psi^m + \psi_m. \tag{1.6}$$

acting on the vacuum $|0\rangle$. Note that different T_m anti-commute, so the sign of the final $C^{(p)}$ after T-dualising depends on the ordering, although this has no physical effect. The following subtleties apply

1. When considering a non-trivial dilaton, we ought to apply the above rules to $e^{\Phi} dC^{(p)}$ instead [22,23]. This is also mentioned in the last paragraph of [81].

- 2. A T-duality along the time direction exchanges real and imaginary fluxes [83], therefore (1.6) applies for $m \neq 0$ and we define $T_0 = i(\psi^0 + \psi_0)$.
- 3. For the case when the *B*-field is non-zero, these operators apply to the modified potential D, introduced in [81], rather than the potential C.

We may express the effects on the RR-fields in another way, as Hassan did in [52]. Here we may act directly on the fields $F^{(p)}$. Then performing a duality in the x^9 -direction, for example, returns new fluxes

$$F_{9no...q}^{(p)'} = F_{no...q}^{(p-1)} - \frac{p-1}{G_{99}}G_{9[n} F_{9o...q]}^{(p-1)};$$

$$F_{mn...q}^{(p)'} = F_{9mn...q}^{(p+1)} - pB_{9[n} F_{o...q]}^{(p)'};$$

Since all our cases deal with diagonal G and a B which is zero, the second terms in each equation above vanish. This approach also requires one to remember to include a factor of i when dualising along time [83]. Appropriate factors also need to be inserted when the dilaton is non-trivial.

1.3 Non-Abelian T-duality

As we have already discussed, T-duality is an equivalence between string theories propa- gating on two different spacetime geometries which contain some isometries [19]. Simply put, it describes strings on a circle of radius R and those strings on a circle of radius α'/R as equivalent. In general, a map, given by the Buscher rules, is provided by the action of T-duality. This map relates one solution of supergravity to another. The Buscher procedure can naturally be generalized to include target spaces equipped with a non-Abelian isometry group H. For such a case, the Buscher procedure is followed almost exactly, except that the gauge fields ¹⁷ now take values in the algebra of H. This also produces a map from one solution of supergravity to another, as in the Abelian case. This acts as a solution generating technique for supergravity theories.

This natural generalisation was proposed in [84] (see [85] for earlier work). Although some attention was paid to this new duality in [86, 87], progress was impeded until it was worked out how to embed this duality symmetry in supergravity. This meant knowing the full transformation, including that of the RR fields, which was later found by Sfetsos and Thompson in [88]. Presently, we are at the stage where much more work and development has been done and progress has been made toward our understanding of non-Abelian T-duality [19,19,90–93], though there are still many things which remain a mystery. For example, non-Abelian T-duality might be extended to an exact duality of the full superstring theory, but at the moment this is something which remains un- clear. Although it appears that the dualisation procedure for Abelian T-dualities are very similar, we list two important differences:

1. The isometry of the initial target space geometry is destroyed by non-Abelian T- duality, even if only partially. Furthermore, the supersymmetry of the model may be potentially destroyed too ¹⁸.

2. Global issues arise as a result of applying the Buscher procedure to worldsheets of higher genera, that is, beyond tree level. Therefore, non-Abelian T-duality is not an exact (or full) symmetry of string perturbation theory, existing only as a tree-level symmetry.

Earlier work on non-Abelian T-duality for cases involving purely Neveu-Schwarz backgrounds can be found in [84,86,87,94–97]. After the publication of [88], there was a resurgence of interest in the subject of non-Abelian T-duality, leading to the extension of the Buscher procedure to geometries containing RR fields in addition to NSNS fields. Furthermore, non-Abelian Tduality was originally studied in the context of backgrounds possessing SU(2) isometries. Then, performing non-Abelian T-duality on the SU(2) isometry group within the sphere of type IIB

¹⁷The gauge field is a 1-form with values in the Lie algebra associated to the gauged isometry group [56].

¹⁸It is possible to recover this lost isometry as a non-local symmetry of the σ -model. The corresponding σ -models are canonically equivalent [84].

 $AdS_5 \times S^5$ causes something interesting to occur. The action of T-duality produces a geometry which is a solution of type IIA supergravity [88]. Consider the background containing an $SO(4) = SU(2)_L SU(2)_R$ isometry, then the round S^3 metric is written as

$$ds^{2} = d\theta^{2} + d\phi^{2} + 2\cos\theta d\phi d\psi + d\psi^{2}.$$

Next, perform the T-duality with respect to one of the SU(2) isometries, say the left ¹⁹, then we will find a geometry that interpolates between $\mathbb{R} \times S^2$ and \mathbb{R}^3 with the following metric

$$\hat{ds}^2 = dr^2 + \frac{r^2}{1+r^2}(d\theta^2 + \sin^2\theta d\phi^2).$$

However, there is also an NSNS antisymmetric 2-form field and scalar dilaton present for the dual geometry:

$$\hat{B} = \frac{r^3}{1+r^2} vol(S^2), \quad \hat{\Phi} = -\frac{1}{2}\ln(1+r^2).$$

We are also interested in how the RR fields transform. Suppose that the initial geometry considered above is supported by a RR 3-form

$$F_3 = vol(S^3).$$

Dual fluxes are extracted using the following equation

$$e^{\hat{\Phi}}\hat{F} = F.\Omega^{-1},$$

$$\Omega^{-1} = \frac{1}{\sqrt{1+r^2}} (-\Gamma^{123} + r\Gamma^r).$$

From this we may determine that the dual geometry contains a 0-form and a 2-form:

$$F_0 = 1$$
, $F_2 = \frac{r^3}{1 + r^2} vol(S^2)$.

where the 0-form, F_0 , is called the *Romans mass*²⁰ [19]. When this dual geometry is embedded into a type IIA supergravity background, a solution of massive IIA supergravity is found. A massive IIA supergravity results whenever the Romans mass is non-zero. Notice that the type of string theory has changed from type IIB to IIA. The fact that the isometry group being dualised has an odd dimension means that the supergravity will change in type. For even-dimensional isometry groups, the supergravity type remains invariant.

¹⁹Note that if, instead, we performed T-duality with respect to the right SU(2), the metric and background would remain unchanged. This is because we are dealing with type IIA string theory, which is parity invariant, and all that relates the two SU(2)'s is a parity transformation.

²⁰The Romans mass was originally introduced as a 'cosmological constant' by Romans in 1986. However it was later understood to be an RR flux to be treated as any of the other RR fluxes, all of which are mixed by T-duality.

1.3.1 Some Technical Details

As mentioned earlier, the most well-studied version of non-Abelian T-duality exists for backgrounds having an SU(2) isometry. Considering such a background we may write the metric as [88]

$$ds^{2} = G_{\mu\nu}(x)dx^{\mu}dx^{\nu} + 2G_{\mu i}(x)dx^{\mu}L^{i} + \frac{1}{2}g_{ij}(x)L^{i}L^{j}$$

where $\mu = 1, 2, ..., 7$ and i, j = 0, 8, 9. The L^i 's are the SU(2) Maurer-Cartan forms ²¹

$$L^i_{\pm} = -i \operatorname{Tr}(t^i g^{-1} \partial_{\pm} g).$$

where $t^i = \frac{1}{\sqrt{2}}\sigma^i$ are the SU(2) generators. The group element is given by

$$g = e^{\frac{i}{2}\phi\sigma_3} \cdot e^{\frac{i}{2}\theta\sigma_2} \cdot e^{\frac{i}{2}\psi\sigma_3}$$

with $0 \le \theta \le \pi$, $0 \le \phi \le 2\pi$ and $0 \le \psi \le 4\pi$. It is assumed that the NS sector also comprises an antisymmetric 2-form

$$B = B_{\mu\nu}(x)dx^{\mu} \wedge dx^{\nu} + B_{\mu i}(x)dx^{\mu} \wedge L^{i} + \frac{1}{2}b_{ij}(x)L^{i} \wedge L^{j}$$

and dilaton

$$\Phi = \Phi(x).$$

Note that all the dependence on the SU(2) Euler angles θ , ϕ and ψ is contained in the Maurer-Cartan 1-forms. All remaining data is dependent on the fields x^{μ} , called *spectator fields*.

NS Sector

The Lagrangian density for the NS sector fields is given by

$$\mathcal{L} = Q_{\mu\nu}\partial_+ X^{\mu}\partial_- X^{\nu} + Q_{\mu i}\partial_+ X^{\mu}L^i_- + Q_{i\mu}L^i_+\partial_- X^{\mu} + E_{ij}L^i_+L^j_-,$$

where

$$Q_{\mu\nu} = G_{\mu\nu} + B_{\mu\nu}, \quad Q_{\mu i} = G_{\mu i} + B_{\mu i}, \quad Q_{i\mu} = G_{i\mu} + B_{i\mu}, \quad E_{ij} = g_{ij} + b_{ij},$$

and $\mu = 1, 2, ..., 7$ and i, j = 0, 8, 9. Non-Abelian T-duality is performed through the replacement of ordinary derivatives with covariant derivatives, as follows

$$\partial_{\pm}g \to D_{\pm}g = \partial_{\pm}g - A_{\pm}g_{\pm}$$

followed by adding a Lagrange multiplier term to ensure that we have a flat connection (which further ensures that there is no curvature)

$$-i\mathrm{Tr}(vF_{\pm}), \quad F_{\pm} = \partial_{+}A_{-} - \partial_{-}A_{+} - [A_{+}, A_{-}].$$

 $^{^{21}\}mathrm{For}$ group theory conventions, see Appendix A.

where the v_i 's are Lagrange multipliers. There is more than one Lagrange multiplier present when considering a non-Abelian theory. This is because instead of having a single field, we now have a matrix of fields. Under this transformation, the spectator fields, x^{μ} remain inert. Partial integration results in the Lagrange multiplier taking the form

$$\operatorname{Tr}(i\partial_+ vA_- - i\partial_- vA_+ - A_+ fA_-), \quad f_{ij} = f_{ij}^k v_k.$$

At this stage, we have managed to introduce three new degrees of freedom. To eliminate three of the variables, we need to fix a gauge. The simplest choice available to us is g = 1. Then integrate out the gauge fields to get

$$A^{i}_{+} = iM^{-1}_{ij}(\partial_{+}v_{j} + Q_{\mu j}\partial_{+}X_{\mu}), \quad A^{i}_{-} = -iM^{-1}_{ij}(\partial_{-}v_{j} - Q_{j\mu}\partial^{\mu}_{X}),$$
(1.7)

where the matrix M is given by

$$M_{ij} = E_{ij} + f_{ij}$$

Then substituting (1.7) back into the original Lagrangian yields the *dual* Lagrangian

$$\hat{\mathcal{L}} = Q_{\mu\nu}\partial_{+}X^{\mu}\partial_{-}X^{\nu} + (\partial_{+}v_{i} + \partial_{+}X^{\mu}Q_{\mu i})M_{ij}^{-1}(\partial_{-}v_{j} - Q_{j\mu}\partial_{-}X^{\mu}).$$
(1.8)

The v_i 's now take on the role of dual co-ordinates, just like the Lagrange multiplier in the Abelian case. The background fields of the NS sector for the T-dual theory may be read off from (1.8) as

$$\hat{Q}_{\mu\nu} = Q_{\mu\nu} - Q_{\mu i} M_{ij}^{-1} Q_{j\nu}, \quad \hat{E}_{ij} = M_{ij}^{-1},
\hat{Q}_{\mu i} = Q_{\mu j} M_{ji}^{-1}, \quad \hat{Q}_{i\mu} = -M_{ij}^{-1} Q_{j\mu}.$$

For the case of Abelian T-duality, the dilaton receives a contribution at the quantum level. The same applies to non-Abelian T-duality with the following contribution

$$\hat{\Phi}(x,v) = \Phi(x) - \frac{1}{2}\ln(\det M).$$

Note that it is the inverse of the matrix M above which determines the dual geometry.

RR Sector

The transformation rules for the RR fields were first discovered in the context of Abelian Tduality in [63] for supergravity theories in 9 spatial dimensions [88]. These rules were arrived at by considering the action of T-duality on spinors. These rules were written down for the spacetime perspective in [52, 98], and for the Green-Schwarz string in [64, 65]. The pure spinor formalism was written down in [67, 99]. The current method [100] combines the RR fields with their Hodge duals ²² to form the bispinor as in equations (1.2) and (1.3). Then the non-Abelian T-dual theory is obtained by multiplication with Ω^{-1} . When transforming a type IIB

²²See Appendix A for conventions.

supergravity into a massive type IIA supergravity, we may obtain the rules for the transformation of the RR fluxes by comparing both sides of [88]

$$\hat{P} = P \cdot \Omega^{-1}.$$

For transformations from massive IIA to IIB, one simply switches the roles of P and \hat{P} .

1.4 Summary and Conclusion

Dualities provide a unifying link between the different string theories. In particular bosonic T-duality relates type IIA supergravity to type IIB supergravity. In its sim- plest form, we have shown that bosonic T-duality is an equivalence between a theory with a large spacetime radius and a theory with a small spacetime radius. A powerful approach used to derive the Buscher rules, which form a recipe for performing bosonic T-duality, is the path integral approach. This procedure comprises three main steps:

1. Begin with the string σ -model describing some spacetime and gauge a U(1) isometry of this spacetime (or SU(2) in the case of non-Abelian T-duality).

2. Using a flat connection for this gauge field, we insert a Lagrange multiplier.

3. Integrate by parts to obtain an action with a non-dynamical gauge field that may be eliminated by its equations of motion, to produce the T-dual σ -model.

In summary, given a non-linear σ -model with a target space that admits an isometry, the Buscher procedure gives us a well defined recipe for obtaining a T-dual σ -model. There are some subtleties that occur when trying to prove that the Buscher procedure holds on worldsheets of higher genera [10,59]. Abelian T-duality Buscher rules are only valid to the lowest order in string perturbation theory, as is true for the non-Abelian case. However, by gauging an Abelian isometry, the duality may be extended to higher genus worldsheets. In this case, the shift symmetry has to be along a compact coordinate [59, 101]. The full set of bosonic T-duality transformations is given by the non-Abelian group SO(D, D, Z). It is generated by T-dualities on all D circles, linear redefinitions of axes and discrete shifts of the Kalb-Ramond field B [49]. Naturally, we may wonder about the role that non-Abelian T-duality plays in the gauge/string correspondence. To answer this question, Sfetsos and Thompson [88] studied the dualisation of $AdS_5 \times S^5$ with an SU(2) isometry group. Non-Abelian T-duality studied in this context has many efforts, which may be found in [19, 90, 102-104] and references therein. A far more thorough review of non-Abelian T-duality can be found in [99]. It is important to stress that Abelian T-duality is an exact duality of superstring theory whereas it is not known whether non-Abelian T-duality is a duality. However, it has proven useful as a solution generating technique, which acts at the level of the supergravity. The take home idea is that bosonic T-duality may be used to generate

new solutions of supergravity. This is very useful for mapping very distinct geometries to one another.

Chapter 2

Introductory Fermionic T-Duality

2.1 Introduction

Fermionic T-duality [21,22] is the cousin of bosonic T-duality which extends the idea to a duality that acts on the whole superspace. Initially, the motivation to study fermionic generalisations to bosonic T-duality arose as a result of the need to understand dual superconformal symmetry with regards to scattering amplitudes of the string, as well as trying to understand its connection to integrability. Dual superconformal symmetry eventually helped show that $AdS_5 \times S^5$ was self-dual [21, 22]. This discovery shed light on hidden symmetries in the scattering amplitudes of supersymmetric theories like $\mathcal{N} = 4$ super Yang-Mills [26–28], and suddenly there was a great desire to explain these symmetries which fuelled research centred on the topic of fermionic Tduality. Let us take a step back to recall some of the things that we already know, for context. We have seen that bosonic T-duality has been crucial in linking the two type II string theories, and that the prerequisite for this duality transformation is an isometry on the background. For bosonic T-duality, Buschers procedure tells us to use a shift symmetry of the target space coordinate, corresponding to the isometry on the target space of the σ -model. Then, we are able to make some field redefinitions in the σ -model. Classically, the new σ -model is equivalent to the original σ -model, and both models have identical forms of the action. However, the new σ -models couplings [101], that is the metric and 2-form NSNS field look different. If the dilaton transforms as well, then the σ -models are equivalent at the quantum level. Additionally, in the bosonic case, the transformed background fields are related to the original fields by the Killing *vectors* corresponding to the isometry.

The generalisation to target spaces possessing fermionic isometries, or supersymmetries, as well as isometries, resulted in the duality of tree-level type II string theory called fermionic T-duality [101]. Under this duality only the RR fields and dilaton transform, leaving the rest of the NSNS sector (metric and NSNS 2-form) invariant. Analogous to bosonic T-duality, the transformation of the background supergravity fields are generated by the Killing spinors

corresponding to the superisometry of the superspace. It was Berkovits and Maldacena [22] and simultaneously Beisert et. al. [21], who in 2008, managed to generalise the Buscher procedure for worldsheets with actions invariant under constant shifts of the spacetime fermionic coordinates θ^J , J = 1, ..., n, where n is the number of supersymmetries. The metric and NSNS 2-form remain invariant and the transformation laws of the RR fields are given in terms of the bispinor field strength. In type IIA string theory, the bispinor field strength is

$$F^{\alpha\beta} = \frac{1}{2} F^{(2)}_{a_1 a_2} (\Gamma^{a_1 a_2})^{\alpha\beta} + \frac{1}{4!} F^{(4)}_{a_1 \dots a_4} (\Gamma^{a_1 \dots a_4})^{\alpha\beta},$$

where the 2-form $F^{(2)}$ and 4-form $F^{(4)}$ are the RR field strengths. In type IIB string theory the bispinor field strength is¹

$$F^{\alpha\beta} = F_a^{(1)}(\Gamma^a \sigma^1) + \frac{1}{3} F_{a_1 \dots a_3}^{(3)}(\Gamma^{a_1 \dots a_3}(i\sigma^2))^{\alpha\beta} + \frac{1}{2 \cdot 5!} F_{a_1 \dots a_5}^{(5)}(\Gamma^{a_1 \dots a_5} \sigma^1)^{\alpha\beta}$$

where the 1-form $F^{(1)}$, the 3-form $F^{(3)}$ and the 5-form $F^{(5)}$ are the RR field strengths. The 5-form $F^{(5)}$ is also self-dual with respect to the Hodge star operator². There are important differences between bosonic T-duality and its fermionic cousin. We will note them here:

- 1. Fermionic T-duality is not an exact duality symmetry of string theory. The symmetry is broken at one loop in the string coupling, g_s .
- 2. The transformation laws for the background fields resulting from the fermionic Buscher procedure are quite different when compared to the bosonic case. Strikingly, the NSNS sector is left untouched save the dilaton, which picks up the following additive correction

$$\phi' = \phi + \frac{1}{2}\log C,$$

where C is determined by the Killing spinors $(\epsilon, \hat{\epsilon})$, which parametrize the fermionic isometries. Note that this pair of Killing spinors represents a single supersymmetry.

3. The transformation of the bosonic fields in the RR sector may be written concisely in terms of the bispinor $F^{\alpha\beta}$:

$$e^{\phi'}F^{\prime\alpha\beta} = e^{\phi}F^{\alpha\beta} + 16i\epsilon^{\alpha}\hat{\epsilon}^{\beta}C^{-1}, \qquad (2.1)$$

where the bispinor³ is found, as before, by contracting all the RR fields in the theory with

¹Notice the Pauli matrices present in the IIB case. Type IIB string theory is chiral (left/right asymmetric). As a result we are dealing with two-component Majorana-Weyl spinors $\eta = (\epsilon, \hat{\epsilon})$. Each of the Killing spinors in the pair have 16-components. Given that we are discussing type IIB string theory the two 16-dimensional Killing spinors have the same chirality.

²See appendix A for conventions.

³Note that the conventions for the factor in front of the second term on the right hand side of (2.1) differ from author to author.

the appropriate antisymmetrized products of Γ -matrices.

- 4. There is an important new feature which occurs for fermionic transformations. The Killing spinors need to be complex in order to satisfy the commutativity condition, which will be discussed below, this ensures that the symmetry we gauge in the Buscher procedure is Abelian. Therefore, our supersymmetries commute. The consequence is that performing a fermionic T-duality transformation will result in a complexified solution of supergravity.
- 5. It is desirable to return to a real background, as we would like to describe physical processes, to this end it is essential to apply a bosonic T-duality along a time-like Killing vector. The returns a real solution with real fluxes. However, this does mean that our solutions obtained by using fermionic T-duality as a 'solution generating technique' are limited to supergravity solutions possessing a time-like Killing vector.

Take note that the transformation law for the dilaton is very similar to the way that the dilaton changes under a bosonic T-duality with the essential difference being that in the fermionic case, the sign of the logarithm has changed. This difference turns out to be extremely important for establishing the exact T-self-duality of the $AdS_5 \times S^5$ background. We will deal with self-duality later but, simply put, it means that the original background is returned to after a sequence of bosonic and fermionic T-duality transformations has been applied. When applying bosonic transformations in order to establish self-duality, we always need to apply an even number. This occurs because the bosonic transformations swap the type IIA and type IIB backgrounds. Thus, to return to the same background, an even number of bosonic T-dualities is required.

2.2 The Fermionic Buscher Procedure

As before, we start by considering a σ -model, in this case a *Green-Schwarz* σ -model on the whole superspace, depending on both bosonic and fermionic worldsheet variables (x^m, θ^μ) . Next, choose one of the fermionic directions, say θ^1 , such that the worldsheet action is invariant under a constant shift of the fermionic variable. This means that

$$\theta^1 \to \theta^1 + \rho, \quad x^m \to x^m, \quad \theta^{\tilde{\mu}} \to \theta^{\tilde{\mu}},$$

$$(2.2)$$

where ρ is a fermionic constant and $\tilde{\mu}$ runs over all fermionic degrees of freedom except $\mu = 1$. These types of backgrounds inherently preserve a supersymmetry [22]. Invariance under (2.2) implies that θ^1 will only appear as derivatives in the action, which is given by

$$S = \int d^2 z \left[B_{11}(Y) \partial \theta^1 \bar{\partial} \theta^1 + L_{1M}(Y) \partial \theta^1 \bar{\partial} Y^M + L_{M1}(Y) \partial Y^M \bar{\partial} \theta^1 + L_{MN} \partial Y^M \bar{\partial} Y^N \right],$$
where $Y^M = (x^m, \theta^{\tilde{\mu}})$ and $M = (m, \tilde{\mu})$ ranges over all indices except for $\mu = 1$. We define $L_{MN}(Y) = G_{MN}(Y) + B_{MN}(Y)$ where G_{MN} is the graded-symmetric tensor and B_{MN} is the graded-antisymmetric tensor defined by

$$G_{MN} = (-1)^{MN} G_{NM}, \quad B_{MN} = -(-1)^{MN} B_{MN},$$

with

$$(-1)^{MN} = \begin{cases} -1 & \text{M, N both fermionic,} \\ +1 & \text{Otherwise} \end{cases}$$

where the background fields are contained as components. Once again, we introduce a vector field (A, \overline{A}) , which is composed of fermionic gauge fields. The derivatives are then replaced with the vector fields as follows:

$$(\partial \theta^1, \bar{\partial} \theta^1) \to (A, \bar{A}),$$

where (A, \overline{A}) forms an auxiliary field on the worldsheet. This replacement is viewed as gauging the shift symmetry of the original Green-Schwarz σ -model by a minimal coupling to the fermionic gauge field. If $B_{11}(Y)$ is non-zero then we may apply the Buscher procedure in order to T-dualize the σ -model with respect to θ^1 . Additionally, a Lagrange multiplier term $\tilde{\theta}^1$ is added to the action for the purpose of imposing that the vector field is a derivative of a fermionic scalar. The resulting action is

$$S = \int d^2 z \left[B_{11}(Y) A \bar{A} + L_{1M}(Y) A \bar{\partial} Y^M + L_{M1}(Y) \partial Y^M \bar{A} + L_{MN} \partial Y^M \bar{\partial} Y^N + \tilde{\theta}^1 (\partial \bar{A} - \bar{\partial} A) \right].$$

To return to the original Green-Schwarz σ -model, the Lagrange multiplier $\tilde{\theta}^1$, must be integrated out. This imposes that $A = \partial \theta^1$ and $\bar{A} = \bar{\partial} \theta^1$. Alternatively, we may integrate out the fermionic gauge field instead, using the equations of motion for A given by

$$\partial \tilde{\theta}^1 = B_{11}A + L_{M1}\partial Y^M,$$

$$\bar{\partial} \tilde{\theta}^1 = B_{11}\bar{A} - (-1)^{s(M)}L_{1M}\bar{\partial}Y^M,$$

where the exponent s(M) is zero if M is a bosonic index and one if M is fermionic. This results

in the T-dualized action

$$\tilde{S} = \int d^2 z \left[B'_{11}(Y) \partial \tilde{\theta}^1 \bar{\partial} \tilde{\theta}^1 + L'_{1M}(Y) \partial \tilde{\theta}^1 \bar{\partial} Y^M + L'_{M1}(Y) \partial Y^M \bar{\partial} \tilde{\theta}^1 + L'_{MN} \partial Y^M \bar{\partial} Y^N \right]$$

where

$$B'_{11} = -(B_{11})^{-1}, \quad L'_{1M} = (B_{11})^{-1}L_{1M},$$
$$L'_{M1} = (B_{11})^{-1}L_{M1}, \quad L'_{MN} = L_{MN} - \frac{1}{B_{11}}L_{1N}L_{M1}.$$

There is a change induced in the dilaton due to the measure factor coming from the integration over the fermionic vector field (see [106] to see how the measure factor arises for the case of integration over a bosonic field). The change is given by

$$\phi' = \phi + \frac{1}{2}\log B_{11}$$

Note that the change in ϕ as a result of fermionic T-duality has a logarithm with the opposite sign from that appearing in the change in ϕ as a result of bosonic T-duality. This is because, whilst the integration of the vector field has the same formal structure as in the bosonic case, we are instead integrating over an anticommuting variable, which picks up an additional minus sign. Fermionic T-duality transformations leave the NSNS fields G_{MN} and B_{MN} invariant, as mentioned above. Therefore, fermionic T-duality is primarily a transformation acting on the RR sector. The transformation is written in terms of the bispinor field strength given in (2.1). Furthermore, the fermionic isometries are assumed to be Abelian [22], so that they satisfy the following *commutativity* constraint

$$\epsilon \gamma^m \epsilon + \hat{\epsilon} \gamma^m \hat{\epsilon} = 0, \tag{2.3}$$

in type IIB and

$$\bar{E}\Gamma^m E = 0 \tag{2.4}$$

in type IIA for a Majorana spinor $E = \epsilon + \hat{\epsilon}$. The two spinors ϵ and $\hat{\epsilon}$ are not independent, instead they are related by the Killing spinor equations and the above constraint. This constraint is unique to fermionic T-duality and cannot be solved for real spinors, therefore the Killing spinors need to be artificially complexified, as we will see in the next section. The consequence of this is that performing a fermionic T-duality transformation will result in complex RR fields. Once a pair of Killing spinors satisfying the commutativity condition have been found, the next step is to calculate the auxiliary scalar field C which is defined as follows

$$\partial_m C = i\epsilon^{\alpha} (\gamma_m)_{\alpha\beta} \epsilon^{\beta} - i\hat{\epsilon}^{\hat{\alpha}} (\gamma_m)_{\hat{\alpha}\hat{\beta}} \hat{\epsilon}^{\hat{\beta}} \quad \text{for type IIB,} \\ \partial_m C = i\bar{E}\Gamma_m \Gamma^{11} E \qquad \text{for type IIA.}$$

The auxiliary field C is the $\theta = \hat{\theta} = 0$ component of B_{11} . Using the constraint (2.3), we are able to somewhat simplify the type IIB expression to

$$\partial_m C = 2i\epsilon\gamma_m\epsilon. \tag{2.5}$$

The dilaton can be written in terms of this auxiliary field as follows

$$\phi' = \phi + \frac{1}{2}\log C,\tag{2.6}$$

and we recall that the RR fields transform according to

$$e^{\phi'}F' = e^{\phi}F + \frac{16}{i}\frac{\epsilon\otimes\hat{\epsilon}}{C}.$$
(2.7)

For the case when multiple fermionic T-dualities are performed, with respect to several supersymmetries parametrized by pairs of Killing spinors $\varepsilon_i = (\epsilon_i, \hat{\epsilon}_i)$ with $i \in 1, ..., n$, we may generalise equations (2.5), (2.6) and (2.7) to

$$\partial_m C_{ij} = 2i\epsilon_i \gamma_m \epsilon_j, \tag{2.8}$$

$$\phi' = \phi + \frac{1}{2} \sum_{i=1}^{n} (\log C)_{ii}, \qquad (2.9)$$

$$e^{\phi'}F' = e^{\phi}F + \frac{16}{i}\sum_{i,j=1}^{n} (\epsilon_i \otimes \hat{\epsilon}_j)(C^{-1})_{ij}.$$
 (2.10)

The commutativity condition, for all $i, j \in 1, .., n$, becomes

$$\epsilon_i \gamma_m \epsilon_j + \hat{\epsilon}_i \gamma_m \hat{\epsilon}_j = 0. \tag{2.11}$$

2.3 A Recipe for Fermionic T-duality

Bakhmatov, in his thesis [16], formulated a great recipe for performing fermionic T-duality on any solution. In this section, we will present this recipe in detail.

Performing Fermionic T-duality

The steps are:

- 1. Find the Killing spinors of the solution.
- 2. Choose a complex linear combination of the Killing spinors from the first step, ensuring that they satisfy the commutativity condition (2.3).
- 3. Calculate the auxiliary scalar field C, using (2.5).
- 4. If there are any RR fields existing in the original background, they should be substituted into

$$F^{\alpha\beta} = (\gamma^{\mu})^{\alpha\beta} F_{\mu} + \frac{1}{3} (\gamma^{\mu_1\mu_2\mu_3})^{\alpha\beta} F_{\mu_1\mu_2\mu_3} + \frac{1}{2.5!} (\gamma^{\mu_1\dots\mu_5})^{\alpha\beta} F_{\mu_1\dots\mu_5}.$$
 (2.12)

- 5. One then uses $F^{\alpha\beta}$, ϵ^{α} , $\hat{\epsilon}^{\beta}$ and C to calculate the transformed RR background $F'^{\alpha\beta}$ using (2.7).
- 6. Use (2.12), but this time use it to find the contributions of F_1 , F_3 and F_5 to the transformed background $F'^{\alpha\beta}$ separately.

A few comments are in order. Type IIA supergravity contains two 16-component, Majorana-Weyl spinors with opposite chirality. It is convenient to combine the two Majorana-Weyl spinors into a single 32-component spinor $E = (\epsilon, \hat{\epsilon})$, and to then work with constraint equations and the Killing spinor equations in terms of the full 32×32 Γ -matrices. The type IIB case is handled differently. Our Killing spinors are, again, $\varepsilon = (\epsilon, \hat{\epsilon})$, but now the two 16-component spinors have the same chirality. For IIB string theory, we work with constraint equations and the Killing spinor equations in terms of the $16 \times 16 \gamma$ -matrices. The fermionic directions along which we dualise are found from the Majorana-Weyl spinors by creating complex linear combinations of the Killing spinors found in step 1. The next step is to use the complex Killing spinors to write down a differential equation (??), which may be integrated to solve for the auxiliary scalar field C. The bispinor (??) in step 4 contains antisymmetrized products of γ -matrices with the RR

fields. The single γ -matrix and the product of five γ -matrices end up being symmetric overall. However, the triple product of γ -matrices is antisymmetric overall. After calculating the new bispinor $F'^{\alpha\beta}$ in step 5, we encounter the first non-trivial, challenging part of the transformation in step 6. Here we need to extract the transformed RR fields from the new bispinor. To find the corresponding 1-form, 3-form and 5-form, we need to separate the matrices occurring in (??) into symmetric and antisymmetric parts. This done by brute force using *Mathematica*. Steps 2, 4 and 5 are all automated with *Mathematica*, since all these equations involve calculations with many 32×32 (type IIA) and 16×16 (type IIB) matrices which are very cumbersome.

2.4 Self-duality of the $AdS_5 \times S^5$ Background

Before moving on to the study of backgrounds with less than maximal supersymmetry, it is instructive to review the maximally supersymmetric solution, $AdS_5 \times S^5$, first. This review follows the structure in [23].

To show that $AdS_5 \times S^5$ is self-dual under a sequence of bosonic and fermionic T-dualities, we choose to perform bosonic T-duality along the (x^0, x^1, x^2, x^3) -directions of AdS_5 in Poincaré coordinates. Following this, we perform fermionic T-duality along 8 Poincaré supersymmetries [21, 22]. The corresponding Poincaré Killing spinors are those spinors independent of the boundary coordinates of AdS_5 . In Poincaré coordinates the metric on the AdS_5 space is

$$ds^{2}(AdS_{5}) = \frac{1}{r^{2}}(dx^{m}dx_{m} + dr^{2}),$$

where m = 0, 1, 2, 3 and r is the radial direction in AdS_5 . Solving for the Poincaré Killing spinors η of $AdS_5 \times S^5$ gives the following solution

$$\eta = \frac{1}{\sqrt{r}}\tilde{\eta}$$

where we have suppressed the angular dependence of the S^5 in $\tilde{\eta}$. We will see this form of the solution appearing for the less supersymmetric backgrounds in Part II. The $AdS_5 \times S^5$ geometry has a self-dual⁴ 5-form RR flux and a trivial (constant) dilaton. We may write the flux bispinor (which is used in the fermionic Buscher procedure) as [22]

$$e^{\phi}F^{\alpha\hat{\beta}} = (\gamma_{01234})^{\alpha\hat{\beta}}$$

⁴Self-dual with respect to the Hodge star operator defined in the Appendices.

where 0, 1, 2, 3 correspond to the x^m directions and 4 corresponds to the radius of AdS_5 . The maximally supersymmetric geometry, $AdS_5 \times S^5$, has 32 supersymmetries which are reduced to 16 by restricting to Poincaré supersymmetries. The commutativity constraint 2.3 further reduces this number to 8 pairs of complexified Killing spinors corresponding to 8 commuting fermionic directions. The matrix C can be calculated using 2.5 where upon being inverted it is used to find the transformed bispinor 2.10. The T-dual bispinor is [21, 22]

$$e^{\tilde{\phi}}\tilde{F}^{\alpha\hat{\beta}} = (i\gamma_4)^{\alpha\hat{\beta}}.$$

Using C again allows us to find the shift in the dilaton

$$\tilde{\phi} = \phi + 4\log r.$$

To recover the original geometry we still need to perform bosonic T-dualities along the chosen x^m . It is during this process that the difference in sign before the logarithmic term in the dilaton shifts due to bosonic and fermionic transformations plays a crucial role. Bosonic T-duality produces a shift $\Delta \phi = -4 \log r$ which undoes the shift caused by fermionic T-duality. Note, that at this stage we are still left with a complex geometry, meaning that we are not done. To arrive at the original background we need to perform a timelike T-duality which attaches an i to all the RR fields [83] and recovers a real solution.

2.5 Summary and Conclusion

The aim of this chapter was to give a brief introduction to the concept of fermionic T-duality. For a thorough review with many references see [23], as well as [107] for some concrete examples, illustrating the fermionic T-duality transformation.

In summary, fermionic T-duality is a cousin of bosonic T-duality which extends the idea to work on the entire superspace. It is a tree-level duality symmetry of string perturbation theory and it preserves the supersymmetry of the background, with the Killing spinors belonging to the Abelian subalgebra of the superalgebra [16]. From the supergravity perspective, fermionic T-duality rotates Killing spinors, preserving chirality in addition to supersymmetry. The pair $\varepsilon = (\epsilon, \hat{\epsilon})$ generates a single supersymmetry transformation.

To apply the Buscher procedure to the fermionic generalisation, we require that our fermionic isometries (or supersymmetries) commute [23]. The immediate consequence of this is that the fermionic T-duality transformation does not admit Majorana (or real) Killing spinors. Therefore, we need complex Killing spinors which result in complex solutions to supergravity.

In the introduction we noted the differences between bosonic and fermionic T-duality transformations. Most notably, the difference in the sign appearing in front of the logarithm term in the dilaton shift changes to 'plus' for the fermionic case, where it was 'minus' in the bosonic case. This change ends up being essential for establishing the self-duality of $AdS_5 \times S^5$. In fact, the shift of the dilaton due to bosonic T-duality along the AdS_5 is exactly cancelled by the dilaton shift caused by fermionic T-duality, which results in a symmetry at the quantum level.

For completeness we give the explicit form of the Killing spinors of the T-dual theory:

$$\tilde{\epsilon}_i^{\alpha} = (C^{-1})_{ij} \epsilon_j^{\alpha}, \quad \tilde{\hat{\epsilon}}_i^{\hat{\alpha}} = (C^{-1})_{ij} \hat{\epsilon}_j^{\hat{\alpha}}.$$

Finally, we note that fermionic T-duality transformations do not commute with bosonic Tduality transformations. Part II

Supergravity

Chapter 3

Fermionic T-duality of $AdS_d \times S^d(\times S^d) \times M$

3.1 Introduction

In this chapter we consider the self-duality of some non-maximally supersymmetric backgrounds as a result of performing a series of bosonic and fermionic T-dualities. We have aimed to keep our computations as simple as possible by working directly with the fields at the level of the supergravity. This simple approach is a great way to develop intuition regarding the mechanism of T-duality transformations, by observing the changes in the RR fields explicitly. With the machinery laid bare, it is easy to notice the need for bosonic T-dualities along some torus directions, a fact which was nearly missed when working at the level of the coset¹. Although we find it most convenient to work with the IIA string theory as in [121], a bosonic T-duality will take us back to a IIB background. Thus, our proofs are also valid in the extension to IIB. The goal of this chapter is to provide an elementary explanation of the duality of these backgrounds, at the expense of generality. The recipe given in section 2.3 is what we apply to every example covered in this chapter. However, subtleties arise when considering the exceptional cases (i.e. backgrounds possessing two spheres) and therefore additional discussion is required. At the end of each example we present a diagram that summarizes the technical details of that section in order to provide a 'big picture' view. T-duality in the general context is considered in Chapter 4 which also discusses some of the examples presented in this chapter. This chapter is based on research covered in [36].

¹The non-linear σ -models that we use in this thesis may be described by cosets G/H, where G is the isometry group of the spacetime manifold and H is the isotropy group. See Chapter 4 for details and the Appendices for mathematical definitions.

3.2 Type IIA Supergravity

Since we are working with IIA supergravity it is necessary to give the basic layout of fermionic T-duality in this formalism. Here we have two Majorana-Weyl spinors of opposite chirality. We use a basis in which the chirality operator is given by $\Gamma_{11} = \sigma^3 \otimes 1$. Then we may write the spinors as follows

$$\begin{pmatrix} \epsilon^{\beta} \\ 0 \end{pmatrix}$$
 and $\begin{pmatrix} 0 \\ \hat{\epsilon}_{\beta} \end{pmatrix}$, $\beta = 1, ..., 16.$

We may combine the Killing spinors into a single 32-dimensional $E = (\epsilon, \hat{\epsilon})$. Greek indices denote curved spatial directions, given by $\mu, \nu = t, x, z, \theta_+, ..., \psi_-, u$. Latin indices represent flat spacetime coordinates given by m, n = 0, 1, ..., 9. Spinor indices are given by α and β . Following [121], the Killing spinors must ensure that the gravitino and dilatino variations² vanish, which implies that

$$D_{\mu}E = -\frac{1}{8}S\Gamma_{\mu}E, \quad TE = 0 \tag{3.1}$$

where the covariant derivative is

$$D_{\mu}E = \partial_{\mu}E + \frac{1}{4}\omega_{\mu np}\Gamma^{np}E$$

and

The supergravity solutions that we are interested in have the fermionic fields set to zero. Variations under supersymmetry have the following form:

$$\delta(\text{boson}) \propto \text{fermion}, \quad \delta(\text{fermion}) \propto \text{boson}.$$

Immediately it follows that the bosonic field supersymmetry variation is $\delta(\text{boson}) = 0$. By a supersymmetric solution, we mean that

²See appendix A for more details.

 $\delta(\text{boson}) = 0 = \delta(\text{fermion}).$

Therefore, the only condition required to have any supersymmetry preserved is that the fermionic field supersymmetry variation vanishes: $\delta(\text{fermion}) = 0$. The Killing spinor equations (3.1) do precisely this, they set these fermionic variations to zero and we are able to determine what supersymmetries are present. The Buscher procedure for the fermionic generalisation, as derived by Berkovits and Maldacena [22], is employed where the direction along which we dualise is specified by a complex Killing spinor. That is, a fermionic isometry or, supersymmetry. For example, $\mathcal{E} = E_1 + iE_2$ is one such combination where $E_i = (\epsilon_i; \hat{\epsilon}i)$ are solutions of (3.1). We are able to perform T-duality along multiple directions \mathcal{E}_i at once, however, they must obey a commutativity condition given by

$$\bar{\mathcal{E}}_i \Gamma_\mu \mathcal{E}_j = 0 \ \forall \mu, i, j, \quad \text{where } \bar{\mathcal{E}} \equiv \mathcal{E}^T \Gamma^0.$$
(3.2)

Equation (3.2) is the statement that our isometry is Abelian³, which is required to perform fermionic T-duality. This is because Berkovits and Maldacena followed Buscher's procedure and gauged an Abelian symmetry (as was done for the bosonic case). Condition (3.2) is not solved by any real spinors, thus we use complex Killing spinors. The next step is to find a matrix Cfrom

$$\partial_{\mu}C_{ij} = i\bar{\mathcal{E}}_i\Gamma_{\mu}\Gamma_{11}\mathcal{E}_j,\tag{3.3}$$

which gives us the change in the dilaton

$$\Delta \Phi \equiv \Phi' - \Phi = \frac{1}{2} \log(\det C), \qquad (3.4)$$

and the change in the RR forms

$$\Delta F \equiv e^{\Phi'} F^{\prime \alpha}_{\beta} - e^{\Phi} F^{\alpha}_{\beta} = \frac{16}{i} (C^{-1})_{ij} \epsilon^{\alpha}_i \hat{\epsilon}_{j\beta}.$$
(3.5)

Note that in (3.5) the lower-case spinors are complex, i.e. $\mathcal{E}_i = (\epsilon_i, \hat{\epsilon}_i)$. The bispinor F encoding the RR forms is given by

 $^{^{3}}$ Note that at the moment, there does not seem to be a meaningful reason for why this must be true. Perhaps future discussions will reveal a deeper understanding. What we can say is that, if our isometries do not commute, the situation is more complex.

$$F^{\alpha}_{\beta} = \frac{1}{2!} F^{(2)}_{mn} (\gamma^m)^{\alpha\gamma} (\gamma^n)_{\gamma\beta} + \frac{1}{4!} F^{(4)}_{mnpq} (\gamma^m \gamma^n \gamma^p \gamma^q)^{\alpha}_{\beta}.$$
(3.6)

The lower-case gamma matrices are related to the upper-case gamma matrices by

$$\Gamma^m = \begin{bmatrix} 0 & (\gamma^m)^{\alpha\beta} \\ (\gamma^m)_{\alpha\beta} & 0 \end{bmatrix} \,.$$

That is, the lower-case gamma matrices are the 16×16 off-diagonal blocks of the larger 32dimensional gamma matrices. See Appendix A for our conventions.

3.3 $AdS_2 \times M$ Backgrounds

This section considers $AdS_2 \times S^2(\times S^2) \times M$ backgrounds, all of which possess 8 supersymmetries⁴. However, we neglect the superconformal Killing spinors in favour of the Poincaré Killing spinors as in [23]. Poincaré Killing spinors depend solely on the radial direction of the AdS subspace. These spinors are most useful in exhibiting self-duality and tend to have a simpler explicit form. Using Poincaré Killing spinors restricts us to considering 4 Killing spinors, which allows for 2 complex directions, upon complexification, along which to perform fermionic T-duality. The metric has a parameter α that has range (0, 1), such that when $\alpha \to 1$, the second sphere approaches f flat spacetime. This decouples the sphere from the $AdS_2 \times S^2$ subspace which results in these directions being re-compactified into a T^2 , contributing to torus directions required to make $AdS_2 \times S^2 \times T^6$. The general metric is given by

$$ds^{2} = ds^{2}_{AdS} + \frac{1}{\alpha} ds^{2}_{S_{+}} + \frac{1}{1 - \alpha} ds^{2}_{S_{-}} + \sum_{i} dy^{2}_{i} = \eta_{mn} e^{m}_{\mu} e^{n}_{\nu} dx^{\mu} dx^{\nu}.$$
 (3.7)

We adopt Poincaré coordinates for AdS_2

$$ds_{AdS_2}^2 = \frac{-dt^2 + dz^2}{z^2},$$

with implied spin connection components $\omega_{t[01]} = \frac{1}{z}$. For the spheres we have the nested form

$$ds_{S_{\pm}^{3}}^{2} = d\theta_{\pm}^{2} + \sin^{2}\theta_{\pm}d\phi_{\pm}^{2}, \qquad (3.8)$$

⁴The notation $AdS_2 \times S^2(\times S^2) \times T$ identifies two classes of solutions: the $AdS_2 \times S^2 \times T^6$ backgrounds and the exceptional $AdS_2 \times S^2(\times S^2) \times T^4$ backgrounds. Therefore, the brackets indicate that the second sphere is optional with the value of the parameter α specifying which background is being dealt with.

with implied spin connection components $\omega_{\phi_+[23]} = -\cos\theta_+$ and $\omega_{\phi_-[45]} = -\cos\theta_-$. Note that the spin connection components $\omega_{\mu ab}$ are independent of α although the vielbeins e^m_{μ} are not. For what follows we take $\Phi = 0$ and use the following RR flux:

$$\mathbf{F}^{(4)} = \Gamma^{01}(-\Gamma^{67} + \Gamma^{89}) + \sqrt{\alpha}\Gamma^{23}(\Gamma^{68} + \Gamma^{79}) + \sqrt{1 - \alpha}\Gamma^{45}(-\Gamma^{78} + \Gamma^{69}).$$
(3.9)

For $\alpha = 1$, bosonic T-duality along the x^5 direction takes us to the IIB $AdS_2 \times S^2 \times T^6$ case studied in [39], with $F^{(5)}$ only. Following [121] we may write

$$\mathbf{F}^{(4)} = -4\Gamma^{0167}P_1(1-P_2).$$

where

$$P_1 = \frac{1}{2}(1 + \Gamma^{6789}), \quad P_2 = \frac{1}{2}(1 + \sqrt{\alpha}\Gamma^{012378} + \sqrt{1 - \alpha}\Gamma^{014568}),$$

are projection operators. The Killing spinor equation is

and we can build up an explicit solution $E = z^{-1/2} R_{S_+} R_{S_-} \xi$ as shown in the box that follows.

Explicit Form of the Killing Spinors

Firstly we restrict to Poincaré Killing spinors, i.e. those independent of the boundary coordinates of AdS. For AdS_2 this means that $\partial_t E = 0$. Thus the $\mu = t$ equation becomes

$$\frac{1}{2z}\Gamma^{01}E = \frac{1}{2}\Gamma^{0167}P_1(1-P_2)\Gamma_t E,$$

which simplifies to the constraint

$$E = -\Gamma^{067} P_1 P_2 E.$$

The $\mu = z$ equation then reads $\partial_z E = -\frac{1}{2z}E$ which has the solution $E \propto z^{-1/2}$.

We then take a look at the first sphere. For $\mu \in S_+$ we get

$$D_{\mu}E = \frac{\sqrt{\alpha}}{2}\Gamma_{\mu}\Gamma^{2368}E, \qquad (3.10)$$

where we have used the fact that $P_1E = P_2E = E$. The Killing spinors for S^2 are given in [?]. The equation

$$D_{\mu}\epsilon = \pm \frac{i}{2}e^{a}_{\mu}\Gamma_{a}\epsilon$$

is solved by

$$\epsilon = \exp\left(\pm i\frac{\theta}{2}\Gamma^2\right)\exp\left(\frac{\phi}{2}\Gamma^{23}\right).$$

The θ equation is easy to solve, but the ϕ equation requires more work. One needs to Taylor expand the exponentials and use the fact that $(i\Gamma^2)^2 = -1$ and $\{i\Gamma^2, \Gamma^{23}\} = 0$. In our case, we replace $i\Gamma^2 \to \Gamma^{368}$ which still satisfies these conditions. Thus

$$R_{S_{+}} = \exp\left(\frac{\theta_{+}}{2}\Gamma^{368}\right)\exp\left(\frac{\phi_{+}}{2}\Gamma^{23}\right).$$

Then we take a look at the second sphere. For $\mu \in S_{-}$ we get

$$D_{\mu}E = \frac{\sqrt{1-\alpha}}{2}\Gamma_{\mu}\Gamma^{4578}E,$$

where $[\Gamma_{\mu}\Gamma^{4578}, R_{S_+}] = [D_{\mu}, R_{S_+}] = 0$, meaning that this is the equation which must be solved by R_{S_-} alone.

Finally, we arrive at the complete solution

$$E(\xi) = \frac{1}{\sqrt{z}} \exp\left(\frac{\theta_+}{2}\Gamma^{368}\right) \exp\left(\frac{\phi_+}{2}\Gamma^{23}\right) \exp\left(\frac{\theta_-}{2}\Gamma^{578}\right) \exp\left(\frac{\phi_-}{2}\Gamma^{45}\right) \xi,$$
(3.11)

with constant ξ obeying

$$-\Gamma^{067}\xi = P_1\xi = P_2\xi = \xi. \tag{3.12}$$

There are exactly 4 Killing spinors ξ_a , and we may take them to be orthonormal and normalised: $\xi_a \cdot \xi_b = \delta_{ab}$.

We need complex combinations \mathcal{E}_i of the Killing spinors E_i above in order to perform fermionic T-duality. In general we may write these as

$$\mathcal{E}_i = b_{ia} E(\xi_a), \quad a = 1, \dots 4.$$

Then the commutativity condition (3.2) becomes

$$b_{ia}V^{\mu}_{ab}b_{jb} = 0, \quad V^{\mu}_{ab} \equiv \bar{E}_a\Gamma^{\mu}E_b.$$
 (3.13)

For the AdS_2 case we can show that $V^{\mu} = 0$, using $\Gamma^{067}E = -E$, except for $\mu = t$, where

$$V_{ab}^t = E_a^T \Gamma_z^0 \Gamma^0 E_b = -\xi_a^T \xi_b = \delta_{ab}.$$

So that the commutativity condition on the coefficients b_{ia} is simply $b_{ia}b_{ja} = 0$.

3.3.1 The case $\alpha = 1$: $AdS_2 \times S^2 \times T^6$

This is the simplest AdS_2 case. Here we can proceed in a similar fashion to [23]'s treatment of type IIB $AdS_3 \times S^3 \times T^4$. However, this case is even simpler as we only have two complexified Killing spinors.

With $P_2E = E$, the $\mu \in S_+$ equation (3.10) simplifies to

$$D_{\mu}E = \frac{1}{2}\Gamma_{\mu}\Gamma^{1}E,$$

and thus the resulting solution is given by

$$E(\xi) = \frac{1}{\sqrt{z}} \exp\left(\frac{\theta}{2}\Gamma^{21}\right) \exp\left(\frac{\phi}{2}\Gamma^{23}\right) \xi$$
(3.14)

with ξ_a , a = 1, ..., 4 still solving (3.12). We may write these as

$$(\xi_a)^{\beta} = \frac{1}{2}\delta_{2a-\beta-1} + \frac{1}{2}\delta_{2a-\beta+8} + \frac{1}{2}\delta_{2a-\beta+16} - \frac{1}{2}\delta_{2a-\beta+23}.$$

Where we note that each ξ has 4 non-zero entries at the positions given by the Kroneckerdelta's, with $\beta = 1, ..., 32$ temporarily. Our complex combinations $\mathcal{E}_i = b_{ia} E(\xi_a)$ are specified by choosing⁵

$$b_1 = (1, 0, i, 0), \quad b_2 = (0, 1, 0, i),$$

which corresponds to

$$\mathcal{E}_1 = E_1 + iE_3, \quad \mathcal{E}_2 = E_2 + iE_4.$$

Then, after integrating (3.3), we arrive at

$$C = \frac{1}{z} \begin{bmatrix} i\cos\theta_{+} - \sin\theta_{+}\sin\phi_{+} & -\cos\phi_{+}\sin\theta_{+} \\ -\cos\phi_{+}\sin\theta_{+} & i\cos\theta_{+} + \sin\theta_{+}\sin\phi_{+} \end{bmatrix}.$$

To find the shift in the dilaton due to fermionic T-duality, we take the determinant of C, then (3.4) gives

$$\Delta \Phi = \Phi' - \Phi = -\log(z). \tag{3.15}$$

Note that the background we arrive at after doing a fermionic T-duality is complex. It also shifts the dilaton. To get back to the original background, which has $\Phi = 0$, we need to cancel this shift and somehow obtain a real background. Bosonic T-duality along the time direction (which is also the only boundary direction of AdS_2) does precisely this. It undoes the shift by giving us $\Phi'' - \Phi' = \log(z)$ and makes our background real. It also changes the metric in the following way:

$$ds_{AdS}^2 = \frac{-dt^2 + dz^2}{z^2} \to -z^2 dt^2 + \frac{dz^2}{z^2} = \frac{-dt^2 + dz'^2}{z'^2},$$
(3.16)

where we have defined z' = 1/z in the last step in order to bring the metric back to its original form.

 $^{^5\}mathrm{Note}$ that choosing these complex combinations is not a well defined process and usually involves some trial and error.

Finally we consider the RR forms. The change in the bispinor F (3.5) has some real terms (which cancel the original flux (3.9)) and imaginary terms giving rise to a 2-form and a 4-form flux.

Then bosonic T-duality along the x^0, x^4, x^6 and x^7 directions recovers the original RR flux. The total number of bosonic T-dualities is four, half of which must be along torus directions. This was also the case for $AdS_5 \times S^5$ in [22] and for $AdS_3 \times S^3 \times T^4$ in [23].

In earlier treatments of $AdS_2 \times S^2$ found in [124,125], only the coset σ -model $PSU(1,1|2)/U(1)^2$ is studied, rather than the critical string theory. Here the need for bosonic T-duality along torus directions was not noticed. However, this approach was extended to include non-coset fermions in [36], which forced the inclusion of T-duality along torus directions.

3.3.2 $AdS_2 \times S^2 \times T^6$ with other **RR** fluxes

Earlier we studied the case $\alpha = 1$ with only a $F^{(4)}$ turned on. There exist other combinations of fluxes with the same underlying geometry. One such case, considered in [36, 39, 126], has flux given by

Showing self-duality for this background is no more difficult than the $F^{(4)}$ only case. We may write

$$S = -4\Gamma^{01}\Gamma_{11}(1-P),$$

in terms of the projection operator

$$P = \frac{1}{4}(1 - \Gamma^{6789} - \Gamma^{4589} - \Gamma^{4567}).$$

Here, the Killing spinors are still given by (3.14), however, the constraint on ξ is now given by

$$\Gamma^0 \Gamma_{11} P \xi = \xi.$$

We choose solutions so that

$$(\xi_a)^{\beta} = 0$$
 for $\beta < a$, $(\xi_a)^{\beta} > 0$ for $\beta = a_{\beta}$

and then make the following complex combinations:

$$\mathcal{E}_1 = iE_1 + E_4, \quad \mathcal{E}_2 = iE_2 - E_3.$$

This gives

$$C = \frac{1}{z} \begin{bmatrix} ie^{i\phi_+} \sin \theta_+ & -i\cos \theta_+ \\ -\cos \theta_+ & -ie^{-i\phi_+} \sin \theta_+ \end{bmatrix},$$

which yields a shift in the dilaton of $\Delta \Phi = -\log(z)$. This is identical to (3.15), and is similarly cancelled by a bosonic T-duality along the time direction. The change in RR forms is

$$\begin{split} \Delta F^{(2)} = &\gamma^{01}, \\ \Delta F^{(4)} = &-\gamma^{23}(\gamma^{45} + \gamma^{67} + \gamma^{89}) - i\gamma^{1468} + i\gamma^{1479} + i\gamma^{1569} + i\gamma^{1578}, \end{split}$$

We can clearly see that the real terms cancel the original flux, as expected. Finally, acting with bosonic T-duality along the directions $t = x^0, x^4, x^6$ and x^8 returns us to (3.18).

3.3.3 The case $\alpha < 1$: $AdS_2 \times S^2 \times S^2 \times T^4$

Our generic complex combinations are given by $\mathcal{E}_i = b_{ia}E_a$. One can show that there does not exist a choice of b_{ia} which leads to a shift in the dilaton of the form $\Delta \Phi = \log(z) + \text{const.}$ Hence, in this case, fermionic T-duality produces a shift in the dilaton which necessarily depends on the sphere coordinates θ_{\pm} , ϕ_{\pm} . Then bosonic T-duality along time (and some flat torus directions) cannot undo this shift.

To circumvent this problem, we allow bosonic T-duality to act along a complex Killing vector. To this end we introduce some new coordinates for one of the spheres. These coordinates arise from



Figure 3.1: A summary of the T-dualisation process for $AdS_2 \times S^2 \times T^6$.

the form of the coset element used in $[36]^6$. Since S^2 is a symmetric space it may be described by a coset space G/H where the Lie algebra \mathfrak{g} of G admits a \mathbb{Z}_2 -grading (or automorphism). Then, as a vector space, \mathfrak{g} can be decomposed into a direct sum of graded subspaces⁷

$$\mathfrak{g} = \mathfrak{g}_{(0)} \oplus \mathfrak{g}_{(2)}.$$

In particular we have $S^2 = SU(2)/U(1)$ and

$$\mathfrak{su}(2) = \mathfrak{g}_{(0)} \oplus \mathfrak{g}_{(2)},$$

where

$$\mathfrak{g}_{(0)} = \langle L_+ + L_- \rangle = \langle L_1 \rangle, \quad \mathfrak{g}_{(2)} = \langle L_+ - L_-, L_3 \rangle = \langle L_2, L_3 \rangle.$$

This coset space is parametrized by the coset element

$$g = e^{\lambda_+ L_+} e^{-\lambda_3 L_3} \in \mathfrak{su}(2),$$

⁶The algebra has been rescaled to avoid factors of $c = \sqrt{\alpha}$. Further discussion of related parametrisations occur in [125].

⁷This will be discussed in more detail in the next chapter.

where $L_{\pm} = iL_1 \pm L_2$, $L_n = \frac{1}{2i}\sigma_n$ and the σ_n are the Pauli matrices. Now we introduce the Maurer-Cartan 1-form which takes values in the Lie algebra \mathfrak{g} :

$$J = g^{-1} dg \in \mathfrak{su}(2),$$

with d denoting the exterior derivative. Using the \mathbb{Z}_2 -automorphism we are able to decompose J, as a vector space, into $J = J_{(0)} + J_{(2)}$. Then $J_{(2)}$ is the projection of the Maurer-Cartan current onto $\mathfrak{g}_{(2)}$:

$$J_{(2)} = [g^{-1}dg]_{(2)} = e^{i\lambda_3}d\lambda_+L_2 + d\lambda_3L_3$$

From this the metric is constructed as follows

$$ds_{S^+}^2 = -2\text{Tr}(J_{(2)}J_{(2)}) = d\lambda_3^2 + e^{2i\lambda_3}d\lambda_+^2$$

The reason for constructing the metric using the trace of the Maurer-Cartan form is because it is the only 1-form that, by construction, satisfies the Maurer-Cartan equation. This ensures that our metric possesses and preserves the symmetries associated with the chosen Lie group G.

Comparing this metric to the metric in (3.8) allows us to show that the new coordinates $x^2 = \lambda_3$, $x^3 = \lambda_+$ are related to θ_+ , ϕ_+ as follows

$$e^{i\lambda_3} \equiv \cos\theta_+ + i\sin\theta_+ \cos\phi_+,$$

$$\lambda_+ \equiv e^{-i\lambda_3}\sin\theta_+ \sin\phi_+ = \frac{\sin\phi_+}{\cot\theta_+ + i\cos\phi_+}.$$
(3.19)

In what follows we write $e^3 = e^3_{\mu} dx^{\mu} = \sin \theta_+ d\phi_+$ and so on for the original coordinates. Then, underlined at indices will represent new coordinates: $x^2 = \lambda_3$, $x^3 = \lambda_+$. This change of coordinates preserves the volume form $e^2 \wedge e^3 = e^2 \wedge e^3$. By changing coordinates we change the original real S^2 (with θ_+ , ϕ_+) into a complex S^2 . When T-duality is applied to this exceptional case we need to choose one of the new complex directions, together with the time-like direction and two torus directions, in order to illustrate self-duality. Therefore, we choose to T-dualise along λ_+ as our new complex direction.

The dual metric is given by

$$ds'^{2}_{AdS_{2}} = -z^{2}dt^{2} + \frac{dz^{2}}{z^{2}}, \quad ds'^{2}_{S_{+}} = d\lambda^{2}_{3} + e^{-2i\lambda_{3}}d\lambda^{2}_{+}, \quad (3.20)$$

and a shift in the dilaton

$$\Delta \Phi = \Phi'' - \Phi' = \log\left(ze^{-i\lambda_3}\right) = \log\left(\frac{z}{\cos\theta_+ + i\sin\theta_+\sin\phi_+}\right)$$

Notice that the dual metric in terms of θ_+ , ϕ_+ is no longer real. To recover a sphere we need to choose a new pair of real coordinates. Define θ' , ϕ' by (3.19) with λ_3 replaced by $-\lambda_3$, i.e.

$$e^{i\lambda_3} = \cos\theta' + i\sin\theta'\cos\phi',$$
$$\lambda_+ = \frac{\sin\phi'}{\cot\theta' + i\cos\phi'}.$$

This returns $ds'_{S_+}^2 = d\theta'^2 + \sin^2 \theta' d\phi'$. As before, in order to recover AdS_2 we need to invert the radial coordinate by defining z' = 1/z which gives $ds'_{AdS_2}^2 = (-dt^2 + dz'^2)/z'^2$.

For the fermionic T-duality we choose the following complex combinations

$$\mathcal{E}_1 = E_1 + iE_4, \quad \mathcal{E}_2 = -E_2 + iE_3.$$

The constant Killing spinors ξ_a which we input to get the E_a 's are easy to find (using *Mathematica* for example), but are rather messy to display here. These constant spinors are ordered by the position of the first non-zero component where we have chosen the sign such that the first term is positive:

$$(\xi_a)^{\beta} = 0$$
 for $\beta < a$. $(\xi_a)^{\beta} > 0$ for $\beta = a$.

Then we use \mathcal{E}_1 , \mathcal{E}_2 to calculate the matrix C

$$C = -2i \frac{\cos \theta_+ + i \sin \theta_+ \cos \phi_+}{z} \begin{bmatrix} -e^{-i\phi_-} \sin \theta_- & \cos \theta_- \\ \cos \theta_- & e^{i\phi_-} \sin \theta_- \end{bmatrix}$$

and find the change in the dilaton

$$\Delta \Phi = \Phi' - \Phi = \log\left(\frac{\cos\theta_+ + i\sin\theta_+ \cos\phi_+}{z}\right) + \log\left(-2i\right). \tag{3.21}$$

which cancels the bosonic shift found earlier perfectly (apart from the constant which may be absorbed).

Considering the change in the RR forms, we notice that a 2-form has appeared

$$e^{i\Phi'}F'^{(2)} = \sqrt{1-\alpha} \frac{(\sin\theta_{+} - i\cos\theta_{+}\cos\phi_{+})e^{1} \wedge e^{2} + i\sin\phi_{+}e^{1} \wedge e^{3}}{\cos\theta_{+} + i\sin\theta_{+}\cos\phi_{+}} = -i\sqrt{1-\alpha}e^{1} \wedge e^{2}$$

Rewriting the 2-form in the new underlined at coordinates simplifies the expression greatly. The change in the 4-form $\Delta F^{(4)}$ has real terms which precisely cancel the original $F^{(4)}$, as required. It also has imaginary terms, greatly simplified by the new coordinates, and given by

$$e^{i\Phi'}F'^{(4)} = -i(e^1e^{\underline{3}}e^6e^8 + e^1e^{\underline{3}}e^7e^9) + i\sqrt{\alpha}(-e^0e^{\underline{2}}e^6e^7 + e^0e^{\underline{2}}e^8e^9) + i\sqrt{1-\alpha}(e^0e^{\underline{3}}e^4e^5).$$

Acting with bosonic T-duality on $F'^{(2)}$ and $F'^{(4)}$ in the usual way and along the directions $(t = x^0; \lambda = x^{\underline{3}}, x^7, x^8)$ gives

$$F''^{(2)} = 0,$$

$$F''^{(4)} = (e^0 e^1 e^6 e^7 + e^0 e^1 e^8 e^9) + \sqrt{\alpha} (e'^2 e'^3 e^6 e^8 + e'^2 e'^3 e^7 e^9) + \sqrt{1 - \alpha} (-e^4 e^5 e^7 e^8 + e^4 e^5 e^6 e^9).$$

The new at coordinates e'^2 , etc. are interpreted as being related to the new metric ds'^2 (3.20) created by the bosonic T-duality. Thus $e'^3 = e'^{-i\lambda_3}d\lambda_+ \neq e^3$. Notice that $e'^2 = -d\lambda_+ = -e^2$ so that we preserve the volume form with this second coordinate change to the new coordinates $\theta' = x^{2'}$, $\phi' = x^{3'}$ we write this as $e'^2 \wedge e'^3 = e^{2'} \wedge e^{3'}$ Thereby recovering the original flux $F^{(4)}$ in (3.9).

3.4 $AdS_3 \times M$ Backgrounds

Here we will start with the $AdS_3 \times S^3 \times S^3 \times S^1$ background immediately. The simpler cases follow in much the same way as the simple cases for AdS_2 , and there is nothing new to learn from them as a result. Also, the case $AdS_3 \times S^3 \times T^4$ was studied in [23] in the context of IIB string theory. Hence the focus on the $0 < \alpha < 1$ case here.

The metric is given by

$$ds_{AdS_3}^2 = \frac{-dt^2 + dx^2 + dz^2}{z^2}$$



Figure 3.2: A summary of the T-dualisation process for $AdS_2 \times S^2 \times S^2 \times T^4$.

which has accompanying spin connection components $\omega_{t[02]} = \omega_{x[21]} = 1/z$. Following [?], we use nested coordinates for the S^3

$$ds_{S_{\pm}^{3}}^{2} = d\theta_{\pm}^{2} + \sin^{2}\theta_{\pm}(d\phi_{\pm}^{2} + \sin^{2}\phi_{\pm}d\psi_{\pm}^{2})$$

with spin connection components $\omega_{\phi_+[34]} = -\cos\theta_+$, $\omega_{\psi_+[35]} = -\cos\theta_+\sin\phi_+$ and $\omega_{\psi_+[45]} = -\cos\phi_+$. The flux is given by

as in [121], where the projection operator is

$$P = \frac{1}{2} (1 + \sqrt{\alpha} \Gamma^{012345} + \sqrt{1 - \alpha} \Gamma^{012678}).$$

Notice that for a bosonic T-duality along the x^9 direction, we return to fluxes considered in [127]. Furthermore, taking the limit $\alpha \to 1$ reduces this to the IIB $AdS_3 \times S^3 \times T^4$ case studied by O'Colgain [23]. This background has 16 supersymmetries, giving us 8 Poincaré Killing spinors and therefore 4 complex directions along which we may perform fermionic T-duality. We arrive at a solution $E = z^{-1/2}R_{S_+}R_{S_-}\xi$ in much the same way as we did for the AdS_2 case. For AdS_3 there is of course an extra boundary direction x, thus we have $\partial_t E = 0$ and $\partial_x E = 0$. The $\mu = t$ equation then gives us the constraint $E = \Gamma^{019}PE$, and the $\mu = x$ equation is identical. As before, the $\mu = z$ equation is solved by $E \propto z^{-1/2}$. Using PE = E, for $\mu \in S_+$ we have

$$D_{\mu}E = \frac{\sqrt{\alpha}}{2}\Gamma_{\mu}\Gamma^{01345}E$$

which is solved by a factor $R_{S_+} = e^{\frac{\theta_+}{2}\Gamma^{0145}} e^{\frac{\phi_+}{2}\Gamma^{34}} e^{\frac{\psi_+}{2}\Gamma^{45}}$. Similarly, for $\mu \in S_-$ we get

$$D_{\mu}E = \frac{\sqrt{1-\alpha}}{2}\Gamma_{\mu}\Gamma^{01678}E$$

which is solved by $R_{S_{-}} = e^{\frac{\theta_{-}}{2}\Gamma^{0178}} e^{\frac{\phi_{-}}{2}\Gamma^{67}} e^{\frac{\psi_{-}}{2}\Gamma^{78}}$ alone, since $[\Gamma_{\mu}\Gamma^{01678}, R_{S_{+}}] = [D_{\mu}, R_{S_{+}}] = 0$. The complete solution is

$$E(\xi) = \frac{1}{\sqrt{z}} \exp\left(\frac{\theta_+}{2}\Gamma^{0145}\right) \exp\left(\frac{\phi_+}{2}\Gamma^{34}\right) \exp\left(\frac{\psi_+}{2}\Gamma^{45}\right) \exp\left(\frac{\theta_-}{2}\Gamma^{0178}\right) \exp\left(\frac{\phi_-}{2}\Gamma^{67}\right) \exp\left(\frac{\psi_-}{2}\Gamma^{78}\right) \xi$$

with constant ξ obeying

$$\Gamma^{019}\xi = P\xi = \xi.$$

As mentioned earlier there are 8 of these spinors. We shall organise the orthogonal ξ_a so that

$$(\xi_a)^{\beta} = 0$$
 for $\beta < a$, $(\xi_a)^{\beta} = \frac{1}{2}$ for $\beta = a$

The commutativity relation (3.13) is only non-zero for $\mu = t, x$:

$$V^{t} = -1, \quad V^{x} = -\sqrt{\alpha}\sigma^{2} \otimes \sigma^{2} \otimes 1_{2 \times 2} - \sqrt{1-\alpha} 1_{2 \times 2} \otimes \sigma^{1} \otimes 1_{2 \times 2}.$$

After some trial and error, we arrive at the following complex combinations

$$\mathcal{E}_1 = E_1 - iE_8, \quad \mathcal{E}_2 = E_2 + iE_7, \quad \mathcal{E}_3 = E_3 - iE_6, \quad \mathcal{E}_4 = E_4 + iE_5,$$

This choice leads to

$$C = C_{+}(z, \theta_{+}, \phi_{+}, \psi_{+})C_{-}(\theta_{-}, \phi_{-}, \psi_{-})$$

where

$$C_{+}(z,\theta_{+},\phi_{+},\psi_{+}) = \frac{2i}{z}\sin\theta_{+}(\cos\phi_{+} + i\sin\phi_{+}\cos\psi_{+})$$

is a number, and

$$C_{-} = \begin{bmatrix} -c_{\theta} - is_{\theta}s_{\phi}(\hat{\alpha}c_{\psi} + \sqrt{\alpha}s_{\psi}) & -i\hat{\alpha}c_{\phi}s_{\theta} & -\sqrt{\alpha}c_{\phi}s_{\theta} & s_{\theta}s_{\phi}(\sqrt{\alpha}c_{\psi} - \hat{\alpha}s_{\psi}) \\ -i\hat{\alpha}c_{\phi}s_{\theta} & is_{\theta}s_{\phi}(\hat{\alpha}c_{\psi} - \sqrt{\alpha}s_{\psi}) - c_{\theta} & s_{\theta}s_{\phi}(\sqrt{\alpha}c_{\psi} + \hat{\alpha}s_{\psi}) & \sqrt{\alpha}c_{\phi}s_{\theta} \\ -\sqrt{\alpha}c_{\phi}s_{\theta} & s_{\theta}s_{\phi}(\sqrt{\alpha}c_{\psi} + \hat{\alpha}s_{\psi}) & c_{\theta} + is_{\theta}s_{\phi}(\hat{\alpha}c_{\psi} - \sqrt{\alpha}s_{\psi}) & i\hat{\alpha}c_{\phi}s_{\theta} \\ s_{\theta}s_{\phi}(\sqrt{\alpha}c_{\psi} - \hat{\alpha}s_{\psi}) & \sqrt{\alpha}c_{\phi}s_{\theta} & i\hat{\alpha}c_{\phi}s_{\theta} & c_{\theta} - is_{\theta}s_{\phi}(\hat{\alpha}c_{\psi} + \sqrt{\alpha}s_{\psi}) \end{bmatrix}$$

is a matrix with unit determinant. Here $s_{\theta} = \sin \theta_{-}$, $c_{\theta} = \cos \theta_{-}$, etc. Also, $\hat{\alpha} = \sqrt{1 - \alpha}$. The change in the dilaton is then

$$\Delta \Phi = \Phi' - \Phi = 2\log \frac{2\sin\theta_+(\cos\phi_+ + i\sin\phi_+\cos\psi_+)}{z}.$$
(3.23)

As for the AdS_2 case with two spheres, we need to perform bosonic T-duality along some complex Killing vectors in order to undo the shift in the dilaton. From the parametrisation of the coset element in [36] we arrive at the metric

$$ds^2_{S^3_+} = d\lambda^2_3 + e^{2i\lambda_3} d\lambda^2_+ - e^{2i\lambda_3} d\lambda^2_-.$$

Notice that this is the generalisation of (3.20). Next, dualise along λ_+ , λ_- and two AdS_3 directions t and x. Doing so gives us a shift in the dilaton of

$$\Delta \Phi = \Phi'' - \Phi' = 2\log(ze^{-i\lambda_3})$$

from which we read off

$$e^{i\lambda_3} = \sin\theta_+(\cos\phi_+ + i\sin\phi_+\cos\psi_+). \tag{3.24}$$

This duality changes the metric $ds_{S_{+}^{3}}^{2}$ to $ds_{S_{+}^{3}}^{\prime 2} = d\lambda_{3}^{2} + e^{-2i\lambda_{3}}d\lambda_{+}^{2} - e^{2i\lambda_{3}}d\lambda_{-}^{2}$. The dual metric is no longer real when expressed in the old coordinates θ_{+} , etc. So, just as before, in order to recover a sphere

we must choose a different real slice given by θ' , ϕ' and ψ' which are defined by (3.24) with λ_3 replaced by $-\lambda_3$. That is, $e^{-i\lambda_3} = \sin \theta' (\cos \phi' + i \sin \phi' \cos \psi')$.

Finally, we consider the changes made to the RR fields. The fermionic T-duality produces a 2-form flux which, as for the AdS_2 case, simplifies when written in the new complex coordinates. Again, using underlined flat indices for the new coordinates, we write $e^3 = d\lambda_3$. Then

$$\Delta F^{(2)} = 2\sqrt{1-\alpha} \left[\cot\theta_{+}e^{2}e^{3} + \frac{i(\cos\phi_{+}\cos\psi_{+} + i\sin\phi_{+})e^{2}e^{4} - i\sin\psi_{+}e^{2}e^{5}}{\sin\theta_{+}(\cos\phi_{+} + i\sin\phi_{+}\cos\psi_{+})} \right]$$

= $2i\sqrt{1-\alpha}e^{2}e^{3}.$ (3.25)

For the 4-form flux

$$\begin{split} \Delta F^{(4)} &= -2\sqrt{\alpha}e^{0}e^{1}e^{9}\left[\cot\theta_{+}e^{3} + \frac{i(\cos\phi_{+}\cos\psi_{+} + i\sin\phi_{+})e^{4} - i\sin\psi_{+}e^{5}}{\sin\theta_{+}(\cos\phi_{+} + i\sin\phi_{+}\cos\psi_{+})}\right] \\ &+ 2\sqrt{\alpha}e^{3}e^{4}e^{5}e^{9} + 2\sqrt{1-\alpha}e^{6}e^{7}e^{8}e^{9} - 2e^{0}e^{1}e^{2}e^{9} \\ &+ 2e^{2}e^{9}\left[\cot\theta_{+}e^{4}e^{5} + \frac{-i(\cos\phi_{+}\cos\psi_{+} + i\sin\phi_{+})e^{3}e^{5} - i\sin\psi_{+}e^{3}e^{4}}{\sin\theta_{+}(\cos\phi_{+} + i\sin\phi_{+}\cos\psi_{+})}\right] \end{split}$$

Notice that the terms on the second line exactly cancel the original $F^{(4)}$. The first line contains e^3 , given by the square bracket, as in (3.25). The square bracket on the third line is just ie^4e^5 , this is seen by noting that the volume form for the underlined coordinates should be the same as for the original coordinates, i.e.

$$[ie^{\underline{3}}] \wedge [ie^{\underline{4}} \wedge e^{\underline{5}}] = (\cot^2 \theta_+ + \ldots)e^3 \wedge e^4 \wedge e^5 = -e^3 \wedge e^4 \wedge e^5.$$

Then, after the fermionic T-duality, the RR fluxes are

$$e^{\Phi'}F^{(2)'} = 2i\sqrt{1-\alpha}e^2e^3 \tag{3.26}$$

$$e^{\Phi'}F^{(4)'} = 2i\sqrt{\alpha}e^0e^1e^3e^9 + 2\sqrt{1-\alpha}e^6e^7e^8e^9.$$
(3.27)

To get back to the original flux we bosonic T-dualise along t, x and λ_{\pm} (i.e. directions 0, 1, 4, 5) which leads to

$$e^{\Phi^{\prime\prime}}F^{(4)^{\prime\prime}} = -2e^{0}e^{1}e^{2}e^{9} + 2\sqrt{\alpha}e^{'\underline{3}}e^{'\underline{4}}e^{'\underline{5}}e^{9} + 2\sqrt{1-\alpha}e^{6}e^{7}e^{8}e^{9}.$$

Finally, we use that $e^{\underline{3}}e^{\underline{4}}e^{\underline{5}} = e^{\underline{3}'}e^{\underline{4}'}e^{\underline{5}'}$ to write $F^{(4)''}$ in terms of the final set of real coordinates θ' , ϕ' and ψ' . Thus, recovering the original RR flux, (3.22).



Figure 3.3: A summary of the T-dualisation process for $AdS_3 \times S^3 \times S^3 \times T^1$.

3.5 Combined Bosonic and Fermionic T-duality: In General

This section deals with proving the invariance of $AdS_n \times S^n \times M^{10-2n}$ backgrounds (and their exceptional extensions) under combined bosonic and fermionic T-duality without using the σ -model approach. In other words, this section generalises the treatment of these backgrounds, dealt with earlier on in this chapter, by applying T-duality transformations directly to the supergravity fields. We consider these geometries whilst being supported by various RR fluxes. This discussion extends the earlier results in [16, 22, 23, 101, 107] to the whole class of $AdS_n \times S^n \times M^{10-2n}$ super-backgrounds supported by RR fluxes.

3.5.1 Rules for Fermionic T-duality

Killing Spinors

The component supergravity fields along the bosonic directions are T-dualised using the original Buscher rules [12, 13, 129]. Generalising these rules to fermionic directions was given in [22]. Fermionic T-duality acts on the dilaton $\Phi(X)$ and the RR fields (which are *p*-forms) whilst leaving the metric and NSNS 2-form B flux invariant (see Chapter 2 for more details). In the fermionic case we dualise along directions specified by Grassmann-even⁸ Killing spinors, $\Xi_{\mu}(X)$ (where μ labels their number), that generate Abelian superisometries. This means that the Killing spinors need to satisfy an additional condition called the commutativity condition

⁸Grassmann algebras are examples of supercommutative algebras. These algebras may be decomposed into even and odd variables which satisfy a graded version of commutativity. In particular, even elements commute whilst odd elements anticommute.

$$\Xi_{\mu}\Gamma_{A}\Xi_{\nu} = 0$$
 for all A, μ, ν with $A = 0, 1, ..., 9.$ (3.28)

This condition has important consequences for complex Killing spinors; the only type of spinor for which non-trivial solutions may be obtained. As a result, they are associated with complex Grassmann-odd directions in superspace. The Killing spinor equations have the following form

The Killing spinor equations in (3.29) are obtained by requiring that the supersymmetry variations of the gravitino and dilatino vanish. They are also determined by the geometry of the background and the chosen RR fluxes (see Chapter 2 for a detailed discussion). The second equation in (3.29) is used together with the integrability requirement for the first equation in (3.29) to obtain a projector $\mathcal{P}_{8(n-1)}$, singling out the 8(n-1) fermionic isometries of the backgrounds of interest. Once this projector has been found, the second equation in (3.29) is identically satisfied.

Fermionic T-duality Rules

As noted in Chapter 2, the main object of interest when performing a fermionic T-duality is the scalar field C which is used to calculate the changes in the fields. We arrive at this matrix C once we have solved the Killing spinor equations which yield symmetry preserving Killing spinors. Then, after making complex combinations of these spinors such that they satisfy (3.28), we are able to construct C from

$$\partial_M C_{\mu\nu} = \begin{cases} E_M^A \bar{\Xi}_\mu \Gamma_A \Gamma^{11} \Xi_\nu & \text{type IIA} \\ E_M^A \bar{\Xi}_\mu \Gamma_A \sigma^3 \Xi_\nu & \text{type IIB} \end{cases}$$

The matrix $C = C_{\mu\nu}(X)$ is formed by the components of the NSNS 2-form B field along the Abelian fermionic isometries (i.e. $\Theta = 0$) [22]

$$d\theta^{\mu} \wedge d\theta^{\nu} B_{\mu\nu}(X,\Theta)|_{\Theta=0} := d\theta^{\mu} \wedge d\theta^{\nu} C_{\mu\nu}(X), \qquad (3.30)$$

where C depends on the bosonic directions X only. Once we have obtained C then we can find the dilaton as follows

$$\Delta \Phi = \Phi' - \Phi = \frac{1}{2} \log \left(\det C \right), \tag{3.31}$$

and the change in the RR fluxes

where Γ is a certain product of Γ -matrices which we use to split the fermionic $E^{(1,2)}$ currents into four pieces. This corresponds to splitting the superalgebra generator Q into Q, \hat{Q}, S and \hat{S} respectively.

Background	RR Flux	Γ
$AdS_5 \times S^5$	F_5 flux	1
$AdS_3 \times S^3 \times T^4$	F_3 flux	$-\Gamma^{23}$
$AdS_3 \times S^3 \times S^3 \times S^1$	F_3 flux	$-\Gamma^{23}$
$AdS_3 \times S^3 \times S^3 \times S^1$	F_4 flux	Γ^{239}
$AdS_2 \times S^2 \times T^6$	F_3 flux	$-\Gamma^{23}$
$AdS_2 \times S^2 \times T^6$	$F_2 \& F_4$ fluxes	$\Gamma^{11}\Gamma^{123}$
$AdS_2 \times S^2 \times T^6$	F_4 flux	Γ^1
$AdS_2 \times S^2 \times S^2 \times T^4$	F_4 flux	Γ^{239}

Table 3.2: Table displaying the Γ corresponding to each background and their various fluxes.

Explicit Form of the Killing Spinors

Berkovits and Maldacena describe a direct way to find the explicit form of the Killing spinors associated with the anti-commuting fermionic isometries along which we dualise the supergravity fields [22]. This method suggests extracting the form of the Killing spinors directly from the corresponding components of the fermionic currents J_Q associated with the generators Q of the superisometry algebra. By definition, the Killing spinors must satisfy the commutativity condition (3.28) and the Killing spinor equations (3.29). Concretely, the Killing spinors are given by the components of the matrix $J^{\beta}_{\alpha}(|y|, y^{\hat{\alpha}}, \lambda_3)$ in

$$J_{Q_{\alpha}}|_{\Theta=0} = d\theta^{\mu}J^{\alpha}_{\mu}(|y|, y, \lambda_3) = d\theta^{\mu}e^{-B}Q_{\mu}e^{B}|_{Q_{\alpha,\hat{\theta}=\hat{\xi}}=0} = d\theta^{\mu}\Xi^{\alpha}_{\mu},$$
(3.33)

where

$$e^B := e^{\hat{\theta}^{\alpha}\hat{Q}_{\alpha} + \hat{\xi}^{\alpha}\hat{S}_{\alpha}} |y|^D e^{-\lambda_3 L_3} e^{-\rho^{\beta}_{\alpha} R^{\alpha}_{\beta}}.$$
(3.34)

Note that $\Theta = 0$ implies that we consider components proportional to the generators of our isometries. The commutativity condition is obtained by differentiating (3.33) (using the exterior derivative) to get

$$0 = d\Xi_{\mu} + [e^{-B}de^{B}, \Xi_{\mu}\mu]|_{\Theta=0},$$
$$e^{-B}de^{B}|_{\Theta=0} = \Omega^{\hat{a}\hat{b}}(y/|y|)R_{\hat{a}\hat{b}} + J_{D}(|y|)D + J_{L_{3}}(\lambda_{3})L_{3}$$

Note that the index μ labels a given Killing spinor. From the structure of e^B (3.34) and the commutation relations

$$[D,Q] = \frac{1}{2}Q,$$
$$[R_{\hat{a}},Q] = -\frac{s^2}{2}Q\Gamma_{\hat{a}}\Gamma^4\mathbb{P},$$
$$[L_3,Q] = \frac{i}{2}Q,$$

we have the following explicit form of the Killing spinors in question⁹

$$\Xi^{\alpha}_{\mu} = J^{\alpha}_{\mu}(|y|, y, \lambda_3) = |y|^{-\frac{1}{2}} e^{\frac{i}{2}c\lambda_3} \mathcal{O}^{\alpha}_{\mu}(y^{\hat{a}}/|y|), \qquad (3.35)$$

where $\mathcal{O}^{\alpha}_{\mu}(y^{\hat{a}}/|y|) := (e^{s\mathbb{P}\Gamma_{\hat{a}}\Gamma_{4}y^{\hat{a}}/(2|y|)})^{\alpha}_{\mu}$ is a *Spin* (n+1) matrix associated with the coset $s^{n} \cong SO(n+1)/SO(n)$ and $\mathbb{P} := \mathbb{P}_{+}\mathcal{P}_{8(n-1)}$ is the projection matrix that singles out the 2(n-1) anti-commuting isometries $Q = Q\mathbb{P}$ for each case of $AdS_{n} \times S^{n} \times M^{10-2n}$. The projector \mathbb{P}_{\pm} is given by $P_{\pm} := \frac{1}{2}(1 \pm i\Gamma^{0123})$. By definition,

$$\mathcal{O}^T \Gamma_4 \mathcal{O} = \Gamma_4 \mathbb{P}. \tag{3.36}$$

The structure of the scalar field $C_{\mu\nu}$ can be read off from (3.30) to give

$$B_{\mu\nu}|_{\Theta=0} = i J^{\gamma}_{\mu} \Gamma^4_{\gamma\delta} J^{\delta}_{\nu}(|y|, y, \lambda_3) = C_{\mu\nu}.$$

The explicit form of the matrix C can be found using (3.35) and (3.36). Then,

⁹The coset representative for $AdS_2 \times S^2 \times T^6$ is obtained from the representative for $AdS_2 \times S^2 \times S^2 \times T^4$ by taking the limit $c \to 0$. To do this, rescale the coordinates $\lambda \to c\lambda$, $\lambda_3 \to c\lambda_3$, and $\rho_{\beta}^{\alpha} \to s\rho_{\beta}^{\alpha}$ finally taking the limit $c \to 0$. Performing this limit makes the metric of the second sphere become flat. This decouples the sphere from the $AdS_2 \times S^2$ subspace. As a result, it re-compactifies into a T^2 which then forms the required T^6 for the $AdS_2 \times S^2 \times T^6$ background.

$$C_{\mu\nu} = i J^{\gamma}_{\mu} \Gamma^{4}_{\gamma\delta} J^{\delta}_{\nu},$$

$$= i \left(|y|^{-\frac{1}{2}} e^{\frac{i}{2}c\lambda_{3}} \mathcal{O}^{\gamma}_{\mu} \right) \Gamma^{4}_{\gamma\delta} \left(|y|^{-\frac{1}{2}} e^{\frac{i}{2}c\lambda_{3}} \mathcal{O}^{\delta}_{\nu} \right),$$

$$= i |y|^{-1} e^{ic\lambda_{3}} (\mathcal{O}\Gamma^{4}\mathcal{O})_{\mu\nu},$$

$$= i |y|^{-1} e^{ic\lambda_{3}} (\Gamma^{4}\mathbb{P})_{\mu\nu},$$
(3.37)

and the inverse is given by

$$(C^{-1})^{\mu\nu} = -i|y|^{-1}e^{ic\lambda_3}(\mathbb{P}\Gamma^4)^{\mu\nu}.$$
(3.38)

From (3.37), the dilaton shift is

$$\Delta \Phi = \frac{1}{2} \log (\det C) = -(n-1) \log |y| + i(n-1)c\lambda_3.$$
(3.39)

Notice that this dilaton shift has the exact form of all the cases studied so far. The change in the RR fields is found using (3.38)

in terms of Γ . Thus, (3.40) is the change in RR fields after a fermionic T-duality transformation for any of the cases dealt with thus far.

Explicit Form of the RR Fluxes

Here we consider the change in the RR fluxes (3.40) supporting the exceptional backgrounds, $AdS_n \times S^n \times S^n \times M^{10-3n}$. For n = 3, we have $AdS_3 \times S^3 \times S^3 \times S^1$ and we may consider the type IIB theory supported by an F_3 flux¹⁰

$$F_{3} = \frac{1}{3} (\varepsilon_{cba} e^{a} \wedge e^{b} \wedge e^{c} + \frac{R_{AdS}}{R_{+}} \varepsilon_{\hat{c}\hat{b}\hat{a}} e^{\hat{a}} \wedge e^{\hat{b}} \wedge e^{\hat{c}} + \frac{R_{AdS}}{R_{-}} \varepsilon_{c'b'a'} e^{a'} \wedge e^{b'} \wedge e^{c'}),$$

In this case $\Gamma = -\Gamma^{23}$ as in the non-exceptional ($\alpha = 0$) case. This supergravity background arises due to the intersection of D1-branes and D5-branes [130]. The solution preserves 16 supersymmetries. If instead we chose to dualise along the ϕ^9 -direction (or the S^1 -coordinate), then we would arrive at a type IIA background supported by an F_4 flux [121]

¹⁰The radii for each subspace have the following values: $R_{AdS} = 1$, $R_{+} = 1/\sqrt{\alpha}$, and $R_{-} = 1/\sqrt{1-\alpha}$.

$$F_4 = d\phi^9 \wedge \frac{1}{3} (\varepsilon_{cba} e^a \wedge e^b \wedge e^c + \frac{R_{AdS}}{R_+} \varepsilon_{\hat{c}\hat{b}\hat{a}} e^{\hat{a}} \wedge e^{\hat{b}} \wedge e^{\hat{c}} + \frac{R_{AdS}}{R_-} \varepsilon_{c'b'a'} e^{a'} \wedge e^{b'} \wedge e^{c'}).$$

where $\Gamma = \Gamma^{239}$, and we use the projector \mathcal{P}_{16} . This supergravity background arises due to the dimensional reduction of the eleven-dimensional $AdS_3 \times S^3 \times S^3 \times T^2$ solution and it represents the near-horizon geometry of two *M*5-branes and an *M*2-brane which intersect over a line [131].

For n = 2, the exceptional case has the same F_4 flux as the $\alpha = 0$ case. Here, we have a rank-8 projector which may be decomposed into a product of two rank-16 projectors as in [121], i.e. $\mathcal{P}_8 = P_1 P_2$. This solution is obtained by dimensional reduction of the eleven-dimensional $AdS_2 \times S^2 \times S^2 \times T^5$ solution and it represents the near-horizon geometry for the intersection of two M2-branes and four M5-branes [132]. Re-numbering the F_4 flux components in accordance with [121] such that 0, ..., 3 represents the directions along which we dualise (with 2, 3, 8 and 9 the T^4 directions). Parametrize one of the spheres by $x^7 = \lambda_3$ and $x^1 = \lambda_+$ and the second sphere by x^5 and x^6 , then we can write

$$I_{4} = 4P_1P_2\Gamma^{0492},$$

with

$$P_1 = \frac{1}{2}(1 + \Gamma^{9238}), \quad P_2 = \frac{1}{2}(1 + \sqrt{\alpha}\Gamma^{047123}\sqrt{1 - \alpha}\Gamma^{045698}).$$

where $\Gamma = \Gamma^{239}$. In all the cases considered here, the shifts (3.39) and (3.40) are undone, precisely, by the corresponding bosonic T-duality.

3.5.2 Compensating Bosonic T-duality

In Chapter 1, we discussed how the complete Buscher rules for the bosonic case form part of a larger O(D, D) symmetry group. It is this group that forms the basis for generalized geometry [133]. In the cases that we have considered, the antisymmetric NSNS B field vanishes and the metric is diagonal, this greatly simplifies the rules. Let \mathcal{I} be the set of directions along which we dualise, the new metric has the following components

$$G_{tt}' = \frac{1}{G_{tt}}, \quad t \in \mathcal{I}$$

and remains unchanged in all other directions. The dilaton shift is given by the determinant of this block of the matrix

$$\Delta \Phi = -\frac{1}{2} \log \left(\det G_{MN} \right) = -\frac{1}{2} \sum_{t \in \mathcal{I}} \log G_{tt}.$$

Finally, we can write down the change in the RR fields as

$$c_t := \begin{cases} -i & t = 0\\ 1 & t \neq 0. \end{cases}$$

Note that for a non-trivial dilaton extra factors need to be included, this was dealt with by Fukuma et al. [81]. Time-like T-duality results in imaginary forms [83], however, the overall sign appearing after performing combined T-duality is not physical. It only depends on the order in which we apply our dualities.

Concerning the backgrounds under consideration, the bosonic T-dualisation was performed along the directions labelled by $t = \{0, ..., 3\}$. The T-dualised RR fields are then

$$\mathbf{F}'' = -i\Gamma^{0123}\mathbf{F}'.$$

3.6 Summary

This chapter considered type II supergravity backgrounds $AdS_2 \times S^2 \times T^6$ and $AdS_d \times S^d_+ \times S^d_- \times T^{10-3d}$ for d = 2, 3, including a general treatment of all the less supersymmetric backgrounds simultaneously. These backgrounds are interesting because they are not maximally supersymmetric, unlike $AdS_5 \times S^5$, which has been widely studied. Furthermore, their self-duality has not been thoroughly studied in the supergravity context. These backgrounds have, however, been studied (in various ways) using coset σ -models. We have shown that these non-maximally supersymmetric backgrounds are self-dual under a series of bosonic and fermionic T-dualities. It is important to stress that it is not possible to avoid performing bosonic T-duality along some of the torus directions, and for the exceptional cases (i.e. $S_+ \times S_-$) we have to bosonic T-dualise along some complex Killing vectors along one of the spheres. Furthermore, these self-dual backgrounds are integrable. This is a useful fact as it provides evidence favouring the proposed link between integrability and self-duality.

Part III

Green-Schwarz σ -model

Introduction to Part III

Part III contains the body of original research focusing on the σ -model perspective. The goal is to investigate whether backgrounds possessing less than maximal supersymmetry are self-dual under a sequence of bosonic and fermionic T-dualities. Two ideas are present in the part. In particular, we consider the AdS backgrounds of the form $AdS_d \times S^d \times T^{10-2d}$, (d = 2, 3) and $AdS_d \times S^d \times S^d \times T^{10-3d}$, (d = 2, 3). Then we prove that $AdS_5 \times S^5$ is self-dual without fixing κ -symmetry gauge. To understand where κ -symmetry plays a role, it is helpful to recall the bosonic string worldsheet. There, we quantize the fields X^{μ} using oscillators that satisfy the following commutation relations¹¹

$$[a_k^{\mu}, a_p^{\dagger \nu}] = \delta_{kp} \eta^{\mu \nu}.$$

Notice that the zero modes (i.e. $\mu = \nu = 0$) are negative norm states. These states are undesirable and can be removed through use of the gauge symmetry. Conformal symmetry is part of the gauge symmetry which we use to remove negative norm states. Moving onto the superstring, we find that negative norm fermionic states arise as well. To remove these negative norm states we need a larger gauge symmetry (as there are more degrees of freedom). This gauge symmetry is called κ -symmetry and it removes half of the fermionic degrees of freedom¹². This new symmetry plays an important role, as we will see, when studying the self-duality of backgrounds that may be written as a graded coset structure (i.e. the supersymmetric cousin of a bosonic symmetric space [36]). The study of $AdS_5 \times S^5$ was greatly facilitated by the observation that the Green-Schwarz action for this background could be written as a \mathbb{Z}_4 -graded coset¹³ [126, 138]

$\frac{PSU(2,2|4)}{SO(1,4) \times SO(5)}.$

The motivation for studying string theory directly, using worldsheet methods, arose shortly after the AdS/CFT correspondence between type IIB string theory on $AdS_5 \times S^5$ and $\mathcal{N} = 4$, D = 4 super Yang-Mills theory [6,139–142] was described. This provided an opportunity to prove that $AdS_5 \times S^5$ was an exact string solution, to write down the corresponding 2-dimensional conformal field theory and

¹¹Here we assume a mostly plus metric.

¹²The existence of κ -symmetry first appeared in superparticle actions in [134–137]. Kappa-symmetry is a gauge symmetry, larger than the conformal group, which is used to remove negative norm bosonic states. Then, κ -symmetry removes negative norm fermionic states. However, the term κ -symmetry was not used in the original references, being introduced later by Townsend [108].

¹³The special projective unitary group PSU(N) is the isometry group of a complex projective space, just as the projective orthogonal group is the isometry group for real projective spaces.

to find the full string spectrum [138]. This idea (of using coset structures) has been used heavily in integrability [143] and attempts to understand fermionic T-duality, for example: Beisert, Ricci, Tseytlin and Wolf [21] work entirely with the coset action, gauge fixing a particular κ -symmetry at the start. They found that the dual σ -model amounts to a different choice of κ -symmetry gauge [36].

The two formulations both have their strengths and weaknesses. The main advantage of using the RNS formalism is that quantization becomes straightforward since the action is free¹⁴ in a flat background [118,118]. There are three important disadvantages. Firstly, the RNS formalism cannot describe superstrings on backgrounds with RR fields present because the RR vertex operators become complicated. Secondly, the symmetry of the spectrum is not manifest. Finally, the extension of the RNS formalism to curved spacetimes is not obvious because there is a lack of spacetime covariance [108]. The GS formalism does allow a covariant extension to curved backgrounds through the existence of κ -symmetry. Furthermore, the GS superstring action may be defined classically in any background satisfying the supergravity equations of motion [118]. For background fields satisfying these equations of motion, the GS action is classically invariant under κ -symmetry, which is required for the removal of non-physical fermionic degrees of freedom. The main disadvantage of the GS formalism is that quantization becomes nontrivial [145, 146]. It is important to stress that in the GS formalism κ -symmetry invariance requires that the background fields are on-shell¹⁵, whereas for the RNS formalism it is quantum Weyl invariance that ensures this self-consistency condition [108]. This is illustrated in Figure 3.4.



Figure 3.4: Different superstring formulations require curved backgrounds to be on-shell [108].

Consider, for the moment, the maximally supersymmetric background $AdS_5 \times S^5$ which is supported by a (Hodge) self-dual RR 5-form flux. As a result, following the discussion above, the RNS formalism is not useful in a straightforward way¹⁶. The GS formalism, being manifestly supersymmetric, appears to be sufficient for non-vanishing RR fields [115]. Although the formal expression for the GS superstring action (in superspace) for a generic type IIB background was presented in [116], it is not very practical for finding the explicit form of the superstring action in terms of the coordinate fields $(X, \Theta)^{17}$. The maximal supersymmetry of the $AdS_5 \times S^5$ background suggests that an alternative approach is needed, one which

¹⁴By free we mean that the action does not possess any interaction terms.

¹⁵By on-shell we mean that the supergravity equations of motion must be obeyed. Since supersymmetry relates bosons to fermions, it means that you can turn any bosonic intial condition into a fermionic intial condition, implying equality of the number of degrees of freedom for the bosonic and fermionic fields.

¹⁶Although the non-local RR vertex operator is known in flat space [114], it is insufficient for describing the complete form of the RNS string action in curved backgrounds

¹⁷For a bosonic background, the corresponding D = 10 type IIB superfields need to be determined explicitly. This is a very complicated problem, not yet solved for any non-trivial cases.
exploits the special properties of this solution [117]. Over a decade ago, such an alternative approach to constructing superstring actions was developed, which combines the advantages from the RNS and GS formalisms [118] (see also [119, 120]). Following the RNS approach, our action reduces to a free action for a flat background in which quantization becomes straightforward. This hybrid approach uses spacetime spinor variables as fundamental fields, as in the GS formalism, allowing for simple RR vertex operators [118]. The hybrid formulation is not without disadvantages either, namely, 10-dimensional Lorentz invariance is no longer manifest [118]. For an alternative approach, see also the *Pure Spinor Formalism* by Berkovits [171].

Chapter 4 uses a general approach, where we verify the self-duality of Green-Schwarz supercoset σ -model $AdS_5 \times S^5$ without gauge fixing κ -symmetry. However, we start with a very general introduction to the most important concepts used throughout Chapter 4 and 5. Furthermore, in Chapter 5, we also consider superstrings on the exceptional backgrounds $AdS_d \times S^d \times S^d$ for (d = 2, 3).

Chapter 4

Supercoset Models

4.1 Introduction

This chapter introduces the techniques used to show T-self-duality using coset σ -models. We begin with a very gentle introduction to the coset geometry of the S^3 manifold. This allows one to introduce graded Lie algebras, coset representatives and the currents that will build our Lagrangians. Simple examples are provided to help expose concepts and permit intuition. Then, we provide a more formal introduction to supercoset models and the general setup we will be using, including the T-duality procedure. Finally we consider the example of $AdS_5 \times S^5$, an exciting example with important consequences for self-duality.

4.2 The Coset Geometry of S^3

As an example, we consider the coset geometry of the 3-sphere, S^3 , as a warm up which will lay bare the techniques used throughout this chapter. Since the rest of the chapter is technically dense and very messy to write down, this section will also serve an introduction to the methods used and will provide some intuition. The backgrounds considered in Chapter 3 are called symmetric or at the very least semi-symmetric spaces. Spaces of this nature may be described by a coset

$$\frac{G}{H} := \{gH|g \in G\},\$$

where G has the interpretation of being the isometry group and H is the isotropy group¹. The key idea is that any Lie group may be described by a manifold. The elements of the corresponding Lie algebra (i.e. the generators of infinitesimal transformations) are the tangent space elements which generate transformations along the flat directions (i.e. in the tangent plane)². In the end, the curved coordinates

¹The isometry group captures all symmetries for which the given metric space will be invariant. The isotropy group forms a subgroup of G that fixes a given point on H. The group is given by $H = \{g \in G | gh = h\}$.

²Roughly speaking, the group H is "divided" out, meaning that the symmetries of the group H do not affect the overall geometry of the S^3 and therefore the metric.

on the manifold are related to the flat coordinates on the tangent space by the *vielbein*:

$$dy^a=e_{\mu}^{\ a}dx^{\mu},$$

where the vielbein components are e_{μ}^{a} . The Latin indices represent tangent space directions whilst the Greek indices represent curved directions. The 3-sphere, S^{3} , is a symmetric space and may be described by the following coset³:

$$S^3 = \frac{SO(4)}{SO(3)}$$

This means that $g \sim gh$ (i.e. they are equivalent) for all $h \in SO(3)$ and $g \in SO(4)/SO(3)$. Points on S^3 correspond to equivalence classes of the coset SO(4)/SO(3), and not to the elements g. However, we may also represent each point on S^3 by an SO(4) vector. Transformations in the SO(4) group rotate points on the S^3 to other points on the S^3 . One might ask why the 3-sphere is not fully described by the SO(4) group. The reason is that, at each point on the curved manifold S^3 , there exists a tangent space. The tangent space at point p on S^3 consists of all rotations about p, which are described by SO(4) vectors. Therefore, there is an entire set of rotations, in the tangent plane at any p, which leaves the vector at that point unchanged with respect to the global S^3 geometry. This is why we need to remove the degrees of freedom which are equivalent to rotations in the tangent space, to remove redundancy, and accurately capture the geometry of the 3-sphere. Cosets are the objects which allow us to do precisely this: they allow us to remove unwanted degrees of freedom. Furthermore, this is the reason that points on the 3-sphere are described by equivalence classes. Note that the 3-sphere is invariant under the action of the SO(4) group [147].

The 3-sphere may be parametrized by a 4-vector, (σ, \vec{v}) . The scalar fields, (σ, \vec{v}) , transform as an SO(4) 4-vector and are subject to the constraint [147]

$$\sigma^2 + \vec{v} \cdot \vec{v} = R^2.$$

In many cases, however, it is more useful to define a set of boosts (which are SO(4) transformations), $L_{\vec{v}}$, which send a given reference point $(\sigma, \vec{v}) = (R, \vec{0})$ into a general point. To illustrate this action explicitly, we choose a representation of the SO(4) group, writing the transformations as 4×4 orthogonal matrices: $O_{\alpha\beta}(p)$ for all $p \in SO(4)$. This is given by

$$\sigma = O_{44}(L_{\vec{v}})R$$
$$v_i = O_{i4}(L_{\vec{v}})R,$$

where are indices run over (4, 1, 2, 3) with i = 1, 2, 3 only. As an example, consider spherical polar coordinates. Then the point $(\sigma, \vec{v}) = (R, \vec{0})$ gets transformed into

³In general, any *n*-sphere can be written as $S^n = SO(n+1)/SO(n)$.

$$\begin{bmatrix} R\cos\theta\\ R\sin\theta\cos\phi\\ R\sin\theta\sin\phi\cos\psi\\ R\sin\theta\phi\sin\phi\psi \end{bmatrix} = L_{\vec{v}} \begin{bmatrix} R\\ 0\\ 0\\ 0\\ 0 \end{bmatrix}$$

when we choose $L_{\vec{v}}$ to be

$$L_{\vec{v}} = e^{-i\psi L_{23}} e^{-i\phi L_{12}} e^{-i\theta L_{41}}.$$
(4.1)

It must be noted that there is an arbitrariness in the choice of the operators $L_{\vec{v}}^4$. In this example we used spherical polar coordinates, but some other transformation may be used, the important feature is that any choice of operator must satisfy the algebra

$$[L_{\alpha\beta}, L_{\gamma\delta}] = \delta_{\beta\gamma} L_{\alpha\rho} - \delta_{\alpha\gamma} L_{\beta\rho} + \delta_{\alpha\rho} L_{\beta\gamma} - \delta_{\beta\rho} L_{\alpha\gamma},$$

where the operator $L_{\alpha\beta} = -L_{\beta\alpha}$ is the generator of infinitesimal rotations in the $\alpha\beta$ -plane. Notice that the coset representative (4.1) contains the generator L_4 1. This might seem odd at first since, for the 2-sphere, the infinitesimal generators were in the planes perpendicular to the chosen direction. Here we have fixed the direction 4 and so we might expect to have planes orthogonal to the 4-direction. But two things are happening. Firstly, in 4-dimensions orthogonality does not look the same as it does in 3-dimensions (i.e. our intuition may get us into trouble). Secondly, we might have chosen the following transformation

$$\begin{bmatrix} R \\ R\sin\theta\cos\phi \\ R\sin\theta\sin\phi\cos\psi \\ R\sin\theta\sin\phi\sin\psi \end{bmatrix} = L_{\vec{v}} \begin{bmatrix} R \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix}$$

Notice that the 4-direction remained invariant. For this case we would certainly have a very different $L_{\vec{v}}$ as compared to (4.1). However, given that our choice transforms our 4-vector to a general 4-vector where the 4-direction depends on one of the SO(3) coordinates θ , it is therefore, not unusual that it appears in (4.1) alongside θ . There is another instance where arbitrariness arises. In four dimensions we have the following planes

41	42	43
14	12	13
24	21	23
<i>3</i> 4	31	3 2,

⁴This point is often glossed over in the main body of the literature. Another way make this point clear is to notice that, choosing a coset representative is not unique. There is always some arbitrariness.

where the crossed out planes are repetitions (i.e. $14 \leftrightarrow 41$). From the third row only plane-23 remains, so it appears in (4.1). The second row has two options, the 12-plane or 13-plane. Only one of these need to be chosen, either choice will work and we have chosen the 12-plane. The first row presents the most options. Here we have the 4*i*-planes (i = 1, 2, 3) to choose from. Again, either choice will satisfy the algebra. We have chosen the 41-plane.

Explicitly, an SO(4) transformation may be written as

$$e^{-i\theta L_{41}} = \begin{bmatrix} \cos\theta & 0 & 0 & \sin\theta \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ -\sin\theta & 0 & 0 & \cos\theta \end{bmatrix}$$

Applying an SO(4) transformation, T, to the 3-sphere will carry the point with coordinates (θ, ϕ, ψ) into the point (θ', ϕ', ψ') . In general, the coordinate transformation is given by

$$O_{\alpha4}(L'_{\vec{v}}) = O_{\alpha\beta}(T)O_{\beta4}(L_{\vec{a}}) = O_{\alpha4}, (TL_{\vec{v}})$$

where $O_{\alpha\beta}$ is a 4×4 orthogonal matrix and $O_{\alpha4}$ and $O_{\beta4}$ are column vectors. For an orthogonal transformation in the 123-subspace (i.e. $S \in SO(3)$), we have

$$O_{\alpha 4}(S) = \delta_{alpha4}.$$

As a result, we may write

$$O_{\alpha 4}(L'_{\vec{v}}S) = O_{\alpha 4}(gL'_{\vec{v}}),$$

which implies that $gL_{\vec{v}} = L'_{\vec{v}}S$ (because $gL_{\vec{v}}$ and $L'_{\vec{v}}S$ have the same representation). Since we can find $L'_{\vec{v}}$ once we know what (θ', ϕ', ψ') are, the SO(3) transformation

$$S = L_{\vec{v}}^{\prime-1}TL_{\vec{v}},$$

becomes well defined [147]. The coset formulation gives a well defined geometric description from which geometrical objects, like the metric and vielbeins, may be found using group theory.

To compute the basis vectors for the 3-sphere consider the differential [147]

$$J = L_{\vec{v}}^{-1} dL_{\vec{v}}.$$
 (4.2)

where $J \in \mathfrak{so}(3)$ and d is the exterior derivative. This might seem mysterious at first glance, therefore let us first consider the following example.

Example: Translations

Suppose we want to do a translation along the x-direction from x to x + a. Then the function f(x) becomes f(x + a). The transformation may be written as

$$T(a)f(x) = e^{a\frac{\partial}{\partial x}f(x)} = f(x+a),$$

where $e^{a}\frac{\partial}{\partial x}$ is an element of the group of translations and $\frac{\partial}{\partial x}$ is the generator of infinitesimal translations, which lives in the Lie algebra of the group. The transformation is parametrized by a. In general, we have the following

$$de^{f(x)} = e^{f(x)}df,$$

where, in the case of a translation this becomes

$$d\left(e^{a\frac{\partial}{\partial x}}\right) = e^{a\frac{\partial}{\partial x}}da\left(\frac{\partial}{\partial x}\right).$$

The Lie algebra element is $\frac{\partial}{\partial x}$, the infinitesimal translation is da and the group element is $e^{a\frac{\partial}{\partial x}}$. Then we can write

$$e^{-a\frac{\partial}{\partial x}}de^{a\frac{\partial}{\partial x}} = da\left(\frac{\partial}{\partial x}\right)$$

Notice that this last equation has the same form as the differential in (4.2). It tells us that the left hand side produces an infinitesimal translation da along the x-direction (right hand side).

This differential has a special name, it is called the Maurer-Cartan 1-form, and is special because it satisfies the Maurer-Cartan equation. One-forms satisfying the Maurer-Cartan equations may be used to construct a metric that preserves the isometries of the Lie group. Computing (4.2) we find

$$J = L_{\vec{v}}^{-1} dL_{\vec{v}}$$

= $e^{i\theta L_{41}} e^{i\phi L_{12}} e^{i\psi L_{23}} de^{-i\psi L_{23}} e^{-i\phi L_{12}} e^{-i\theta L_{41}}$
= $- d\theta L_{41} - d\phi (L_{12} \cos \theta + L_{42} \sin \theta)$
 $- d\psi (L_{23} \cos \phi + (L_{13} \cos \theta - L_{43} \sin \theta) \sin \phi))$
= $- d\theta L_{41} - d\phi \sin \theta L_{42} + d\psi \sin \theta \sin \phi L_{43}$
 $- d\phi L_{12} - d\psi \cos \phi L_{23} + d\phi \cos \theta \sin \phi L_{31}$ (4.3)

From the expression for J we can extract the following coefficients

$$e^{a}_{\mu} = \begin{pmatrix} 41 & 42 & 43 \\ -1 & 1 & 1 \\ 0 & -\sin\theta & 0 \\ \psi & 0 & \sin\theta\sin\phi \end{bmatrix}$$
(4.4)

where the labelling of rows and columns is indicated [147]. The elements in (4.4) are precisely the components of the vielbein $e_{\mu}^{\ a}dx^{\mu}$. Note that small displacements are generated by the generators which are Lie algebra valued. These generators are labelled by the plane (*ij*) that they rotate in. Thus, (4.4) tells us which generator generates a given small displacement. As an example, making a small rotation in the 41-plane means you will change the value of θ . From the vielbeins, we can write down the metric as follows

$$ds^{2} = (d\theta e^{\alpha}_{\theta} + d\phi e^{\alpha}_{\phi} + d\psi e^{\alpha}_{\psi})^{2}$$
$$= d\theta^{2} + \sin\theta^{2} d\phi^{2} + \sin\theta^{2} \sin\phi^{2} d\psi^{2}$$
(4.5)

which is expected for S^3 in polar coordinates.

4.3 The General Setup

In this section we recall some basic facts about superstring coset σ -models. We also lay the foundation required to study the T-dualisation of these σ -models. The Green-Schwarz action of a superstring that propagates in a 10-dimensional background is given by [116]

$$S = -\frac{T}{2} \int_{\Sigma} (\star \mathcal{E}^A \wedge \mathcal{E}^B \eta_{AB} + 2\kappa B_2), \qquad (4.6)$$

where T denotes the string tension, Σ represents the 2-dimensional worldsheet with curved metric $h_{pq}(\tau, \sigma)$ having Lorentz signature such that the corresponding worldsheet Hodge duality operation \star squares to one ($\star^2 = 1$) when acting on 1-forms⁵. The $\mathcal{E}^A = \mathcal{E}^A(X, \Theta)$ are vector space supervielbeins for all A, B = 0, ..., 9 where (X, Θ) are target space coordinates (10 Grassmann-even (or bosonic) coordinates X and 32 Grassmann-odd (or fermionic) coordinates Θ). The metric η_{AB} is the 10-dimensional target space Minkowski metric. Additionally, there are spinor supervielbeins $\mathcal{E}^{\hat{\alpha}} = \mathcal{E}^{\hat{\alpha}}(X, \Theta)$ for $\hat{\alpha}, \hat{\beta} = 1, ..., 32$ which also describe the geometry of the full 10-dimensional superstring. Finally, B_2 is the NSNS 2-form flux and we consider models for which B_2 has vanishing field strength at $\Theta = 0$, or $dB_2|_{\Theta=0} = 0$.

We want to study backgrounds which are symmetric or semi-symmetric⁶ . These manifolds \mathcal{M} may be described by a coset

⁵In local coordinates (i.e. (τ, σ)) on $\sigma, \star \mathcal{E}^A \wedge \mathcal{E}^B = \sqrt{-\det h_{rs}} h^{pq} \mathcal{E}^A_p \wedge \mathcal{E}^B_q$

⁶Semi-symmetric backgrounds have vanishing NSNS flux and a Lie algebra which admits a \mathbb{Z}_4 -grading with accompanying automorphism $\Omega: G \to G$ and fixed point set H. This means that $\Omega^4 = 1$ and $\Omega(H) = H$.

\mathbb{Z}_4 -graded Coset Superspaces						
Background Type	d	Superspace	Coset	$n_b + n_f$		
	5	$AdS_5 \times S^5$	$\frac{PSU(2,2 4)}{SO(1,4)\times SO(5)}$	10 + 32		
$\alpha = 0$	3	$AdS_3 imes S^3$	$\frac{PSU(1,1 2) \times PSU(1,1 2)}{SU(1,1) \times SU(2)}$	6 + 16		
	2	$AdS_2 \times S^2$	$\frac{PSU(1,1 2)}{SO(1,1)\times U(1)}$	4 + 8		
	3	$AdS_3 \times S^3 \times S^3$	$\frac{D(2,1;\alpha) \times D(2,1;\alpha)}{S0(1,2) \times SO(3) \times SO(3)}$	9 + 16		
$0 < \alpha < 1$	2	$AdS_2 \times S^2 \times S^2$	$\frac{D(2,1;\alpha)}{S0(1,1)\times SO(2)\times SO(2)}$	6+8		

Table 4.2: The cosets corresponding to the various backgrounds for $0 \le \alpha < 1$. Here n_b and n_f are the number of bosonic and fermionic coordinates, respectively.

$$\mathcal{M} := \frac{G}{H},$$

as mentioned in the previous section. The Green-Schwarz action describes a semi-symmetric space, therefore, it admits a \mathbb{Z}_4 -automorphism and there exists a truncation of the Green-Schwarz action to a supercoset σ -model⁷.

4.3.1 Maurer-Cartan Forms and \mathbb{Z}_4 -graded Coset Superspaces

The \mathbb{Z}_4 -automorphism (or grading) $\Omega : G \to G$ induces an automorphism on the corresponding Lie superalgebra \mathfrak{g} of G, which we will also denote by $\Omega : \mathfrak{g} \to \mathfrak{g}$. This automorphism implies that we may decompose the Lie algebra \mathfrak{g} as follows

$$\mathfrak{g} \cong \mathfrak{g}_{(0)} + \mathfrak{g}_{(1)} + \mathfrak{g}_{(2)} + \mathfrak{g}_{(3)}.$$

As a result, the eigenspaces of may also be decomposed as follows

$$\Omega(V_m) = i^m V_{(m)}$$
 for $V_{(m)} \in \mathfrak{g}_{(m)}$.

The Lie algebra \mathfrak{g} contains the generators which generate all possible transformations that leave the metric invariant. The superalgebra elements satisfy the commutators: $[\mathfrak{g}_{(m)}, \mathfrak{g}_{(n)}] \subseteq \mathfrak{g}_{(m+n \mod 4)}$, where $\mathfrak{g}_{(0)}$ is the Lie algebra of H. Table 4.2 contains all the \mathbb{Z}_4 -graded coset Superspaces considered herein. Note that the d = 5 coset describes the full superstring. For d = 2, 3, the cosets listed in Table 4.2 describe those subsectors of the full superstring theory in which the non-supersymmetric fermions have been removed.

There is also a \mathbb{Z}_2 -grading on \mathfrak{g} because we are dealing with a superspace and this means that we have both bosonic and fermionic generators. This divides all the generators into two subgroups, one containing the bosonic generators $\mathfrak{g}_{(0)}$ and $\mathfrak{g}_{(2)}$, and one containing the fermionic generators $\mathfrak{g}_{(1)}$ and $\mathfrak{g}_{(3)}$. For a detailed discussion on the general properties of the Lie superalgebra associated with the Lie supergroups

⁷For backgrounds which posses at least 16 supersymmetries, their corresponding σ -model can be viewed as a κ -symmetry gauge fixing of the full superstring.

considered in Table 4.2, see [148].

To define the Maurer-Cartan form we consider maps of the form $g: \Sigma \to G$ from the 2- dimensional worldsheet Riemann surface Σ onto G. Then, we introduce the pull-back⁸ to Σ via g of the Maurer-Cartan form by

$$J := g^{-1} dg,$$

where the exterior derivative is given by d and in our conventions it acts from the right. By construction, J is a g-valued differential 1-form, which satisfies the Maurer-Cartan equation

$$dJ - J \wedge J = 0.$$

The Maurer-Cartan Equation

We constructed the general Maurer-Cartan form J to satisfy the Maurer-Cartan equation. In this aside we show how this can be proved. Start with the definition

$$J = g^{-1}dg$$

and take the derivative

$$d(g^{-1}dg) = dg^{-1} \wedge dg + g^{-1}d^2g = -g^{-1}dgg^{-1} \wedge dg = -g^{-1}dg \wedge g^{-1}dg$$

which is just $dJ = J \wedge J \implies dJ + JJ = 0$ and we used that $dg^1 = (g^{-1}dg)g^{-1}$ and $d^2g = 0$.

To gain some intuition about why J is a 1-form, consider the following. We may write

$$J = g^{-1}dg = f_u dx^\mu$$

where the f_{μ} are the components of the one form J. Because d is an exterior derivative, J is a 1-form and under a change in coordinates, f_{μ} transforms exactly like something with a lower index should. However, it deserves a special name because it is not a number. It is an element of the Lie algebra \mathfrak{g} , that is, a matrix. These forms are called Lie algebra valued (one) forms. Then pullback in the usual way, for example, if x^{μ} are spacetime coordinates and σ^{A} are the worldsheet coordinates, then

$$f_A = f_\mu \frac{\partial x^\mu}{\partial \sigma^A}$$

⁸Roughly speaking, suppose you have a map defined by $\phi : X \to Y$ and f is a form defined on Y, then the pull-back $\phi * (f) : Y \to X$ is the form on X whose value at $x \in X$ is the value of f at $\phi(x) \in Y$. That is, the form on Y defines a form on X.

The 1-form J is invariant under global left G-transformations⁹ $g_0g \to g$ for $g_0 \in G$. The \mathbb{Z}_4 - automorphism $\Omega : \mathfrak{g} \to \mathfrak{g}$ allows us to decompose J into the eigenspaces of Ω according to

$$J = J_{(0)} + J_{(1)} + J_{(2)} + J_{(3)}$$
, with $\Omega(J_{(m)}) = i^m J_{(m)}$.

The map Ω is the set of all \mathbb{Z}_4 transformations. Hence, the algebra can be thought of as a vector space and as a result the algebra may be decomposed into the eigenspaces of the map Ω as shown above. This is equivalent to a matrix acting on some vector space where you are able to use the eigenvectors of the matrix as a basis. Under local right H -transformations¹⁰ $gh \to g$ for $h \in H$, the $J_{(0)}$ term is a $g_{(0)}$ -valued connection 1-form (i.e. $J_{(0)} \in Lie(H)$). The H-transformations do not move physical points around in the spacetime (i.e. recall that these are transformations in the tangent space at each point which therefore leaves the point invariant with respect to H). Therefore, there is some redundancy in this description.

Using bosonic and fermionic supervielbeins $J_{(m)}$ where m = 1, 2, 3, the supercoset action may be defined. It has the following form

$$S = -T \int_{\Sigma} \mathcal{L}_{G/H} = -\frac{T}{2} \int_{\Sigma} \text{Str}(\star J_{(2)} \wedge J_{(2)} + J_{(1)} \wedge J_{(3)})$$
(4.7)

where Str denotes the supertrace. It is compatible with the \mathbb{Z}_4 -grading

$$Str(V_{(m)}V_{(n)}) = 0$$
 (4.8)

for $V_{(m)} \in \mathfrak{g}_{(m)}$ and $m + n \mod 4$. For the non-exceptional cases, the relative coefficients of the two terms in (4.7) are fixed by κ -symmetry. However, for the exceptional cases these coefficients are not fixed, therefore lacking κ -symmetry invariance [125, 149]. Instead, the coefficients are fixed by integrability conditions belonging to the respective σ -model. Comparing (4.7) to the Green-Schwarz action in (4.6) we can see that the Wess-Zumino term in (4.7), i.e. the second term in the action, is given by B_2 in (4.6). Thus

$$B_2 = \frac{1}{2} \text{Str}(J_{(1)} \wedge J_{(3)})$$

The $\mathfrak{g}_{(0)}$ - valued forms are scalars (or invariant) under the $\mathbb{Z}\mathbb{Z}_4$ - automorphism because the automorphism takes G into G and $\mathfrak{g}_{(0)} \in H$. Therefore, these are the Lie algebra elements that generate displacements in the coordinates, for example. These elements form the *connection* and transform as follows

$$J_{(0)} \to h^{-1} J_{(0)} h + h^{-1} dh$$

The $J_{(m)}$'s for m = 1, 2, 3 transform adjointly as follows

 9 By global we mean that we use the same G transformation at each point on the manifold.

¹⁰By local we mean that a different transformation is applied at each point on the manifold.

$$J_{(m)} \to h^{-1} J_{(m)} h.$$

These Lie algebra elements do not generate coordinate displacements, they act more like internal symmetries. The physical fields will take values in the coset superspace $G/H = gH|g \in G$, thus the corresponding action (geometry) must be invariant under such local H - transformations. This is a reminder that the H- transformations do not move physical fields around. In turn the action may only contain $J_{(m)}$ for m = 1, 2, 3. These are precisely those terms that combine to form Casimirs¹¹ in order to return an action with the correct symmetries for the problem being studied. The coset removes all trivial rotations (i.e. those leaving a point invariant).

The coset G/H is parametrised by d_b bosonic local coordinates X and d_f fermionic local coordinates ϑ , where $d_b + d_f = \dim(G/H) = \dim(G) - \dim(H)$. The maps considered here are

$$(\mathbb{X}, \vartheta) : \Sigma \to G/H.$$

The bosonic supervielbein is given by $J_{J(2)}$ whilst the fermionic supervielbeins are given by $J_{(1)}$ and $J_{(3)}$.

4.3.2 Choosing a \mathbb{Z}_4 -grading

The \mathbb{Z}_4 -grading and its decomposition into Abelian sub-isometries must be chosen with care. If an inappropriate choice is made one cannot successfully apply the T-duality transformations. Thus, making the proof of self-duality complicated, if at all possible. To write down the grading first decompose the R-symmetry generators into $R = (R_{(0)}, R_{(2)})$ with $R_{(0)} \in \mathfrak{g}_{(0)}$ and $R_{(2)} \in \mathfrak{g}_{(2)}$. Then the \mathbb{Z}_4 -grading we use is written as follows:

$$\mathfrak{g}_{(0)} := \langle P + K, M, R_{(0)} \rangle, \quad \mathfrak{g}_{(2)} := \langle P - K, D, R_{(2)} \rangle \\
\mathfrak{g}_{(1)} := \langle Q - S, \hat{Q} - \hat{S} \rangle, \quad \mathfrak{g}_{(3)} := \langle Q + S, \hat{Q} + \hat{S} \rangle.$$
(4.9)

The specific decomposition of R will be discussed in the next section. The sub-algebras $\mathfrak{g}_{(0)}$ and $\mathfrak{g}_{(2)}$ contain bosonic symmetry generators, whilst $\mathfrak{g}_{(1)}$ and $\mathfrak{g}_{(3)}$ contain the fermionic symmetry generators.

4.3.3 Schematic Form of the Superconformal Algebra

The T-dualisation of the action (4.7) is performed along some bosonic and fermionic directions. These T-duality transformations correspond to an (anti-) commuting subgroup of isometries belonging to the underlying supercoset space. These isometries are identified through a choice of basis obtained from the Lie superalgebra \mathfrak{g} of G. Associated with this basis is the superconformal group on the Minkowski

¹¹Casimirs are combinations of Lie algebra elements that commute with everything else in the Lie algebra. For example, consider SO(3). Here, the algebra is made up of $\{L_x, L_y, L_z\}$ and the Casimir is $\vec{L} \cdot \vec{L}$.

Symmetries and their Generators						
Symmetry	Generator	Ν	Type			
Translations	P	d-1	$R^{1,d-2}$			
Lorentz Boosts	M	$\frac{1}{2}(d-1)(d-2)$	$R^{1,d-2}$			
Conformal	K	$\overline{d}-1$	$R^{1,d-2}$			
Dilatation	D	1	$R^{1,d-2}$			
R-symmetry	R	2	$AdS_d \times S^d$			
Fermionic	Q,\hat{Q},S,\hat{S}	2(d-1)	$AdS_d \times S^d$			

Table 4.4: Symmetries and their number of associated generators, N.

conformal boundary, $R^{(1,d2)}$ of the AdS_d space [36]. We may describe the basis for g schematically as

$$[P,K] \sim D + M, \quad [D,P] \sim P, \quad [D,K] \sim K, \quad [M,P] \sim P$$

 $[M,K] \sim K, \quad [M,M] \sim M, \quad [R,R] \sim R,$ (4.10)

where we have chosen not to display vanishing commutators.

In the superconformal extension of (4.10) we find the fermionic generators listed in Table 4.4. These are the complex supersymmetry and superconformal generators [36]. The hatted and ordinary generators are related by Hermitian conjugation. The non vanishing commutators are given, schematically, by

$$[D,Q] \sim Q , \quad [M,Q] \sim Q , \quad [K,Q] \sim \hat{S} , \quad [R,Q] \sim Q + \alpha \hat{Q} ,$$

$$[D,S] \sim S , \quad [M,S] \sim S , \quad [P,S] \sim \hat{Q} , \quad [R,S] \sim S + \alpha \hat{S} ,$$

$$(4.11)$$

and similarly for the hatted generators. The anti-commutators are

$$\{Q, \hat{Q}\} \sim P , \quad \{S, \hat{S}\} \sim K , \quad \{Q, \hat{S}\} \sim \alpha R , \quad \{\hat{Q}, S\} \sim \alpha R \{Q, S\} \sim D + M + R , \quad \{\hat{Q}, \hat{S}\} \sim D + M + R .$$

$$(4.12)$$

Note that the α in the above (anti-) commutators is precisely the same one appearing in Table 4.2. The non-exceptional cases are obtained from the exceptional cases (i.e the last two entries in Table 4.2) by taking the limit $\alpha \to 0$. Therefore $\mathfrak{g} = \langle P, K, D, M, R, Q, \hat{Q}, S, \hat{S} \rangle$.

4.3.4 Coset Representative and Associated Current

For the non-exceptional cases ($\alpha = 0$), the form of the supersymmetry algebra (4.10), (4.11) and (??) imply that the generators P and the complex supercharges Q are in involution¹². Thus, the maximal Abelian subalgebra of \mathfrak{g} is just $\langle P, Q \rangle$, this means that the (anti-) commuting isometries of the $G/H \sigma$ -

¹²If two things are in involution, they commute. This means that they can define an Abelian algebra.

model are associated with $\langle P, Q \rangle$. For the exceptional cases ($\alpha \neq 0$), the situation is trickier. The maximal Abelian subalgebra of \mathfrak{g} is now generated by P and Q in addition to some R-symmetry generators, denoted by L_+ . The latter generator is complex with Hermitian conjugate denoted by L. For d = 2 there is one such generator given by $L_+ \equiv L_+^1$, whilst for d = 3 there are two generators given by $L_+ \equiv L_+^{1,2}$. Thus, the (anti-) commuting isometries in the exceptional cases are associated with $\langle P, Q, L_+ \rangle$. Then

$$[L_{+}^{1}, L_{-}^{1}] \sim L_{3} \sim [L_{+}^{2}, L_{-}^{2}]$$

With respect to the \mathbb{Z}_4 -grading (4.9), we find that $L_+^1 + L_-^1$ and $L_+^2 - L_-^2$ belong to $\mathfrak{g}_{(0)}$. On the other hand, $L_+^1 - L_-^1$, $L_+^2 + L_-^2$ and L_3 belong to $\mathfrak{g}_{(2)}$. This is the grading we chose here. Using this information, we are able to see that in order to perform T-dualisation on the supercoset action (4.7) along the isometries that we have already mentioned, we take the supercoset representation \mathfrak{g} in the form following [21,22,125]

$$g := e^{xP + \theta Q + \sqrt{\alpha} \lambda_{+} L_{+}} e^{B} e^{\xi S} , \quad e^{B} := e^{\hat{\theta} \hat{Q} + \hat{\xi} \hat{S}} |y|^{D} e^{-\sqrt{\alpha} \lambda_{3} L_{3}} \Lambda_{\alpha}(y) , \qquad (4.13)$$

where x are the coordinates of the Minkowski boundary and |y| is related to the radial direction in AdS_d . In the non-exceptional case ($\alpha = 0$) the coordinates y parametrize the sphere S^d . In the exceptional cases ($\alpha \neq 0$), one S^d is parametrized by y and the second by λ_+ and λ_3 . These coordinates, λ_+ and λ_3 , are assumed to be complex. The specific form of the $\Lambda(y)$ will depend on the underlying geometry that is used. The Grassman-odd directions of the coset superspace are parametrized by 2(d-1) fermionic coordinates ($\theta, \hat{\theta}, \xi, \hat{\xi}$). The form of (4.13) was achieved using *local right* H - transformations. Also, because P, Q and L_+ are in involution our choice of coset representative guarantees that the action (4.7) depends on x, θ and L_+ alone. It depends on the latter coordinates through their derivatives dx, $d\theta$ and dL_+ .

As we have mentioned the proof of self-duality of the supercoset σ -models (4.7) under bosonic and fermionic T-duality has up till now been performed by fixing κ -symmetry gauge [36,38,39,65,108,125,126]. The most convenient choice is setting $\xi = 0$ [21,22,125]. However, there is a problem, if the supercoset model has been gauge fixed with respect to the associated superstring action, then κ -symmetry has already been used to set the non-supersymmetric fermions to zero. As a result, κ -symmetry gauge fixing can no longer be used in the T-dualisation procedure. The situation is more severe when dealing with the exceptional backgrounds where the rank of the κ - symmetry is zero (i.e. none of the non-supersymmetric fermions can be set to zero) [37,125,149]. Thereby motivating our choice to study these σ -models without fixing κ -symmetry gauge. All fermionic coordinates in (4.9) will be considered. With respect to (4.13), the Mauer Cartan current has the following form

$$J = g^{-1} dg = e^{-\xi S} J^{(0)} e^{\xi S} + d\xi S , \qquad (4.14)$$

where $J_{(0)}$ is the current at $\xi = 0$. For a detailed list of the components of J and their explicit forms, see [36]. However, schematically the components of J are

$$J_{P} = J_{P}^{(0)}, \quad J_{Q} = J_{Q}^{(0)}, \quad J_{L_{+}} = J_{L_{+}}^{(0)}, J_{D} = J_{D}^{(0)} + J_{Q}^{(0)}\xi, \quad J_{M} = J_{M}^{(0)} + J_{Q}^{(0)}\xi, J_{R} = J_{R}^{(0)} + J_{Q}^{(0)}\xi, \quad J_{L_{3}} = J_{L_{3}}^{(0)} + \alpha J_{Q}^{(0)}\xi, J_{\hat{Q}} = J_{\hat{Q}}^{(0)} + J_{P}^{(0)}\xi, \quad J_{\hat{S}} = J_{\hat{S}}^{(0)} + \alpha J_{L_{+}}^{(0)}\xi, J_{K} = J_{\hat{S}}^{(0)}\xi + \alpha J_{L_{+}}^{(0)}\xi^{2}, \quad J_{L_{-}} = \alpha J_{\hat{Q}}^{(0)}\xi + \alpha J_{P}^{(0)}\xi^{2}, J_{S} = d\xi + (J_{D}^{(0)} + J_{M}^{(0)} + J_{R}^{(0)} + \alpha J_{L_{3}}^{(0)})\xi + J_{Q}^{(0)}\xi^{2}.$$

$$(4.15)$$

Here the current J depends at most on ξ quadratically. This is very important as it simplifies the T-dualisation procedure significantly.

4.3.5 The T-duality Procedure

Proceeding as in [22,150], the non-exceptional cases are T-dualised along x and θ . The exceptional cases are T-dualised along x, and $_+$, following [125]. In the standard procedure [12,13,129] we start with the supercoset σ -model¹³ and then substitute

$$(\mathrm{d}x,\mathrm{d}\theta,\mathrm{d}\lambda_+)\mapsto (A_\mathrm{b},A_\mathrm{f},A_+).$$

These field redefinitions modify the action in the following way

$$S = S[(\mathrm{d}x, \mathrm{d}\theta, \mathrm{d}\lambda_{+}) \mapsto (A_{\mathrm{b}}, A_{\mathrm{f}}, A_{+})] + \int_{\Sigma} \left(\tilde{x} \mathrm{d}A_{\mathrm{b}} + \tilde{\theta} \mathrm{d}A_{\mathrm{f}} + \sqrt{\alpha} \,\tilde{\lambda}_{+} \mathrm{d}A_{+} \right). \tag{4.16}$$

The auxiliary fields $\{A_b, A_f, A_+\}$ are differential 1-forms and $\{\tilde{x}, \tilde{\theta}, \tilde{\lambda}_+\}$ are Lagrange multipliers ensuring that

$$dA_b = d^2 A_b = 0, \ dA_f = d^2 A_f = 0, \ dA_{\lambda_+} = d^2 A_{\lambda_+} = 0, \tag{4.17}$$

since $\{A_b = dx, A_f = d\theta, A_+ = d\lambda_+\}$ and $d^2 = 0$. As a result (see Part I), when we integrate out the Lagrange multipliers we recover the original σ -model action. To find the T-dualised action \tilde{S} out the differential 1-forms $\{A_b, A_f, A_+\}$ instead. A simplification can be made [36]

$$e^{-B} (A_{b}P + A_{f}Q + \sqrt{\alpha} A_{+}L_{+}) e^{B} = A'_{b}P + A'_{f}Q + \sqrt{\alpha} A'_{+}L_{+}$$
(4.18)

when performing this operation, since the Abelian algebra $\langle P, Q, L_+ \rangle$ is invariant under conjugation by e^B . We may thus write (4.18) as

¹³The form of this action in terms of the decomposition of J can be found in [36], equation (??).

$$A_{\rm b}P + A_{\rm f}Q + \sqrt{\alpha}A_{+}L_{+} = e^{B}(A_{\rm b}'P + A_{\rm f}'Q + \sqrt{\alpha}A_{+}'L_{+})e^{-B}.$$
(4.19)

Consider the following field redefinitions

$$(A_{\rm b}, A_{\rm f}, A_{+}) \mapsto (A'_{\rm b}P, A'_{\rm f}Q, \sqrt{\alpha}A'_{+}L_{+}),$$

then $A_b = dx$, $A_f = d\theta$, $A_+ = d\lambda_+$ and (4.15) imply that $A'_b = J_P$, $A'_f = J_Q$, and $A'_+ = J_{L_+}$ [36]. Next, substitute

$$A_{\rm b} = \left[e^{B} \left(A'_{\rm b} P + A'_{\rm f} Q + \sqrt{\alpha} A'_{+} L_{+} \right) e^{-B} \right]_{P},$$

$$A_{\rm f} = \left[e^{B} \left(A'_{\rm b} P + A'_{\rm f} Q + \sqrt{\alpha} A'_{+} L_{+} \right) e^{-B} \right]_{Q},$$

$$A_{+} = \left[e^{B} \left(A'_{\rm b} P + A'_{\rm f} Q + \sqrt{\alpha} A'_{+} L_{+} \right) e^{-B} \right]_{L_{+}},$$

(4.20)

into (4.16) and integrate out $\{A'_b, A'_f, A'_+\}$ to obtain the T-dualised action \tilde{S} . Our next goal is to illustrate that \tilde{S} is the Green-Schwartz model in (4.7), with the new coordinates associated with the new choice of coset representative

$$\tilde{g} := e^{\tilde{x}K + \tilde{\theta}M^{-1}S + \sqrt{\alpha}\,\tilde{\lambda}_{+}L_{-}} e^{B} e^{F(\xi)} , \qquad (4.21)$$

where M := (QS) and e^B is given in (4.13). The term $F(\xi)$ depends on the background under consideration. For AdS_5 it is given by

$$F(\xi) \sim -[\xi + \xi^5]Q + [\xi^3 + \xi^7]S$$
 (4.22)

For the AdS_2 cases, we have $F(\xi) = \xi Q$ and for AdS_3 , we have that $F(\xi) = \xi Q + \xi^3 S$. Due to the presence of $F(\xi)$ in (4.21), the $\tilde{J} = \tilde{g}^{-1}d\tilde{g}$ arising from (4.21) will generally not contain quadratic terms of fermionic coordinates ξ . Finally, through further complicated field redefinitions $(x, \theta) \to (x', \theta')$ we can bring the dual coset element (4.21) into the form of (4.13)

$$\tilde{g} \ = \ \mathrm{e}^{\tilde{x}'K + \tilde{\theta}'M^{-1}S + \sqrt{\alpha}\,\tilde{\lambda}'_{+}L_{-}} \mathrm{e}^{B'}\mathrm{e}^{-\xi'Q} \,.$$

4.4 Example: $AdS_5 \times S^5$ Self-duality

The $AdS_5 \times S^5$ superstring is probably the most important example in the thesis outside of the less than maximal supersymmetric cases, which are the focus of this thesis. In this section we show that $AdS_5 \times S^5$ is self-dual in a κ -covariant way. That is, we have considered all fermionic directions. This result is exciting because it proves that $AdS_5 \times S^5$ is *exactly* self-dual.

4.4.1 Supercoset Action

The $AdS_5 \times S^5$ superspace is accompanied by the following coset

$$\frac{PSU(2,2|4)}{SO(1,4)\times SO(5)}$$

This coset describes the full type IIB $AdS_5 \times S^5$ background, which is parametrized by ten bosonic coordinates $(X^M) = (x^m, |y|, y^{\hat{a}})$ with m, n, ... = 0, ..., 3 and $\hat{a}, \hat{b}, ... = 5, ..., 9$. Additionally there are two 16-component Majorana-Weyl spinor coordinates $\Theta^i = \frac{1}{2}(1 + \Gamma^{11})\Theta^i$ with i, j, ... = 1, 2. Both spinors have the same chirality. With this parametrization, we write the line element for $AdS_5 \times S^5$ as

$$ds^{2} = \frac{1}{|y|^{2}} (dx^{m} dx^{n} \eta_{mn} + dy^{\hat{a}} dy^{\hat{a}} + d\hat{y} d\hat{y}), \qquad (4.23)$$

where $|y|^2 = (y^{\hat{a}}y^{\hat{a}} + \hat{y}\hat{y})$. The maximally supersymmetric background is supported by the following non-vanishing 5-form components

$$F_{01234} = -F_{56789} = 4, (4.24)$$

or, equivalently

$$F_5 = 4(1+*) \operatorname{Vol}_{\operatorname{AdS}_5} = 4(1+*) e^0 \wedge \dots \wedge e^4$$
(4.25)

and a dilaton that we set to zero for simplicity.

Lie Superalgebra and Cartan Forms

The general form of a Lie superalgebra for a symmetric space was given in [121]. Inserting (4.25) into this algebra yields the following form of the psu(2, 2|4) superalgebra

$$= \eta_{AC}M_{BD} - \eta_{AD}M_{BC} - \eta_{BC}M_{AD} + \eta_{BD}M_{AC} ,$$

$$[P_A, P_B] = -\frac{1}{2}R_{AB}{}^{CD}M_{CD} ,$$

$$[M_{AB}, P_C] = \eta_{AC}P_B - \eta_{BC}P_A , \qquad [M_{AB}, \mathcal{Q}_{\alpha i}] = -\frac{1}{2}(\mathcal{Q}\Gamma_{AB})_{\alpha i} ,$$

$$[P_A, \mathcal{Q}_{\alpha i}] = -\frac{1}{2}(\mathcal{Q}\varepsilon\Gamma^{01234}\Gamma_A)_{\alpha i} ,$$

$$\{\mathcal{Q}_{\alpha i}, \mathcal{Q}_{\beta j}\} = i\delta_{ij}(\Gamma^A)_{\alpha\beta}P_A - \frac{i}{2}\varepsilon_{ij}(\Gamma^A\Gamma^{01234}\Gamma^B)_{\alpha\beta}M_{AB} ,$$

$$(4.26)$$

where $\varepsilon^{ij} = \varepsilon_{ji}$ and $\varepsilon^{12} = 1$. $(M_{AB}) = (M_{ab}, M_{\hat{a}\hat{b}})$ with $a, b, \dots = 0, \dots, 4$ and $\hat{a}, \hat{b}, \dots = 5, \dots 9$ generate the $SO(1, 4) \times SO(5)$ rotations, whilst $(P_A) = (\mathcal{P}_a, P_{\hat{a}})$ generate the $AdS_5 \times S^5$ translations. The curvature of

 AdS_5 and S^5 are given by $R_{ab}^{cd} = 2\delta^c_{[a}\delta^d_{b]}$ and $R_{\hat{a}\hat{b}}^{\hat{c}\hat{d}} = -2\delta^{\hat{c}}_{[\hat{a}}\delta^{\hat{d}}_{\hat{b}]}$, respectively [36]. The $\mathcal{Q}_{\alpha i}$ are supercharges. For details on the superconformal form of the Lie superalgebra consult [36]. Finally, the corresponding Maurer-Cartan form is

$$J(X,\Theta) = \frac{1}{2}\Omega^{AB}M_{AB} + E^{A}P_{A} + E^{\alpha i}\mathcal{Q}_{\alpha i}, \qquad (4.27)$$

which is made from the super-connection $\Omega^{AB}(X, \Theta)$ and the supervielbeins $E^A(X, \Theta)$ and $E^{\alpha i}(X, \Theta)$. The $\mathfrak{psu}(2, 2|4)$ Lie superalgebra in superconformal form (4.10), (4.11) and (4.12) can be written down by defining new bosonic and fermionic generators. The bosonic generators are (a = m, 4 with m = 0, ..., 3)

$$D := \mathcal{P}_4 , \quad P_m := \mathcal{P}_m + M_{m4} , \quad K_m := -\mathcal{P}_m + M_{m4} , M_{mn} , \quad R_{\hat{a}} := P_{\hat{a}} , \quad R_{\hat{a}\hat{b}} := -M_{\hat{a}\hat{b}}.$$
(4.28)

The fermionic generators are now defined by

$$Q := -\frac{1}{\sqrt{2}} (\mathcal{Q}^{1} - i\mathcal{Q}^{2}) \mathbb{P}_{+} , \quad \hat{Q} := -\frac{1}{\sqrt{2}} (\mathcal{Q}^{1} + i\mathcal{Q}^{2}) \mathbb{P}_{-} ,$$

$$S := \frac{1}{\sqrt{2}} (\mathcal{Q}^{1} + i\mathcal{Q}^{2}) \mathbb{P}_{+} , \quad \hat{S} := \frac{1}{\sqrt{2}} (\mathcal{Q}^{1} - i\mathcal{Q}^{2}) \mathbb{P}_{-} ,$$
(4.29)

with projection operators given by

$$\mathbb{P}_{\pm} := \frac{1}{2} (1 \pm i \Gamma^{0123})$$

The commutation relations for these generators can be found in [36]. The corresponding basis provides the following non-zero components of the invariant form on $\mathfrak{psu}(2,2|4)$

$$\operatorname{Str}(K_n P_m) = -2\eta_{mn}$$
, $\operatorname{Str}(DD) = 1$, $\operatorname{Str}(S_\alpha Q_\beta) = 2i(\Gamma^4 \mathbb{P}_+)_{\alpha\beta}$.

Currents and the Supercoset Action

Given the parametrization we have chosen, the coset representative (4.13) becomes

$$g := e^{x^m P_m + \theta^{\alpha} Q_{\alpha}} e^B e^{\xi^{\alpha} S_{\alpha}}, \quad e^B := e^{\hat{\theta}^{\alpha} \hat{Q}_{\alpha} + \hat{\xi}^{\alpha} \hat{S}_{\alpha}} |y|^D e^{y^{\hat{\alpha}} R_{\hat{\alpha}}/|y|}.$$
(4.30)

As a consequence of the definitions in (4.29), the fermionic variables satisfy the following projection relations $\mathbb{P}_+\theta = \theta$, $\mathbb{P}_+\xi = \xi$, $\mathbb{P}_-\hat{\theta} = \hat{\theta}$ and $\mathbb{P}_-\hat{\xi} = \hat{\xi}$. Following the general setup in Section 4.3, we can derive the form of the Lie superalgebra currents explicitly [36]. Using the commutation relations in [36], we can obtain the components of the current (4.14). Those independent of ξ are

$$J_{P_m} = \left[e^{-B} (\mathrm{d}x^n P_n + \mathrm{d}\theta Q) e^B \right]_{P_m}, \quad J_{Q_\alpha} = \left[e^{-B} (\mathrm{d}x^n P_n + \mathrm{d}\theta Q) e^B \right]_{Q_\alpha},$$

$$J_{\hat{S}_\beta} = \left[e^{-B} \mathrm{d}e^B \right]_{\hat{S}_\beta}.$$
(4.31)

The components depending linearly on ξ are given by

$$J_{K_{m}} = -i(\Gamma^{m}\xi)_{\alpha} J_{\hat{S}_{\alpha}} , \quad J_{D} = J_{D}^{(0)} - i(\Gamma^{4}\xi)_{\alpha} J_{Q_{\alpha}} , \quad J_{R_{\hat{a}}} = J_{R_{\hat{a}}}^{(0)} - i(\Gamma^{\hat{a}}\xi)_{\alpha} J_{Q_{\alpha}} ,
J_{\hat{Q}} = J_{\hat{Q}}^{(0)} + (\Gamma_{m4}\xi)^{\alpha} J_{P_{m}} ,
J_{M_{mn}} = J_{M_{mn}}^{(0)} - \frac{i}{2} (\xi\Gamma^{mn}\Gamma_{4})_{\alpha} J_{Q_{\alpha}} , \quad J_{R_{\hat{a}\hat{b}}} = J_{R_{\hat{a}\hat{b}}}^{(0)} - \frac{i}{2} (\xi\Gamma^{\hat{a}\hat{b}}\Gamma^{4})_{\alpha} J_{Q_{\alpha}} ,$$

$$(4.32)$$

where the label (0) indicates those terms independent of ξ (i.e. for which $\xi = 0$). Finally there is a current which depends quadratically on ξ

$$J_{S_{\alpha}} = d\xi^{\alpha} - \frac{1}{2}\xi^{\alpha}J_{D}^{(0)} - \frac{1}{2}(\Gamma_{mn}\xi)^{\alpha}J_{M_{mn}}^{(0)} + \frac{1}{2}(\Gamma_{\hat{a}4}\xi)^{\alpha}J_{R_{\hat{a}}}^{(0)} + \frac{1}{2}(\Gamma_{\hat{a}\hat{b}\xi})^{\alpha}J_{R_{\hat{a}\hat{b}}}^{(0)} + S^{\alpha}{}_{\beta}J_{Q_{\beta}}$$

$$=: J_{S_{\alpha}}^{(1)} + S^{\alpha}{}_{\beta}J_{Q_{\beta}}, \qquad (4.33)$$

where we define

$$\mathcal{S}^{\alpha}{}_{\beta} := \frac{i}{4}\xi^{\alpha}(\xi\Gamma^{4})_{\beta} + \frac{i}{4}(\Gamma^{4}\Gamma_{\hat{a}}\xi)^{\alpha}(\xi\Gamma^{\hat{a}})_{\beta} + \frac{i}{8}(\Gamma_{mn}\xi)^{\alpha}(\xi\Gamma^{mn}\Gamma_{4})_{\beta} - \frac{i}{8}(\Gamma^{\alpha}_{\hat{a}\hat{b}\xi})(\xi\Gamma^{\hat{a}\hat{b}}\Gamma^{4})_{\beta}, \tag{4.34}$$

for which $S^T = -\Gamma^4 S \Gamma^4$. Comparing the Maurer-Cartan current (4.27) with the coset expression in [36] and utilizing the definition of the superconformal generators in (4.28) and (4.29) in terms of the 10dimensional ones, we are able to write down the relation between the 10-dimensional geometric objects and the components of the supercoset current J. Explicitly,

$$E^{m} = J_{P_{m}} - J_{K_{m}} , \quad E^{4} = J_{D} , \quad E^{\hat{a}} = J_{R_{\hat{a}}},$$

$$\Omega^{mn} = 2J_{M_{mn}} , \quad \Omega^{m4} = J_{P_{m}} + J_{K_{m}} , \quad \Omega^{\hat{a}\hat{b}} = -2J_{R_{\hat{a}\hat{b}}}$$
(4.35)

and

$$E^{1} = \frac{1}{\sqrt{2}}(J_{S} + J_{\hat{S}} - J_{Q} - J_{\hat{Q}}), \quad E^{2} = \frac{i}{\sqrt{2}}(J_{S} - J_{\hat{S}} + J_{Q} - J_{\hat{Q}}).$$
(4.36)

Since $E^1 = J_{(1)}$ and $E^2 = J_{(3)}$, E^1 and E^2 have \mathbb{Z}_4 -gradings 1 and 3, respectively. Then, using (4.31) - (4.33) and the Lagrangians from Section 4.3 for the PSU(2, 2|4) supercoset model, the resulting Lagrangian takes the following form

$$\mathcal{L} = \frac{1}{2} * E^{A} \wedge E^{B} \eta_{AB} - iE^{1} \wedge \Gamma^{01234} E^{2}$$

$$= \frac{1}{2} * (J_{P_{m}} - J_{K_{m}}) \wedge (J_{P_{n}} - J_{K_{n}}) \eta_{mn} + \frac{1}{2} * J_{D} \wedge J_{D} + \frac{1}{2} * J_{R_{\hat{a}}} \wedge J_{R_{\hat{a}}} -$$
$$- \frac{i}{2} J_{S} \wedge \Gamma^{4} J_{S} - \frac{i}{2} J_{\hat{S}} \wedge \Gamma^{4} J_{\hat{S}} + \frac{i}{2} J_{Q} \wedge \Gamma^{4} J_{Q} + \frac{i}{2} J_{\hat{Q}} \wedge \Gamma^{4} J_{\hat{Q}}.$$
(4.37)

Note that we have used (4.36) and the projection properties belonging to the generators (4.29). The NSNS 2-form $B_2 = iE^1 \wedge \Gamma^{01234}E^2$ was calculated from the type IIB supergravity constraints associated with the background (4.25).

4.4.2 T-duality Transformations

Our goal in this section is to T-dualise along x^m and θ^{α} , that is, all the bosonic and fermionic directions. To do so, we follow the steps detailed in Section 4.3 and then introduce the auxiliary 1-form fields A'^m and A'^{α} given in (4.18). Using this information together with the dual variables \tilde{x}^m and $\tilde{\theta}^{\alpha}$ the Lagrangian takes the following form

$$\mathcal{L} = \mathcal{L}_1 + \mathcal{L}_2 + \mathcal{L}_3 \tag{4.38}$$

where

$$\mathcal{L}_{1} := \frac{1}{2} * A'_{m} \wedge A'_{n} \eta^{mn} - \frac{1}{2} A'^{m} \wedge A'^{n} \mathcal{M}_{mn} + A'^{m} \wedge \mathcal{J}_{m},
\mathcal{L}_{2} := \frac{i}{2} A'^{\alpha} \wedge A'^{\beta} N_{\alpha\beta} - \frac{i}{2} * A'^{\alpha} \wedge A'^{\beta} (NL)_{\alpha\beta} + A'^{\alpha} \wedge \mathcal{J}_{\alpha},
\mathcal{L}_{3} := \frac{1}{2} * J_{K_{m}} \wedge J_{K_{n}} \eta_{mn} + \frac{1}{2} * J_{D}^{(0)} \wedge J_{D}^{(0)} + \frac{1}{2} * J_{R_{a}}^{(0)} \wedge J_{R_{a}}^{(0)} - \frac{-\frac{i}{2} J_{S}^{(1)} \wedge \Gamma^{4} J_{S}^{(1)} - \frac{i}{2} J_{\hat{S}} \wedge \Gamma^{4} J_{\hat{S}} + \frac{i}{2} J_{\hat{Q}}^{(0)} \wedge \Gamma^{4} J_{\hat{Q}}^{(0)}.$$

$$(4.39)$$

We also have

$$\mathcal{J}_{m} := -d\tilde{x}^{n} \left[e^{B} P_{m} e^{-B} \right]_{P_{n}} - d\tilde{\theta}_{\alpha} \left[e^{B} P_{m} e^{-B} \right]_{Q_{\alpha}} + i J_{\hat{Q}}^{(0)} \Gamma_{m} \xi + * J_{K_{m}},
\mathcal{J}_{\alpha} := -d\tilde{x}^{m} \left[e^{B} Q_{\alpha} e^{-B} \right]_{P_{m}} + d\tilde{\theta}_{\beta} \left[e^{B} Q_{\alpha} e^{-B} \right]_{Q_{\beta}} -
-i * J_{D}^{(0)} (\Gamma^{4} \xi)_{\alpha} - i * J_{R_{\hat{\alpha}}}^{(0)} (\Gamma^{\hat{\alpha}} \xi)_{\alpha} - i (J_{S}^{(1)} \Gamma^{4} S)_{\alpha}$$
(4.40)

and

$$\mathcal{M}_{AB} := i\xi\Gamma_{AB}\Gamma_{4}\xi ,$$

$$N_{\alpha\beta} := \left(\Gamma^{4}(1+\mathcal{S}^{2})\right)_{\alpha\beta}, \quad (NL)_{\alpha\beta} := i(\Gamma^{4}\xi)_{\alpha}(\Gamma^{4}\xi)_{\beta} + i(\Gamma^{\hat{a}}\xi)_{\alpha}(\Gamma_{\hat{a}}\xi)_{\beta}.$$
(4.41)

$$\tilde{\mathcal{L}} = \tilde{\mathcal{L}}_1 + \tilde{\mathcal{L}}_2 + \mathcal{L}_3 \tag{4.42}$$

with

$$\widetilde{\mathcal{L}}_{1} := \frac{1}{2} * \mathcal{J}^{m} \wedge \mathcal{J}^{n} [(\eta - \mathcal{M}^{2})^{-1}]_{mn} + \frac{1}{2} \mathcal{J}^{m} \wedge \mathcal{J}^{n} [\mathcal{M}(1 - \mathcal{M}^{2})^{-1}]_{mn},$$

$$\widetilde{\mathcal{L}}_{2} := \frac{1}{2} \mathcal{J} \wedge (N - NL^{2})^{-1} \mathcal{J} + \frac{1}{2} * \mathcal{J} \wedge L(N - NL^{2})^{-1} \mathcal{J}.$$
(4.43)

and \mathcal{L}_3 is the same as before¹⁴. We can simplify $\tilde{\mathcal{L}}_1$, then we may write it as (see [36] for details)

$$\tilde{\mathcal{L}}_1 = \frac{1}{2} (1 - \frac{1}{4} \mathcal{M}_{\hat{a}\hat{b}} \mathcal{M}^{\hat{a}\hat{b}})^{-1} (* \mathcal{J}^m \wedge \mathcal{J}^n \eta_{mn} + \mathcal{J}^m \wedge \mathcal{J}^n \mathcal{M}_{mn}) + \mathcal{J}^m \mathcal{M}_{mn} \mathcal{M}_{mn} + \mathcal{J}^m + \mathcal{J}^m + \mathcal{J}^m + \mathcal{J}^m +$$

To complete the self-dualisation process, we would like to cast the dual Lagrangian in the form of the original Lagrangian¹⁵. The result of this action are the new expressions for the currents \mathcal{J}_m and \mathcal{J}_α

$$\begin{aligned}
\mathcal{J}_{m} &= \tilde{J}_{P_{m}}^{(1)} - \tilde{J}_{K_{m}}^{(0)} + *J_{K_{m}} , \\
\mathcal{J}_{\alpha} &= \mathrm{i}(\Gamma^{4}\tilde{J}_{S}^{(0)})_{\alpha} - \mathrm{i}J_{S_{\beta}}^{(1)}\Gamma_{\beta\gamma}^{4}\mathcal{S}^{\gamma}{}_{\alpha} - \mathrm{i}*J_{D}^{(0)}(\Gamma^{4}\xi)_{\alpha} - \mathrm{i}*J_{R_{\hat{a}}}^{(0)}(\Gamma_{\hat{a}}\xi)_{\alpha} .
\end{aligned} \tag{4.44}$$

The new (dual) choice of coset representative which leads to the Maurer-Cartan forms¹⁶ is given by

$$\tilde{J} = \tilde{g}^{-1} \mathrm{d}\tilde{g} , \quad \tilde{g} := \mathrm{e}^{\tilde{x}^n K_n - \mathrm{i}\tilde{\theta}\Gamma^4 S} \mathrm{e}^B \mathrm{e}^{-(\xi Q + S_\alpha S^\alpha{}_\beta \xi^\beta)(1 - \frac{1}{4}\mathcal{M}_{\hat{a}\hat{b}}\mathcal{M}^{\hat{a}\hat{b}})^{-1}} , \qquad (4.45)$$

Our goal is to show that $\mathcal{L}_g = \mathcal{L}_{\tilde{g}}$, therefore we need to show that $g \sim \tilde{g}$. Through a complicated change of variables [36], the dual coset element (4.45) can be brought into the same form as the original coset element (4.30). This means that the last exponent to the right takes the form¹⁷ $e^{-\xi'Q}$, similar to g. The dual coset element becomes

$$\tilde{g} \cong e^{\tilde{x}'^n K_n - i\tilde{\theta}' \Gamma^4 S} e^{B'} e^{-\xi' Q}, \qquad (4.46)$$

where $e^{B'} = e^B e^{-\frac{i}{2}D(\xi\Gamma^4 S\xi + \mathcal{O}(\xi^8)) - \frac{i}{2}R_{\hat{a}}\xi\Gamma^{\hat{a}}S\xi}$, $\tilde{x}'^n = \tilde{x}^n + f^n(y,\hat{\theta},\hat{\xi},\xi)$, $\tilde{\theta}'_{\alpha} = \tilde{\theta}_{\alpha} + f_{\alpha}(y,\hat{\theta},\hat{\xi},\xi)$ and f^n and f_{α} are functions of the coordinates $y^{\hat{\alpha}}$ of S^5 , the radial direction |y| of AdS_5 and the Grassmann-odd coordinates $\hat{\theta}, \hat{\xi}$, and ξ . The choices (4.45) and (4.46) have associated \mathbb{Z}_4 - automorphism

 $^{^{14}\}mathcal{L}$ is a function of currents not involved in the T-dualisation process and therefore remains invariant under dualisation.

¹⁵For details see [36].

¹⁶Found in (3.21) in [36].

¹⁷Details may be found in [36].

$$P_m \leftrightarrow K_m, \quad D \to -D, \quad R_{\hat{a}} \to -R_{\hat{a}}, S \to -iQ, \quad \hat{S} \to -i\hat{Q}, \quad Q \to -iS, \quad \hat{Q} \to -i\hat{S}$$

$$(4.47)$$

of the $\mathfrak{p}su(2,2|4)$ Lie superalgebra. The choice (4.45) of the dual element \tilde{g} , tells us that the ξ -independent components $\tilde{J}_Q^{(0)}$, $\tilde{J}_{\hat{S}}^{(0)}$, $\tilde{J}_D^{(0)}$, $\tilde{J}_{R_{\hat{a}}}^{(0)}$, $\tilde{J}_{R_{\hat{a}}\hat{b}}^{(0)}$ and $\tilde{J}_{M_{mn}}^{(0)}$ of the currents are the same as the ones without tilde, whereas the full expressions for the dual currents \tilde{J}_m , \tilde{J}_{K_m} , $\tilde{J}_{\hat{S}}$, and $\tilde{J}_{\hat{Q}}$ are

$$\begin{split} \tilde{J}_{P_{m}} &= \tilde{J}_{P_{m}}^{(1)} \left(1 - \frac{1}{4} \mathcal{M}_{\hat{a}\hat{b}} \mathcal{M}^{\hat{a}\hat{b}}\right)^{-\frac{1}{2}}, \\ \tilde{J}_{K_{m}} &= \left(\tilde{J}_{K_{m}}^{(0)} - J_{K_{l}} \mathcal{M}_{l}^{m}\right) \left(1 - \frac{1}{4} \mathcal{M}_{\hat{a}\hat{b}} \mathcal{M}^{\hat{a}\hat{b}}\right)^{-\frac{1}{2}}, \\ \tilde{J}_{\hat{S}} &= \tilde{J}_{\hat{S}}^{(0)} + \tilde{J}_{K_{m}} (\Gamma_{4} \Gamma_{m} \xi) + \frac{1}{2} J_{K_{n}} \mathcal{M}_{nm} (\Gamma^{4} \Gamma^{m} \xi) \left(1 + \frac{1}{16} \mathcal{M}_{\hat{a}\hat{b}} \mathcal{M}^{\hat{a}\hat{b}}\right), \\ \tilde{J}_{\hat{Q}} &= \tilde{J}_{\hat{Q}}^{(0)} + \frac{1}{2} \tilde{J}_{P_{m}}^{(1)} (\Gamma^{4} \Gamma^{n} \xi) \mathcal{M}_{nm} \left(1 + \frac{3}{16} \mathcal{M}_{\hat{a}\hat{b}} \mathcal{M}^{\hat{a}\hat{b}}\right). \end{split}$$
(4.48)

Similarly, the dual currents $\tilde{J}_{\hat{R}_{\underline{a}}}=(\tilde{J}_D\;,\;\tilde{J}_{R_{\hat{a}}}),\;\tilde{J}_Q$ and \tilde{J}_S are

$$\begin{split} \tilde{J}_{R_{\underline{a}}} &= \tilde{J}_{R_{\underline{a}}}^{(0)} - \mathrm{i}(\tilde{J}_{S}^{(0)}\Gamma^{4} - J_{S}^{(1)}\Gamma^{4}S)(N - NL^{2})^{-1}\Gamma^{\underline{a}}\xi + \frac{\mathrm{i}}{2}J_{R_{\underline{b}}}^{(0)}\xi\Gamma^{\underline{a}}L(N - NL^{2})^{-1}\Gamma_{\underline{b}}\xi \\ &- \frac{1}{2}(\tilde{J}_{S}^{(0)} + \mathcal{S}J_{S}^{(1)})\Gamma_{\underline{b}}\xi(\xi\Gamma^{\underline{b}}L\Gamma^{4}\Gamma^{\underline{a}}\xi) + \frac{1}{8}J_{R_{\underline{c}}}^{(0)}\xi\Gamma^{\underline{b}}L\Gamma^{4}\Gamma^{\underline{a}}\xi(\xi\Gamma_{\underline{b}}L\Gamma^{4}\Gamma_{\underline{c}}\xi), \\ \tilde{J}_{Q} &= \left[N^{2}(1 - L^{2})\right]^{-\frac{1}{2}}(\tilde{J}_{Q}^{(1)} - \mathcal{S}\tilde{J}_{S}^{(0)}) , \quad \tilde{J}_{Q}^{(1)} = -J_{S}^{(1)} - J_{R_{\underline{a}}}^{(0)}(\Gamma^{4}\Gamma_{\underline{a}}\xi), \\ \tilde{J}_{S} &= \left[N^{2}(1 - L^{2})\right]^{-\frac{1}{2}}\left[\tilde{J}_{S}^{(0)} - \mathcal{S}\tilde{J}_{Q}^{(1)} + (2\mathcal{S} - \Gamma^{4}NL)J_{S}^{(1)}\right]. \end{split}$$

$$\tag{4.49}$$

Then, upon substituting the first two equations in (44.48) into (4.44), we get the following expression for \mathcal{J}_m

$$\mathcal{J}^{m} = (\tilde{J}_{P_{m}} - \tilde{J}_{K_{m}})(1 - \frac{1}{4}\mathcal{M}_{\hat{a}\hat{b}}\mathcal{M}^{\hat{a}\hat{b}})^{\frac{1}{2}} + *J_{K_{m}} - J_{K_{n}}\mathcal{M}_{n}^{m}.$$
(4.50)

Using (4.48) and (4.49), we find the relations

$$-\frac{i}{2}\tilde{J}_{\hat{S}}\Gamma^{4}\tilde{J}_{\hat{S}} + \frac{i}{2}\tilde{J}_{\hat{S}}^{(0)}\Gamma^{4}\tilde{J}_{\hat{S}}^{(0)} = \tilde{J}_{K_{m}}^{(0)}J_{K_{n}}\eta_{mn} + \frac{1}{2}\tilde{J}_{K_{m}}\tilde{J}_{K_{n}}\mathcal{M}_{mn} - \frac{1}{2}J_{K_{m}}J_{K_{n}}\mathcal{M}_{mn}$$
(4.51)

and

$$\frac{1}{2}\tilde{J}_{P_m}\tilde{J}_{P_n}\mathcal{M}_{mn} = \frac{1}{2}\tilde{J}_{\hat{Q}}\Gamma^4\tilde{J}_{\hat{Q}} - \frac{1}{2}\tilde{J}_{\hat{Q}}^{(0)}\Gamma^4\tilde{J}_{\hat{Q}}^{(0)}.$$
(4.52)

Furthermore, combining (4.50), (4.51) and (4.52) with the Lagrangians $\tilde{\mathcal{L}}_1$ and \mathcal{L}_3 , it follows that $\tilde{\mathcal{L}} + \mathcal{L}_3$ has the following form

$$\tilde{\mathcal{L}}_{1} + \mathcal{L}_{3} = \frac{1}{2} * (\tilde{J}_{P_{m}} - \tilde{J}_{K_{m}}) \wedge (\tilde{J}_{P_{n}} - \tilde{J}_{K_{n}})\eta_{mn} - \frac{1}{2}\tilde{J}_{\hat{S}} \wedge \Gamma^{4}\tilde{J}_{\hat{S}} + \frac{1}{2}\tilde{J}_{\hat{Q}} \wedge \Gamma^{4}\tilde{J}_{\hat{Q}} + \\
+ \frac{1}{2} * J_{D}^{(0)} \wedge J_{D}^{(0)} + \frac{1}{2} * J_{R_{\hat{a}}}^{(0)} \wedge J_{R_{\hat{a}}}^{(0)} - \frac{1}{2}J_{S}^{(1)} \wedge \Gamma^{4}J_{S}^{(1)} - \\
- \tilde{J}_{K_{m}} \wedge \tilde{J}_{P_{n}}\mathcal{M}_{mn} - \tilde{J}_{P_{m}}^{(1)} \wedge J_{K_{n}}\eta_{mn} .$$
(4.53)

Additionally, using the first equation in (4.49), we find that the Lagrangian $\tilde{\mathcal{L}}_2$ becomes

$$\tilde{\mathcal{L}}_{2} = \frac{1}{2} * \tilde{J}_{R_{\underline{a}}} \wedge \tilde{J}_{R_{\underline{a}}} - \frac{1}{2} * J_{R_{\underline{a}}}^{(0)} \wedge J_{R_{\underline{a}}}^{(0)} - \\
- \frac{i}{2} \left(\tilde{J}_{S}^{(0)} + \mathcal{S} J_{S}^{(1)} \right) \wedge (N - NL^{2})^{-1} \left(\tilde{J}_{S}^{(0)} + \mathcal{S} J_{S}^{(1)} \right) \\
+ i J_{R_{\underline{a}}}^{(0)} \wedge \left(\tilde{J}_{S}^{(0)} + \mathcal{S} J_{S}^{(1)} \right) \Gamma^{4} L (N - NL^{2})^{-1} \Gamma_{\underline{a}} \xi + \\
+ \frac{i}{2} J_{R_{\underline{a}}}^{(0)} \wedge J_{R_{\underline{b}}}^{(0)} \xi \Gamma_{\underline{a}} (N - NL^{2})^{-1} \Gamma_{\underline{b}} \xi .$$
(4.54)

Using the above equations (4.53) and (4.54), we obtain

$$\tilde{\mathcal{L}} = \tilde{\mathcal{L}}_1 + \tilde{\mathcal{L}}_2 + \mathcal{L}_3 = \mathcal{L}_{\tilde{g}} + \mathcal{L}' + \mathcal{L}'', \qquad (4.55)$$

where

$$\mathcal{L}_{\tilde{g}} := \frac{1}{2} * (\tilde{J}_{P_m} - \tilde{J}_{K_m}) \wedge (\tilde{J}_{P_n} - \tilde{J}_{K_n}) \eta_{mn} + \frac{1}{2} * \tilde{J}_D \wedge \tilde{J}_D + \frac{1}{2} * \tilde{J}_{R_{\hat{a}}} \wedge \tilde{J}_{R_{\hat{a}}} - \frac{1}{2} \tilde{J}_S \wedge \Gamma^4 \tilde{J}_S - \frac{1}{2} \tilde{J}_{\hat{S}} \wedge \Gamma^4 \tilde{J}_{\hat{S}} + \frac{1}{2} \tilde{J}_Q \wedge \Gamma^4 \tilde{J}_Q + \frac{1}{2} \tilde{J}_{\hat{Q}} \wedge \Gamma^4 \tilde{J}_{\hat{Q}}$$

$$(4.56)$$

is constructed in terms of the $\frac{PSU(2,2|4)}{SO(1,4)\times SO(5)}$ currents built from the dual coset element. Whilst, \mathcal{L}' and \mathcal{L}'' are given by

$$\mathcal{L}' := i J_S^{(1)} \wedge (N - NL^2)^{-1} N^{-1} \Gamma^4 \tilde{J}_Q^{(1)} - i J_S^{(1)} \wedge \Gamma^4 \mathcal{S} (N - NL^2)^{-1} \mathcal{S} \tilde{J}_Q^{(1)} + J_{K_m} \wedge \tilde{J}_{P_n}^{(1)} \eta_{mn} (1 + \frac{1}{4} \mathcal{M}_{\hat{a}\hat{b}} \mathcal{M}^{\hat{a}\hat{b}})$$

$$(4.57)$$

and

$$\mathcal{L}'' := i \tilde{J}_{S}^{(0)} \wedge (N - NL^{2})^{-1} (4N^{-1} \Gamma^{4} S - S - L) J_{S}^{(1)} - - i J_{R_{\underline{a}}}^{(0)} \wedge \tilde{J}_{S}^{(0)} (N - NL^{2})^{-1} N^{-1} \Gamma^{4} (2S - \Gamma^{4} NL) \Gamma^{4} \Gamma_{\underline{a}} \xi - - \tilde{J}_{K_{m}}^{(0)} \wedge \tilde{J}_{P_{n}}^{(1)} \mathcal{M}_{mn} (1 + \frac{1}{4} \mathcal{M}_{\hat{a}\hat{b}} \mathcal{M}^{\hat{a}\hat{b}}).$$

$$(4.58)$$

Realising that (4.56) has the same explicit form as the initial Lagrangian (4.38), hence the Lagrangian of the superstring on $AdS_5 \times S^5$ will be self-dual provided that $\mathcal{L}' + \mathcal{L}''$ is a total derivative. This can be demonstrated by performing some involved computations¹⁸. The results are

 $^{^{18}}$ See [36] for details.

$$\mathcal{L}' = -\mathrm{id} \Big[J_S^{(1)} \Gamma^4 \xi \Big(1 + \frac{1}{4} \mathcal{M}_{\hat{a}\hat{b}} \mathcal{M}^{\hat{a}\hat{b}} \Big) \Big], \qquad (4.59)$$

and

$$\mathcal{L}'' = -\frac{1}{2} d \Big[\tilde{J}_{M_{mn}}^{(1)} \mathcal{M}_{mn} \Big(1 + \frac{1}{4} \mathcal{M}_{\hat{a}\hat{b}} \mathcal{M}^{\hat{a}\hat{b}} \Big) \Big], \qquad (4.60)$$

where the matrices \mathcal{M}_{mn} and $\mathcal{M}_{\hat{a}\hat{b}}$ have been defined by the first equation in (4.41).

Lastly, we have showed that $\int \mathcal{L}_g = \int \mathcal{L}_{\tilde{g}}$ which demonstrates that $AdS_5 \times S^5$ is exactly self-dual. That is, the $AdS_5 \times S^5$ superstring action is self-dual under the worldsheet duality transformation on (x^m, θ^α) coordinates, without gauge fixing κ -symmetry.

4.5 Summary

In this chapter we have introduced the technicalities involved in working with worldsheet T-duality, which contrasts the methods of Chapter 3, involving the supergravity. After an introduction, the ideas involving coset σ -models were slowly introduced. After some useful examples, the general setup for the work that follows was given. Finally, we end with the important example of the self-duality of the $AdS_5 \times S^5$ background. The worldsheet T-duality transforms the superstring σ -model action, which is constructed with the use of the supercoset representative g, into the action constructed using the supercoset element \tilde{g} . The duality makes use of a \mathbb{Z}_4 -automorphism of the $\mathfrak{psu}(2,2|4)$ Lie superalgebra. Whilst the worldsheet approach is challenging as a result of the difficult mathematical tools involved, the worldsheet approach is invaluable as it offers us insight that is missed when working in the supergravity picture. It beautifully demonstrates the T-self-duality exactly, that is, without fixing κ -symmetry gauge.

Chapter 5

Less Than Maximally Supersymmetric Coset Models

5.1 Introduction

This chapter deals with the self-duality of some of the backgrounds considered in Chapter 3 (i.e. $AdS_d \times S^d$ (d = 2, 3) and $AdS_d \times S^d \times S^d$ (d = 2, 3)), but it does so from the perspective of the coset geometry. This is also known as the worldsheet perspective. When we consider the string theory, we attribute the existence of dual superconformal symmetry to the self-duality of the chosen superstring σ -model given certain T-duality transformations along bosonic and fermionic string modes on the worldsheet which we associate with (anti-) commuting isometries of the $AdS_5 \times S^5$ background [21, 22]. Self-duality is the direct consequence of performing a sequence of bosonic and fermionic T-duality transformations. Specifically, we say a background is self-dual when these transformations do not change the values of the background fields. In particular the RR fields and the dilaton¹ remain invariant. The relationship between fermionic T-duality and dual superconformal symmetry, in the context of the $AdS_5 \times S^5$ superstring and the corresponding super Yang-Mills theory, is very well understood [21–23]. However, there are cases involving σ -models which possess less than maximal supersymmetry. These backgrounds are also integrable, but are less understood as compared to the maximal case. They will be the focus of this chapter.

We concentrate on the remaining issues regarding the T-duality of superstrings on $AdS_d \times S^d \times M^{10-2d}$ backgrounds. We would like to obtain a better understanding of these cases with the hope that it provides insights into the still problematic $AdS_4 \times \mathbb{C}P^3$ background. However, for some recent developments in this regard, see [42]. The problem stems from issues which appear when performing fermionic T-duality on the associated σ -model [124, 151] and the supergravity background itself [16, 101, 107, 151]. T-selfduality is shown for supercoset σ -models associated with strings propagating on $AdS_d \times S^d$ (d = 2, 3) backgrounds upon imposing a partial gauge fixing of κ -symmetry of the σ -model actions by putting to zero one quarter of the supercoset fermion modes [36]. An additional issue arises when considering the coset supermodels on $AdS_d \times S^d$ (d = 2, 3) backgrounds as a result of the fact that coset supermodels only

¹See [23] for a review and references.

86

describe some sectors of the full superstring theory on $AdS_d \times S^d \times M^{10-2d}$ backgrounds [36]. For d = 3, only 16 out of 32 supersymmetries in 10-dimensions are preserved by these backgrounds. For d = 2, only 8 out of 32 supersymmetries are preserved. This implies that 16 and 8 fermionic modes, respectively, correspond to the fermionic directions of the associated coset superspace, with the remaining 16 and 24 fermionic modes corresponding to broken supersymmetries [36]. These sectors of the theory are coupled non-trivially to the non-supercoset directions of M^{10-2d} via these modes.

For d = 3, we may use κ -symmetry to set all of the 16 non-supercoset fermionic modes to zero. However, this gauge fixing does not work for a large class of classical string solutions (including strings moving on the $AdS_3 \times S^3$ sub-space only [152]). Even though the $AdS_3 \times S^3$ supercoset σ -model with 16 fermions possesses κ -symmetry [36], this κ -symmetry is broken when the supercoset model is coupled to the T^4 sector (via the Virasoro constraints) of the full superstring action in $AdS_3 \times S^3 \times T^4$ [37]. Here, the 16 non-supercoset fermions have been κ gauge fixed to zero [36]. This means that the κ -symmetry of the $AdS_3 \times S^3$ supercoset sub-sector is part of the κ -symmetry of the complete 10-dimensional superstring which is lost when the non-supercoset fermions have been completely gauge fixed. For the d = 2 case we may use κ -symmetry to remove 16 of the 24 non-supercoset fermions, then at least 8 of the remaining non-supercoset fermionic modes are always present in the $AdS_2 \times S^2 \times T^6$ string spectrum [37, 39]. Self-duality of the associated supercoset models have been demonstrated for partially gauge fixed κ symmetry. Here, we have set some of the fermionic supercoset coordinates to zero [122, 124]. However, when supercoset models are used to describe gauge fixed sectors of the superstring σ -model where κ symmetry has already been used to remove part of the non-supercoset fermions, we may no longer use κ -symmetry to demonstrate the self-duality of the corresponding supercoset sectors of $AdS_d \times S^d \times M^{10-2d}$ superstrings.

Resulting from these issues is an appreciation of the importance of proving the T-self-duality of superstrings on $AdS_d \times S^d \times M^{10-2d}$ backgrounds without gauge fixing κ -symmetry, that is, by taking into account the non-supercoset fermions. This is precisely what we will demonstrate in the following section. Figure 5.1 illustrates an example of the $AdS_2 \times S^2 \times T^6$ case. Furthermore, we consider the exceptional cases, those backgrounds with two sphere subspaces. There we isolate strings movement to the $AdS_d \times S^d \times S^d$ sectors and fix the non-supercoset fermions to zero. We show that these backgrounds are self dual under combined bosonic and fermionic T-duality. This chapter is based on research presented in [36].

5.2 Self-duality of $AdS_3 \times S^3 \times T^4$ Superstrings

Solutions of the type $AdS_3 \times S^3 \times T^4$ preserve 16 supersymmetries. These supersymmetries generate the required superisometries to form the $PSU(1, 1|2) \times PSU(1, 1|2)$ supergroup (see for example [153]), a subgroup of PSU(2, 2|4). The curvature of the $AdS_3 \times S^3$ subspace is given by

$$R^{ab}_{{}_{\mathrm{AdS}_3}} = -e^a \wedge e^b , \quad R^{\hat{a}\hat{b}}_{{}_{\mathrm{S}^3}} = e^{\hat{a}} \wedge e^{\hat{b}}, \tag{5.1}$$

where $e^a = e^a(x)$ for $a, b, \ldots = 0, 1, 4$ and $e^{\hat{a}} = e^{\hat{a}}(y)$ for $\hat{a}, \hat{b}, \ldots = 5, 6, 7$ are the vielbeins of AdS_3 and S^3 , respectively. The radius of curvature for both manifolds have been set to one for convenience.



Figure 5.1: The idea of self-duality we study is that a sequence of bosonic and fermionic Tdualities returns us to the same background. This is shown here for the case in which we start with type IIB $AdS_2 \times S^2 \times T^6$ supported by an $F^{(5)}$ Ramond-Ramond flux [36].

Backgrounds of this type may be supported by a 5-form flux given by

$$F_5 = \frac{1}{3} (\varepsilon_{cba} e^a \wedge e^b \wedge e^c + \varepsilon_{\hat{c}\hat{b}\hat{a}} e^{\hat{a}} \wedge e^{\hat{b}} \wedge e^{\hat{c}}) \wedge (\mathrm{d}\varphi^2 \wedge \mathrm{d}\varphi^3 + \mathrm{d}\varphi^8 \wedge \mathrm{d}\varphi^9), \tag{5.2}$$

where $d\varphi^{a'}(a', b', \ldots = 2, 3, 8, 9)$ are the flat vielbeins along T^4 . It is important to note the change in the form of the F_5 -flux (5.2) as compared with its value in the $AdS_5 \times S^5$ solution. This difference results in changing the geometry of D = 10 spacetime geometry, which breaks half of the 32 supersymmetries. Only half of the maximal supersymmetry is preserved, the fermionic modes of the string in these backgrounds split into 16 fermions ϑ which are associated with the preserved, and because of this, the string fermionic modes in such a background split into 16 fermions ϑ associated with the preserved symmetries and 16 fermions v associated with the broken symmetries.

Explicitly, the splitting is realized by using the additional projectors $\frac{1}{2}(1 \pm \Gamma^{2389})$ as follows

$$\vartheta^{i} = \frac{1}{2}(1 - \Gamma^{2389})\Theta^{i}, \quad \upsilon^{i} = \frac{1}{2}(1 + \Gamma^{2389})\Theta^{i}.$$
 (5.3)

As for the $AdS_5 \times S^5$ case, the fermions ϑ can be regarded as Grassmann-odd directions of the supercoset space

$$\frac{PSU(1,1|2) \times PSU(1,1|2)}{SO(1,2) \times SU(2)}$$

which contains $\operatorname{AdS}_3 \times S^3$ as the bosonic subspace. The T^4 directions and the non-supercoset fermions v are responsible for extending this supercoset to a full solution obeying the 10-dimensional type IIB supergravity constraints. For certain classical string solutions in $AdS_3 \times S^3 \times T^4$, κ -symmetry may be used to gauge fix to zero all the non-supercoset fermions v. Modulo some constraints (see for example [36]), this gauge permits fluctuations of the string along T^4 to decouple from the $\frac{PSU(1,1|2) \times PSU(1,1|2)}{SO(1,2) \times SU(2)}$ modes

(see [37] for details). Hence, the superstring action reduces to its supercoset part, which can be obtained as a truncation of the $AdS_5 \times S^5$ action, once we select the $PSU(1,1|2) \times PSU(1,1|2)$ subgroup of PSU(2,2|4) and reduce to it.

First, we identify the 10-dimensional indices 2, 3, 8, and 9 as associated with the T^4 directions $(\varphi^{a'}) = (\varphi^I, \varphi^{I'})$ $(I, J, \ldots = 2, 3; I', J', \ldots = 8, 9)$. This particular choice of the tangent space indices associated with $AdS_3 \times S^3 \times T^4$ are related to the way we truncate the $\mathfrak{psu}(2, 2|4)$ superalgebra to $\mathfrak{psu}(1, 1|2) \oplus \mathfrak{psu}(1, 1|2)$. Next, we remove from the algebra all the bosonic generators with the indices m = 2, 3 and $\hat{a} = 8, 9$ from the algebra, and halve the number of fermionic generators by acting on the original 32 generators defined in (4.26) with the additional projector introduced in section 4.4

$$Q^i = \frac{1}{2}Q^i(1-\Gamma^{2389}), \quad i = 1, 2.$$
 (5.4)

The generators (Q, \hat{Q}, S, \hat{S}) defined in section 4.4 are subject to the same projection. The algebra $\mathfrak{psu}(1,1|2) \oplus \mathfrak{psu}(1,1|2)$ may be found in [36]. From a geometrical point of view, this truncation corresponds to obtaining the $AdS_3 \times S^3 \times T^4$ background from $AdS_5 \times S^5$ by formally compactifying two directions of the 4-dimensional Minkowski boundary of AdS_5 and two directions of S^5 onto $T^4 \cong T^2 \times T^2$. Finally, one deforms the value of the F_5 flux as in (5.2).

5.2.1 Self-duality of the Supercoset Model

As already mentioned, $\frac{PSU(1,1|2) \times PSU(1,1|2)}{SO(1,2) \times SU(2)}$ can be described as a truncation of the $AdS_5 \times S^5$ supercoset model. Then, it follows that we may use the results of the previous Chapter (section 4.4) to show that $AdS_3 \times S^3 \times T^4$ is self-dual under the combined T-dualisation of the bosonic coordinates (along the 2-dimensional Minkowski boundary of AdS_3) and four fermionic directions (associated with a commuting subalgebra of the $PSU(1,1|2) \times PSU(1,1|2)$ isometries). Conveniently, the $\frac{PSU(1,1|2) \times PSU(1,1|2)}{SO(1,2) \times SU(2)}$ supercoset σ -model Lagrangian has the same form as (4.37) where the currents constructed with the coset element having a form similar to (4.30). Then, if we follow the steps in section 4.3 we find that, after the T-dualisation process, the Lagrangian turns is equal (up to a total derivative) to the $AdS_3 \times S^3 \sigma$ -model Lagrangian constructed in terms of the supercoset element

$$\tilde{J} = \tilde{g}^{-1} \mathrm{d}\tilde{g} , \qquad \tilde{g} := \mathrm{e}^{\tilde{x}^n K_n - \mathrm{i}\tilde{\theta}\Gamma^4 S} \mathrm{e}^B \mathrm{e}^{-(\xi Q + S_\alpha \mathcal{S}^\alpha{}_\beta \xi^\beta)} , \qquad (5.5)$$

where $S^{\alpha}{}_{\beta}$ was defined in (4.34). For the case $AdS_3 \times S^3$, at most ξ^4 powers are present and thus, the factor $(1 - \frac{1}{4}\mathcal{M}_{\hat{a}\hat{b}}\mathcal{M}^{\hat{a}\hat{b}})^{-1}$ appearing in the $AdS_5 \times S^5$ supercoset element does not enter the expression for \tilde{g} here. Furthermore, we may bring the coset element \tilde{g} in (5.5) into a form similar to that of the original coset element (4.30). Again, this means that we arrive at an expression for \tilde{g} with the last factor to the right being composed of $e^{-\xi Q}$ alone. This was achieved in the $AdS_5 \times S^5$ case as well. The final form of \tilde{g} (after applying a chain of equalities as in [36]) is

$$\tilde{g} := \mathrm{e}^{\tilde{x}^n K_n - \mathrm{i}\bar{\theta}\Gamma^4 S} \mathrm{e}^{B'} \mathrm{e}^{-\xi Q} \tag{5.6}$$

where $e^{B'} = e^B e^{-\frac{i}{2}D\xi\Gamma^4S\xi}$, $\tilde{x}'^n = \tilde{x}^n + f^n(y, \hat{\theta}, \hat{\xi}, \xi)$, $\tilde{\theta}'_{\alpha} = \tilde{\theta}_{\alpha} + f_{\alpha}(y, \hat{\theta}, \hat{\xi}, \xi)$ and f^n and f_{α} are certain functions of the coordinates $y^{\hat{\alpha}}$ of S^3 , the radial direction |y| of AdS_3 and the Grassmann-odd coordinates $\hat{\theta}, \hat{\xi}$, and ξ . The $AdS_2 \times S^2$ case follows similarly (see [36] for a detailed presentation). There we have to consider type IIA and IIB backgrounds separately, although both backgrounds are related by a bosonic T-duality.

5.2.2 Non-supercoset Fermions

So far we have demonstrated that, in a particular κ -symmetry gauge in which the 16 non-supercoset fermions v are set to zero, the $AdS_3 \times S^3 \times T^4$ superstring action is self-dual under the combined fermionic and bosonic T-duality. However, as mentioned before, this gauge is not always permitted. For example, when the classical strings move entirely in the $AdS_3 \times S^3$ subspace. The κ -symmetry projector (M in [152]) fixes the gauge. Meaning that it projects out all but the physical degrees of freedom. Suppose that the κ -symmetry projector commutes with the projector which singles out the unbroken supersymmetries (16 in this case). Then, it cannot eliminate enough non-supercoset fermions to leave us with a physical set of modes. Some unphysical modes will always remain. Consequently, some solutions cannot be found in some gauges and what we have discovered is that the motion of the string in the AdS_3 subspace is important. It is not consistent with our gauge fixing. Thus, it is important to understand how the T-dualisation works in different gauges or without fixing κ -symmetry.

To derive the form of the Green–Schwarz superstring Lagrangian (4.6) in the $AdS_3 \times S^3 \times T^4$ background, we need expressions for the supervielbeins $\mathcal{E}^A(X, \vartheta, v)$ and the NSNS 2-form $B_2(X, \vartheta, v)$ as a power series in v^i . This may be determined in the same way as the Θ -expansion derived in [154] (see for example [155]). Then

$$\mathcal{E}^{A} = E^{A}(X, \vartheta) - iE\Gamma^{A}\upsilon - \frac{i}{2}\mathcal{D}\upsilon\Gamma^{A}\upsilon, \qquad (5.7)$$

up to quadratic order in v. For the NSNS 2-form we have

$$B_2 = B_2^{\text{coset}}(x, y, \vartheta) - iE^A \wedge E\Gamma_A \sigma^3 \upsilon - \frac{1}{2} E\Gamma^A \upsilon \wedge E\Gamma_A \sigma^3 \upsilon - \frac{i}{2} E^A \wedge \mathcal{D}\upsilon\Gamma_A \sigma^3 \upsilon, \qquad (5.8)$$

where $E^{\alpha i}(x, y, \vartheta) = \frac{1}{2}(1 - \Gamma^{2389})^{\alpha}{}_{\beta}E^{\beta i}(x, y, \vartheta)$, $E^{a}(x, y, \vartheta)$, and $E^{\hat{a}}(x, y, \vartheta)$ are the supervielbeins constructed in terms of the $\frac{PSU(1,1|2) \times PSU(1,1|2)}{SO(1,2) \times SU(2)}$ supercoset currents while $E^{a'} = d\varphi^{a'}$ is the flat vielbein along T^4 . The Pauli matrix σ^3_{ij} contracts the indices i, j = 1, 2 of the spinors. Moreover, the NSNS gauge potential B_2^{coset} has again the form as in (4.37), and for the background under consideration the covariant derivative \mathcal{D} is given by

$$\mathcal{D}\upsilon = \nabla \upsilon - \frac{\mathrm{i}}{16 \cdot 5!} E^A F_{B_1 \cdots B_5} \Gamma^{B_1 \cdots B_5} \Gamma_A \sigma^2 \upsilon = \nabla \upsilon - \frac{\mathrm{i}}{4} E^A (1 - \Gamma^{2389}) \Gamma^{01234} \Gamma_A \sigma^2 \upsilon,$$

where $\nabla := d - \frac{1}{4}\Omega^{AB}\Gamma_{AB}$ and $\Omega^{AB}(x, y, \vartheta)$ is the spin connection on $\frac{PSU(1,1|2) \times PSU(1,1|2)}{SO(1,2) \times SU(2)}$ defined in terms of the currents as in section 4.4. Substituting the expressions (5.7) and (5.8) into the string action

(4.6) we arrive at

$$\mathcal{L} = \mathcal{L}_{\text{coset}} + \frac{1}{2} * \mathrm{d}\varphi^{a'} \wedge \mathrm{d}\varphi^{a'} - i * \mathrm{d}\varphi^{a'} \wedge E\Gamma_{a'}\upsilon - \mathrm{id}\varphi^{a'} \wedge E\Gamma_{a'}\sigma^{3}\upsilon - \frac{1}{2} * E\Gamma^{a'}\upsilon \wedge E\Gamma_{a'}\upsilon - \frac{1}{2}E\Gamma^{a'}\upsilon \wedge E\Gamma_{a'}\sigma^{3}\upsilon - \frac{i}{2} * E^{A} \wedge \mathcal{D}\upsilon\Gamma_{A}\upsilon - \frac{i}{2}E^{A} \wedge \mathcal{D}\upsilon\Gamma_{A}\sigma^{3}\upsilon,$$

$$(5.9)$$

where we have used the projector properties (5.3). Here, $\mathcal{L}_{\text{coset}}$ has the same form as (2.1) in [36] (with $\gamma = \Gamma^4$). The Lagrangian above contains a lot of *v*-dependent terms in the gauge in which the *v* are non-zero. These *v* contribute to the T-dualisation along the supercoset directions (x^m, θ) .

5.3 Self-duality of $AdS_d \times S^d \times S^d \times T^{10-3d}$ Superstrings

In this section we finally address the self-duality of the non-exceptional backgrounds $AdS_d \times S^d \times S^d \times T^{10-3d}$ (d = 2,3) by extending the discussion in section 4.4 as well as the previous section. These backgrounds preserve 1/4 and 1/2 of the 10-dimensional supersymmetry and can be supported by either NSNS or RR fluxes [130, 131, 156–158]. In what follows we shall consider these backgrounds to be supported by the following 3-flux

$$F_3 = \frac{1}{3} \left(\varepsilon_{cba} e^a \wedge e^b \wedge e^c + \frac{R_{AdS}}{R_+} \varepsilon_{\hat{c}\hat{b}\hat{a}} e^{\hat{a}} \wedge e^{\hat{b}} \wedge e^{\hat{c}} + \frac{R_{AdS}}{R_-} \varepsilon_{c'b'a'} e^{a'} \wedge e^{b'} \wedge e^{c'} \right), \tag{5.10}$$

where \hat{a} and a' are, respectively the tangent space indices of the two three-spheres and R_{\pm} are their radii. T-dualising the latter background along the S^1 (a bosonic direction), one gets the type IIA $AdS_3 \times S^3 \times S^3 \times S^1$ with the 4-flux

$$F_4 = \mathrm{d}\varphi^9 \wedge \left(\frac{1}{3} (\varepsilon_{cba} e^a \wedge e^b \wedge e^c + \frac{R_{AdS}}{R_+} \varepsilon_{\hat{c}\hat{b}\hat{a}} e^{\hat{a}} \wedge e^{\hat{b}} \wedge e^{\hat{c}} + \frac{R_{AdS}}{R_-} \varepsilon_{c'b'a'} e^{a'} \wedge e^{b'} \wedge e^{c'}\right). \tag{5.11}$$

We will show that the $AdS_d \times S^d \times S^d$ sectors described by supercoset σ -models are T-self-dual under combined bosonic and fermionic T-dualities, provided that T-dualisation involves one of the spheres S^d . The isometries are governed by the exceptional Lie supergroups $D(2,1;\alpha)$ (for d = 2) and $D(2,1;\alpha) \times D(2,1;\alpha)$ (for d = 3). In general, the corresponding calculations are complex and challenging, as a result we will not present details here. To make computations simpler, we set the non-supersymmetric fermionic modes of the string to zero (v = 0). For the d = 3 case we use κ -symmetry to do this, whilst setting them to zero by hand for the d = 2 case. Then the T^{10-3d} sector decouples (modulo the Virasoro constraints) leaving the $AdS_d \times S^d \times S^d$ sector free to be concentrated on.

5.3.1 The Self-duality of $AdS_2 \times S^2 \times S^2$

Supercoset Structure

The $\sigma\text{-model}$ on $AdS_2\times S^2\times S^2$ is based on the supercoset

$$\frac{D(2,1;\alpha)}{SO(1,1) \times SO(2) \times SO(2)}.$$
(5.12)

Before constructing the action to analyse its T-duality properties, let us discuss the Lie superalgebra $\mathfrak{d}(2,1;\alpha)$ of $D(2,1;\alpha)$ first. For general properties of the exceptional Lie superalgebra $\mathfrak{d}(2,1;\alpha)$ see for example [148, 159]. For the 10-dimensional supergravity solutions under consideration, the values of the parameter α are restricted to the interval [0,1]. They determine the relation between the radii of $AdS_2 \times S^2_+ \times S^2_-$,

$$\alpha = \frac{R_{AdS}^2}{R_{-}^2} \text{ and } 1 - \alpha = \frac{R_{AdS}^2}{R_{+}}.$$
(5.13)

To avoid confusion regarding the parameter α and the spinor index α , in what follows we will set $\alpha := \cos^2(\tau) := c^2$ and $1 - \alpha := \sin^2(\tau) := s^2$, respectively.

Lie Superalgebra $\mathfrak{d}(2,1;c^2)$

The maximal Grassmann-even subalgebra of the Lie superalgebra $\mathfrak{d}(2,1;c^2)$ is $\mathfrak{sl}(2,\mathbb{R}) \oplus \mathfrak{su}(2) \oplus \mathfrak{su}(2)$, and we set $\mathfrak{sl}(2,\mathbb{R}) := \langle P, K, D \rangle$, $\mathfrak{su}(2) := \langle L_a \rangle$, and $\mathfrak{su}(2) := \langle R^{\alpha}{}_{\beta} \rangle$, respectively, for $a, b, \ldots = 1, 2, 3$ and $\alpha, \beta, \ldots = 1, 2$. The corresponding commutation relations are

$$= P , [D, K] = -K , [P, K] = 2D ,$$

$$[L_{+}, L_{-}] = -2iL_{3} , [L_{3}, L_{\pm}] = \pm iL_{\pm} , L_{\pm} := iL_{1} \pm L_{2} ,$$

$$[R^{\alpha}{}_{\beta}, R^{\gamma}{}_{\delta}] = i(\delta^{\gamma}{}_{\beta}R^{\alpha}{}_{\delta} - \delta^{\alpha}{}_{\delta}R^{\gamma}{}_{\beta}) .$$
(5.14)

Furthermore, $\mathfrak{d}(2,1;c^2)$ contains eight fermionic generators which we denote by Q_{α} , \hat{Q}_{α} , S_{α} , and \hat{S}_{α} , respectively. Letting $\sigma_{\alpha\beta}^{1,2,3}$ be the Pauli matrices², the remaining non-vanishing (anti-)commutation relations of $\mathfrak{d}(2,1;c^2)$ are given by

$$\{Q_{\alpha}, \hat{Q}_{\beta}\} = -\sigma_{\alpha\beta}^{2} P , \ \{S_{\alpha}, \hat{S}_{\beta}\} = -\sigma_{\alpha\beta}^{2} K , \\ \{Q_{\alpha}, \hat{S}_{\beta}\} = -c^{2} \sigma_{\alpha\beta}^{2} L_{+} , \ \{\hat{Q}_{\alpha}, S_{\beta}\} = c^{2} \sigma_{\alpha\beta}^{2} L_{-} , \\ \{Q_{\alpha}, S_{\beta}\} = -\sigma_{\alpha\beta}^{2} (D + ic^{2} L_{3}) - is^{2} \sigma_{\alpha\gamma}^{2} R^{\gamma}{}_{\beta} , \\ \{\hat{Q}_{\alpha}, \hat{S}_{\beta}\} = \sigma_{\alpha\beta}^{2} (D - ic^{2} L_{3}) + is^{2} \sigma_{\alpha\gamma}^{2} R^{\gamma}{}_{\beta} , \\ [P, S_{\alpha}] = -\hat{Q}_{\alpha} , \ [P, \hat{S}_{\alpha}] = -Q_{\alpha} , \ [K, Q_{\alpha}] = -\hat{S}_{\alpha} , \ [K, \hat{Q}_{\alpha}] = -S_{\alpha} , \\ [D, Q_{\alpha}] = \frac{1}{2} Q_{\alpha} , \ [D, \hat{Q}_{\alpha}] = \frac{1}{2} \hat{Q}_{\alpha} , \ [D, S_{\alpha}] = -\frac{1}{2} S_{\alpha} , \\ [L_{3}, Q_{\alpha}] = \frac{i}{2} Q_{\alpha} , \ [L_{3}, \hat{Q}_{\alpha}] = -\frac{i}{2} \hat{Q}_{\alpha} , \ [L_{3}, S_{\alpha}] = -\frac{i}{2} S_{\alpha} , \\ [L_{+}, S_{\alpha}] = \hat{S}_{\alpha} , \ [L_{-}, \hat{S}_{\alpha}] = -S_{\alpha} , \ [L_{-}, Q_{\alpha}] = \hat{Q}_{\alpha} , \ [L_{+}, \hat{Q}_{\alpha}] = -Q_{\alpha} , \\ [R^{\alpha}{}_{\beta}, T_{\gamma}] = -i(\delta^{\alpha}{}_{\gamma}T_{\beta} - \frac{1}{2}\delta^{\alpha}{}_{\beta}T_{\gamma}) , \ \text{for} \ T_{\alpha} \in \{Q_{\alpha}, \hat{Q}_{\alpha}, S_{\alpha}, \hat{S}_{\alpha}\} . \end{cases}$$

²We lower and raise Greek indices using $\epsilon_{\alpha\beta} = i\sigma_{\alpha\beta}^2$ and $\epsilon^{\alpha\beta} = i\sigma^{2\,\alpha\beta}$ with $\epsilon^{\alpha\gamma}\epsilon_{\gamma\beta} = \delta^{\alpha}{}_{\beta}$, so that for example $\sigma^{1\,\alpha}{}_{\beta} := i\sigma^{2\,\alpha\gamma}\sigma_{\gamma\beta}^1 = -i\sigma^{1\,\alpha\gamma}\sigma_{\gamma\beta}^2$.

The bosonic generators P, K, D, and L_a are skew-Hermitian while $(R^1_1)^{\dagger} = R^2_2$ and $(R^1_2)^{\dagger} = -R^2_1$. The following reality conditions hold for the fermionic generators: $Q_1^{\dagger} = \hat{Q}_2$, $Q_2^{\dagger} = -\hat{Q}_1$ and $S_1^{\dagger} = -\hat{S}_2$, $S_2^{\dagger} = \hat{S}_1$. The superalgebra (5.14) and (5.15) are invariant under these reality conditions. When $c^2 \to 0$ we recover the superalgebra $\mathfrak{psu}(1,1|2)$ associated with $AdS_2 \times S^2 \times T^6$ since in that limit the generators L_{\pm} and L_3 decouple. This is to be expected since in the limit $\alpha \to 0$ (which is $c^2 \to 0$) we recover the non-exceptional background from its exceptional counterpart. Additionally, non-vanishing components of the invariant form of $\mathfrak{d}(2,1;c^2)$ which are compatible with the above choice of basis are

$$\operatorname{Str}(PK) = 2 , \quad \operatorname{Str}(DD) = 1 ,$$

$$\operatorname{Str}(L_{+}L_{-}) = -\frac{2}{c^{2}} , \quad \operatorname{Str}(L_{3}L_{3}) = \frac{1}{c^{2}} ,$$

$$\operatorname{Str}(R^{\alpha}{}_{\beta}R^{\gamma}{}_{\delta}) = \frac{2}{s^{2}} \left(\delta^{\alpha}{}_{\delta}\delta^{\gamma}{}_{\beta} - \frac{1}{2}\delta^{\alpha}{}_{\beta}\delta^{\gamma}{}_{\delta} \right) ,$$

$$\operatorname{Str}(Q_{\alpha}S_{\beta}) = -2\sigma^{2}_{\alpha\beta} , \quad \operatorname{Str}(\hat{Q}_{\alpha}\hat{S}_{\beta}) = 2\sigma^{2}_{\alpha\beta} .$$
(5.16)

\mathbb{Z}_4 -grading and order-4 Automorphism

To formulate the supercoset action based on (5.12), we need to fix a \mathbb{Z}_4 -grading of the superalgebra $\mathfrak{d}(2,1;c^2) \otimes \mathbb{C} \cong \bigoplus_{m=0}^3 \mathfrak{g}_{(m)}$. Our decomposition is as follows

$$\mathfrak{g}_{(0)} := \langle P + K, L_{+} + L_{-}, \sigma_{\gamma[\alpha}^{1} R^{\gamma}{}_{\beta]} \rangle,$$

$$\mathfrak{g}_{(1)} := \langle Q_{\alpha} - \sigma^{1\beta}{}_{\alpha}S_{\beta}, \hat{Q}_{\alpha} - \sigma^{1\beta}{}_{\alpha}\hat{S}_{\beta} \rangle,$$

$$\mathfrak{g}_{(2)} := \langle P - K, D, L_{+} - L_{-}, L_{3}, \sigma_{\gamma(\alpha}^{1} R^{\gamma}{}_{\beta)} \rangle,$$

$$\mathfrak{g}_{(3)} := \langle Q_{\alpha} + \sigma^{1\beta}{}_{\alpha}S_{\beta}, \hat{Q}_{\alpha} + \sigma^{1\beta}{}_{\alpha}\hat{S}_{\beta} \rangle,$$
(5.17)

where brackets (respectively, parentheses) indicate normalised anti-symmetrisation (respectively, symmetrisation) of the enclosed indices. Note that $\mathfrak{g}_{(0)} \cong \mathfrak{so}(1,1) \oplus \mathfrak{so}(2) \oplus \mathfrak{so}(2)$ as required. The order-4 automorphism $\Omega: \mathfrak{d}(2,1;c^2) \to \mathfrak{d}(2,1;c^2)$ associated with this \mathbb{Z}_4 -grading is given by

$$\begin{split} \Omega(P) \ &= \ K \ , \ \ \Omega(K) \ = \ P \ , \ \ \Omega(D) \ = \ -D \ , \\ \Omega(L_3) \ &= \ -L_3 \ , \ \ \Omega(L_{\pm}) \ = \ L_{\mp} \ , \ \ \Omega(R^{\alpha}{}_{\beta}) \ = \ \sigma^{1\,\alpha\gamma}\sigma^1_{\beta\delta}R^{\delta}{}_{\gamma} \ , \\ \Omega(Q_{\alpha}) \ &= \ -\mathrm{i}\sigma^{1\,\beta}{}_{\alpha}S_{\beta} \ , \ \ \Omega(\hat{Q}_{\alpha}) \ = \ -\mathrm{i}\sigma^{1\,\beta}{}_{\alpha}\hat{S}_{\beta} \ , \\ \Omega(S_{\alpha}) \ &= \ -\mathrm{i}\sigma^{1\,\beta}{}_{\alpha}Q_{\beta} \ , \ \ \Omega(\hat{S}_{\alpha}) \ = \ -\mathrm{i}\sigma^{1\,\beta}{}_{\alpha}\hat{Q}_{\beta}. \end{split}$$

Furthermore,

$$\operatorname{Str}[(P \pm K)(P \pm K)] = \pm 4, \quad \operatorname{Str}(DD) = 1,$$

$$\operatorname{Str}[(L_{+} \pm L_{-})(L_{+} \pm L_{-})] = \mp \frac{4}{c^{2}}, \quad \operatorname{Str}(L_{3}L_{3}) = \frac{1}{c^{2}},$$

$$\operatorname{Str}[(\sigma_{\mu[\alpha}^{1}R^{\mu}{}_{\beta]})(\sigma_{\nu[\gamma}^{1}R^{\nu}{}_{\delta]})] = -\frac{1}{s^{2}}\sigma_{\alpha\beta}^{2}\sigma_{\gamma\delta}^{2},$$

$$\operatorname{Str}[(\sigma_{\mu(\alpha}^{1}R^{\mu}{}_{\beta)})(\sigma_{\nu(\gamma}^{1}R^{\nu}{}_{\delta)})] = -\frac{1}{s^{2}}(\sigma_{\alpha\beta}^{1}\sigma_{\gamma\delta}^{1} - \sigma_{\alpha\gamma}^{1}\sigma_{\beta\delta}^{1} - \sigma_{\alpha\delta}^{1}\sigma_{\gamma\beta}^{1}),$$

$$\operatorname{Str}[(Q_{\alpha} \pm \sigma^{1}{}_{\alpha}{}_{\alpha}S_{\gamma})(Q_{\beta} \mp \sigma^{1}{}_{\beta}S_{\delta})] = \mp 4\mathrm{i}\sigma_{\alpha\beta}^{1},$$

$$\operatorname{Str}[(\hat{Q}_{\alpha} \pm \sigma^{1}{}_{\alpha}{}_{\alpha}\hat{S}_{\gamma})(\hat{Q}_{\beta} \mp \sigma^{1}{}_{\beta}{}_{\beta}\hat{S}_{\delta})] = \pm 4\mathrm{i}\sigma_{\alpha\beta}^{1},$$

$$(5.18)$$

93

which follow from (5.16).

Coset Representative and Associated Current

As always, the next step involves choosing a coset representative for the supercoset space (5.12) based on the superalgebra and \mathbb{Z} -grading. From (5.14) we can see that the generators P, Q_{α} , and L_{+} are in involution and therefore form an Abelian subalgebra³. Consequently, the associated directions are those along which we will perform T-dualisation. Following our general discussion in section 4.3, an appropriate form of the coset representative is⁴

$$g := e^{xP + \theta^{\alpha}Q_{\alpha} + \lambda_{+}L_{+}} e^{B} e^{\xi^{\alpha}S_{\alpha}} ,$$

$$e^{B} := e^{\hat{\theta}^{\alpha}\hat{Q}_{\alpha} + \hat{\xi}^{\alpha}\hat{S}_{\alpha}} |y|^{D} e^{-\lambda_{3}L_{3}} e^{-\rho^{\beta}} e^{R^{\alpha}} \beta .$$
(5.19)

where we assume that both λ_+ and λ_3 are complex⁵. We are therefore essentially dealing with the complexification $SL(2,\mathbb{C})/\mathbb{C}^*$ of the coset $SO(3)/SO(2) \cong SU(2)/U(1) \cong S^2$. From the perspective of fermionic T-duality, such a complexification is rather natural (see [22] for a similar case in $AdS_5 \times S^5$). Note that the resulting line element (as seen in Chapter 3) on $SL(2,\mathbb{C})/\mathbb{C}^*$ is

$$ds^{2} = \frac{1}{4c^{2}} \left[(d\lambda_{3})^{2} + e^{2i\lambda_{3}} (d\lambda_{+})^{2} \right].$$
(5.20)

Upon performing the change of coordinates $(\lambda_+, \lambda_3) \mapsto (\varphi, \vartheta)$,

$$\lambda_{+} = \frac{2 \tan(\frac{\vartheta}{2}) \sin(\varphi)}{1 + 2i \tan(\frac{\vartheta}{2}) \cos(\varphi) - \tan^{2}(\frac{\vartheta}{2})} ,$$

$$e^{-i\lambda_{3}} = \frac{1 + \tan^{2}(\frac{\vartheta}{2})}{1 + 2i \tan(\frac{\vartheta}{2}) \cos(\varphi) - \tan^{2}(\frac{\vartheta}{2})}$$
(5.21)

for $\varphi, \vartheta \in \mathbb{C}$, we find the line element

$$(\mathrm{d}s)^2 = \frac{1}{4c^2} \left[(\mathrm{d}\vartheta)^2 + \sin^2(\vartheta) \, (\mathrm{d}\varphi)^2 \right] \,, \tag{5.22}$$

which, when considering the real slice $\varphi^* = \varphi$ and $\vartheta^* = \vartheta$, becomes the standard line element on the two-sphere S^2 . The Maurer–Cartan form $J = g^{-1}dg$ corresponding to the coset representative (5.19) is of the form

³The maximal Abelian subalgebra for $\mathfrak{d}(2,1;c^2)$ has two bosonic and two fermionic generators.

⁴To arrive at the coset representative for $AdS_2 \times S^2 \times T^2$ from the representative (5.19) in the limit $c \to 0$, one first needs to re-scale the coordinates $\lambda_+ \to c\lambda_+$, $\lambda_3 \to c\lambda_3$, and $\rho^{\alpha}{}_{\beta} \to s\rho^{\alpha}{}_{\beta}$ first. Then, we perform $c \to 0$. Upon taking this limit, the second sphere S^2 , whose metric becomes flat, decouples from the $AdS_2 \times S^2$ supercoset and re-compactifies into T^2 which is part of the larger T^6 .

⁵This technical assumption is used to help ease the T-duality transformations considered herein.

$$J = e^{-\xi^{\alpha} S_{\alpha}} J^{(0)} e^{\xi^{\alpha} S_{\alpha}} + d\xi^{\alpha} S_{\alpha}$$

= $J^{(0)} - \xi^{\alpha} [S_{\alpha}, J^{(0)}] + \frac{i}{4} \xi^{2} \sigma^{2 \alpha \beta} \{ S_{\alpha}, [S_{\beta}, J^{(0)}] \} + d\xi^{\alpha} S_{\alpha} ,$ (5.23)

where, as before, $J^{(0)}$ does not depend on the fermionic coordinate ξ^{α} , and we have set $\xi^2 := i\sigma_{\alpha\beta}^2 \xi^{\alpha} \xi^{\beta}$. The explicit form of the components of the current J are given in Appendix B. Using the \mathbb{Z}_4 -grading (5.17), the coset current J decomposes according to $J = J_{(0)} + J_{(1)} + J_{(2)} + J_{(3)}$ with

$$\begin{aligned} J_{(0)} &= \frac{1}{2} (J_P + J_K) (P + K) + \frac{1}{2} (J_{L_+} + J_{L_-}) (L_+ + L_-) - J_{R^{\alpha}{}_{\beta}} \sigma^{1\,\alpha\gamma} \sigma^{1}_{\delta[\gamma} R^{\delta}{}_{\beta]} ,\\ J_{(1)} &= \frac{1}{2} (J_{Q_{\alpha}} - \sigma^{1\,\alpha}{}_{\beta} J_{S_{\beta}}) (Q_{\alpha} - \sigma^{1\,\beta}{}_{\alpha} S_{\beta}) \frac{1}{2} (J_{\hat{Q}_{\alpha}} - \sigma^{1\,\alpha}{}_{\beta} J_{\hat{S}_{\beta}}) (\hat{Q}_{\alpha} - \sigma^{1\,\beta}{}_{\alpha} \hat{S}_{\beta}) ,\\ J_{(2)} &= \frac{1}{2} (J_P - J_K) (P - K) + J_D D + \frac{1}{2} (J_{L_+} - J_{L_-}) (L_+ - L_-) + J_{L_3} L_3 - \\ &- J_{R^{\alpha}{}_{\beta}} \sigma^{1\,\alpha\gamma} \sigma^{1}_{\delta(\gamma} R^{\delta}{}_{\beta)} ,\\ J_{(3)} &= \frac{1}{2} (J_{Q_{\alpha}} + \sigma^{1\,\alpha}{}_{\beta} J_{S_{\beta}}) (Q_{\alpha} + \sigma^{1\,\beta}{}_{\alpha} S_{\beta}) + \frac{1}{2} (J_{\hat{Q}_{\alpha}} + \sigma^{1\,\alpha}{}_{\beta} J_{\hat{S}_{\beta}}) (\hat{Q}_{\alpha} + \sigma^{1\,\beta}{}_{\alpha} \hat{S}_{\beta}) . \end{aligned}$$
(5.24)

Supercoset Action

Using the \mathbb{Z}_4 -grading (5.17) together with the currents (5.24) and the invariant form (5.16), the σ -model action becomes

$$S = -\frac{T}{2} \int_{\Sigma} \left\{ -*(J_{P} - J_{K}) \wedge (J_{P} - J_{K}) + *J_{D} \wedge J_{D} + \frac{1}{c^{2}} * (J_{L_{+}} - J_{L_{-}}) \wedge (J_{L_{+}} - J_{L_{-}}) + \frac{1}{c^{2}} * J_{L_{3}} \wedge J_{L_{3}} + \frac{1}{s^{2}} (*J_{R^{\alpha}{}_{\beta}} \wedge J_{R^{\beta}{}_{\alpha}} - \sigma^{1}{}^{\alpha\gamma} \sigma^{1}_{\beta\delta} * J_{R^{\alpha}{}_{\beta}} \wedge J_{R^{\gamma}{}_{\delta}}) - i\sigma^{1}_{\alpha\beta} (J_{Q_{\alpha}} \wedge J_{Q_{\beta}} + J_{S_{\alpha}} \wedge J_{S_{\beta}} - J_{\hat{Q}_{\alpha}} \wedge J_{\hat{Q}_{\beta}} - J_{\hat{S}_{\alpha}} \wedge J_{\hat{S}_{\beta}}) \right\}.$$
(5.25)

For the 'non-exceptional' cases, the matrix $(-\sigma^1)$ can be identified with the matrix $\Gamma^4\mathbb{P}$ along the AdS_2 radial direction with \mathbb{P} being the projector which singles out eight unbroken supersymmetries of the background under consideration. The same is true for the exceptional cases.

T-dualisation

We are now ready to perform the T-duality on the action (5.25) according to the procedure laid out in section 4.3. After some technical algebra, a field re-definition and the use of the Maurer-Cartan equations we find that the resulting dual action has the same form as the initial action, but with currents constructed with the different coset element

$$\tilde{g} := e^{\tilde{x}K - i\sigma^{2\,\alpha\beta}\tilde{\theta}_{\alpha}S_{\beta} + \tilde{\lambda}_{+}L_{-}} e^{B} e^{\sigma^{1\beta}{}_{\alpha}\xi^{\alpha}Q_{\beta}}, \qquad (5.26)$$

where, e^B is the same as in (5.19). Therefore, the supercoset sigma model on $AdS_2 \times S^2 \times S^2$ is self-dual

under the combined T-dualities along x, θ^{α} , and λ_+ . As a check, in the limit $c^2 \to 0$, upon an appropriate re-scaling of the J_L -currents, the action reduces to the $\frac{PSU(1,1|2)}{SO(1,1) \times U(1)}$ supercoset σ -model for $AdS_2 \times S^2$. In this limit, the dualised sphere S^2 gets 'decompactified' into a T^2 torus which completely decouples from the $AdS_2 \times S^2$ and fermionic sector.

5.3.2 Self-duality for $AdS_3 \times S^3 \times S^3$

Considering the subsector of the $AdS_3 \times S^3 \times S^3 \times S^1$ theory in which the string moves only in $AdS_3 \times S^3 \times S^3 \times S^3$ while its non-supersymmetric fermionic modes are gauge fixed to zero and the S^1 -fluctuations decouple from the rest. For this case, the T-dualisation process is almost identical to the discussion above for the $AdS_2 \times S^2 \times S^2$ case, however, the explicit calculations are technically more involved. Therefore, we only outline the basic steps here.

The supercoset sigma model on $AdS_3 \times S^3 \times S^3$ is based on the supercoset

$$\frac{\mathrm{D}(2,1;c^2) \times \mathrm{D}(2,1;c^2)}{SO(1,2) \times SO(3) \times SO(3)} .$$
(5.27)

The Lie superalgebra $\mathfrak{d}(2,1;c^2) \oplus \mathfrak{d}(2,1;c^2)$ has $\{P_m, D, K_m, L_a^{\pm}, R^{\pm i}{}_j\}$ for m = 0, 1, a = 1, 2, 3, and i, j = 1, 2 as its bosonic generators and $\{Q_{i\alpha}, S_{i\alpha}, \hat{Q}_{i\alpha}, \hat{S}_{i\alpha}\}$ for $\alpha = 1, 2$ as its fermionic generators, respectively. Here, the L_a^{\pm} and $R^{\pm i}{}_j$ are the generators of $\mathfrak{so}(3) \oplus \mathfrak{so}(3) \oplus \mathfrak{so}(3) \oplus \mathfrak{so}(3)$. Furthermore, the generators $\{P_m, Q_{i\alpha}, L^{\pm} := iL_1^{\pm} + L_2^{\pm}\}$ are in involution⁶ so that the coset representative will have the left factor of the form $e^{x^m P_m + \theta^{i\alpha} Q_{i\alpha} + \lambda_+ L^+ + \lambda_- L^-}$. The coordinates x^m parametrize the 2-dimensional Minkowski boundary of AdS_3 . As for the $AdS_2 \times S^2 \times S^2$ case, we shall work with the complexification $SO(4, \mathbb{C})/SO(3, \mathbb{C})$ of $SO(4)/SO(3) \cong [SU(2) \times SU(2)]/SU(2) \cong SU(2) \cong S^3$. Thus, the coordinates λ_{\pm} are assumed to be complex. The resulting line element on $SO(4, \mathbb{C})/SO(3, \mathbb{C})$ will be of the form

$$ds^{2} = \frac{1}{4c^{2}} \left[(d\lambda_{3})^{2} + e^{2i\lambda_{3}} (d\lambda_{+})^{2} + e^{2i\lambda_{3}} (d\lambda_{-})^{2} \right].$$
(5.28)

The next step is to choose an appropriate \mathbb{Z}_4 -grading for the space (5.27). The T-duality is then performed along the bosonic directions x^m and λ_{\pm} and the fermionic directions $\theta^{i\alpha}$. The T-self-duality of the supercoset σ -model follows. We have explicitly checked this up to the second order in the four-component fermions $\xi^{i\alpha}$, like in the $\operatorname{AdS}_2 \times S^2 \times S^2$ case. We believe that the invariance holds to the highest (4thorder) in $\xi^{i\alpha}$. This is supported by the fact that at $\alpha = 0$, the model reduces to the $\operatorname{AdS}_3 \times S^3$ supercoset sigma model times the torus sector, which have shown to be duality invariant.

⁶Note that the maximal Abelian subalgebra of $\mathfrak{d}(2,1;c^2) \oplus \mathfrak{d}(2,1;c^2)$ has four bosonic and four fermionic generators.

5.4 Combined bosonic-fermionic T-duality of the Ramond-Ramond $AdS_d \times S^d \times M^{10-2d}$ backgrounds

This section provides additional evidence for the T-self-duality of the complete $AdS_3 \times S^3 \times S^3 \times S^1$ theory by proving invariance under the combined bosonic and fermionic T-duality of its supergravity background. We shall apply the T-duality rules directly to the corresponding supergravity component fields. We are therefore extending the earlier results of [16, 22, 23, 101, 107] to the whole class of the Ramond-Ramond $AdS_d \times S^d \times M^{10-2d}$ super-backgrounds. This general approach deals with all the backgrounds considered in this thesis at once.

5.4.1 Rules for Fermionic T-duality

Killing Spinors

Following the procedure laid out in Chapter 1 which was based on the conventional rules [12,13, 129] (and [66] for a generalization to the whole superspace), we may T-dualise the bosonic directions. After this procedure was generalized for fermionic directions in [22], we found out that we could T-dualise along Grassman-even Killing spinors, denoted by $\Xi_{\mu}(X)$, which generate Abelian super-isometries. Here μ labels the number of the Killing spinor. Recall that fermionic T-duality acts on the dilaton $\Phi(X)$ and the RR fields, but leaves the metric and the NSNS 2-form invariant. That we dualise along Grassman-even directions implies the following condition

$$\Xi_{\mu}\Gamma_{A}\Xi_{\nu} = 0 \quad \text{for all} \quad A, \mu, \nu \quad \text{with} \quad A = 0, 1, \dots, 9.$$

$$(5.29)$$

This condition has non-trivial solutions if the Killing spinors are complex, thus manifesting the fact that they are associated with complex Grassmann-odd directions in superspace. The Killing spinor conditions themselves have the following form

$$\partial_M \Xi - \frac{1}{4} \Omega_M^{AB}(X) \Gamma_{AB} \Xi = -\frac{1}{8} \not\!\!\!\!\! E \mathcal{E}_M^A(X) \Gamma_A \Xi ,$$

$$\frac{1}{16} \Gamma_A \not\!\!\!\! E \Gamma^A \Xi = 0$$
(5.30)

$$\not F = \begin{cases} e^{\Phi} \left(\frac{1}{2} F_{AB}^{(2)} \Gamma^{AB} \Gamma_{11} + \frac{1}{4!} F_{ABCD}^{(4)} \Gamma^{ABCD} \right) & \text{type IIA} \\ -\frac{e^{\Phi}}{2} (1 + \Gamma^{11}) \left(i F_A^{(1)} \Gamma^A \sigma^2 + \frac{1}{3!} F_{ABC}^{(3)} \Gamma^{ABC} \sigma^1 + \frac{i}{2 \cdot 5!} F_{A \cdots E}^{(5)} \Gamma^{A \cdots E} \sigma^2 \right) & \text{type IIB} \end{cases}$$

$$(5.31)$$

We arrive at (5.30) by requiring that the supersymmetry variations for the dilatino and gravitino vanish. They are also determined by the individual geometry and its accompanying RR fluxes. Requiring that the first equation in (5.30) be integrable determines the projector $\mathcal{P}_{8(d-1)}$. This singles out 8(d-1) fermionic isometries for the background considered. The second equation in (5.30) is then identically satisfied.

Fermionic T-duality Rules

Upon solving for the Killing spinor equations (5.30), one can derive from⁷

$$\partial_M \mathcal{C}_{\mu\nu} = \begin{cases} E_M^A \bar{\Xi}_\mu \Gamma_A \Gamma^{11} \Xi_\nu & \text{type IIA} \\ E_M^A \bar{\Xi}_\mu \Gamma_A \sigma^3 \Xi_\nu & \text{type IIB} \end{cases}$$
(5.32)

the matrix $\mathcal{C} = (\mathcal{C}_{\mu\nu}(X))$ which is formed by the components of the NSNS 2-form B_2 along the Abelian fermionic isometries, that is,

$$d\theta^{\mu} \wedge d\theta^{\nu} B_{\mu\nu}(X, \Theta)|_{\Theta=0} := d\theta^{\mu} \wedge d\theta^{\nu} \mathcal{C}_{\mu\nu}(X).$$
(5.33)

Once the matrix $\mathcal{C}_{\mu\nu}$ is known, one obtains a shift of the dilaton under fermionic T-duality

$$\Delta \Phi = \Phi' - \Phi = \frac{1}{2} \log(\det \mathcal{C}) \tag{5.34}$$

and of the RR fields, which in our conventions is

where Γ is a certain product of Γ -matrices that is used to split the fermionic $E^{(1,2)}$ currents into four pieces corresponding to the splitting up of the superalgebra generators Q into Q, \hat{Q} , S, and \hat{S} , respectively. In particular, backgrounds with a 5-flux alone, have $\Gamma = 1$. For $AdS_d \times S^d \times M^{10-2d}$ (with d = 2, 3) with 3-flux, we have $\Gamma = -\Gamma^{23}$. For backgrounds with both 2- and 4-fluxes, we have $\Gamma = \Gamma^{11}\Gamma^{123}$, while for backgrounds with 4-flux only, we have $\Gamma = \Gamma^1$.

Explicit Form of the Killing Spinors

A direct way to get the explicit form of the Killing spinors is to read them off from the corresponding components of the fermionic currents J_Q associated with the generators Q of the superisometry algebra. By construction, the Killing spinors satisfy the defining relations (5.29) and (5.30) and are the components of $J_{\alpha}{}^{\beta}(|y|, y^{\hat{a}}, \lambda_3)$ in

⁷Equations (5.32) determine the components H_M of the field strength $H_3 = dB_2$ of the NSNS 2-form for the super-backgrounds under consideration.
$$J_{Q_{\alpha}}|_{\Theta=0} = \mathrm{d}\theta^{\mu}J_{\mu}^{\alpha}(|y|, y, \lambda_{3}) = \mathrm{d}\theta^{\mu}\mathrm{e}^{-B}Q_{\mu}\mathrm{e}^{B}|_{Q_{\alpha},\hat{\theta}=\hat{\xi}=0} \stackrel{!}{=} \mathrm{d}\theta^{\mu}\Xi_{\mu}^{\alpha}, \tag{5.36}$$

where e^B was defined in the last section . The Killing spinor condition, a particular form of (5.30), is obtained by differentiation of (5.36)

$$d\Xi_{\mu} + \left[e^{-B} de^{B}, \Xi_{\mu} \right] |_{\Theta=0} = 0 ,$$

$$e^{-B} de^{B} |_{\Theta=0} = \Omega^{\hat{a}\hat{b}}(y/|y|) R_{\hat{a}\hat{b}} + J_{D}(|y|) D + J_{L_{3}}(\lambda_{3}) L_{3} .$$
(5.37)

Note that the index μ should be regarded as an external one, labelling the number of each Killing spinor. Considering the structure of the coset element $e^{B(|y|,y,\lambda_3)}$ and the commutation relations $[D,Q] = \frac{1}{2}Q$, $[R_{\hat{a}},Q] = -\frac{s^2}{2}Q\Gamma_{\hat{a}}\Gamma^4\mathbb{P}$, and $[L_3,Q] = \frac{1}{2}Q$, we have the following generic form of the Killing spinors in question⁸

$$\Xi_{\mu}^{\ \alpha} = J_{\mu}^{\ \alpha}(|y|, y, \lambda_3) = |y|^{-\frac{1}{2}} e^{\frac{1}{2}c\lambda_3} \mathcal{O}_{\mu}^{\ \alpha}(y^{\hat{a}}/|y|), \tag{5.38}$$

where $\mathcal{O}_{\mu}{}^{\alpha}(y^{\hat{a}}/|y|) := (e^{s\mathbb{P}\Gamma_{\hat{a}}\Gamma_{4}}y^{\hat{a}}/(2|y|))_{\mu}{}^{\alpha}$ is a Spin(d+1)-matrix associated with the coset $S^{d} \cong SO(d+1)/SO(d)$ and $\mathbb{P} := \mathbb{P}_{+}\mathcal{P}_{8(d-1)}$ is the projector matrix which singles out the 2(d-1) anti-commuting isometries $Q = Q\mathbb{P}$ for each case of $AdS_{d} \times S^{d} \times M^{10-2d}$, as was described in the previous sections. By definition, we have

$$\mathcal{O}^T \Gamma_4 \mathcal{O} = \Gamma_4 \mathbb{P}, \tag{5.39}$$

The structure of the matrix $C_{\mu\nu}$, see (5.33), is read off from the form of the Wess-Zumino term of the Green-Schwarz superstring action, which in our conventions has the form

$$B_{\mu\nu}|_{\Theta=0} = \mathrm{i} J_{\mu}{}^{\gamma} \Gamma^{4}_{\gamma\delta} J_{\nu}{}^{\delta}(|y|, y, \lambda_{3}) \stackrel{!}{=} \mathcal{C}_{\mu\nu}.$$

$$(5.40)$$

Using (5.38) and (5.39), we find that

$$\mathcal{C}_{\mu\nu} = \mathbf{i}|y|^{-1} \mathrm{e}^{\mathbf{i}c\lambda_3} (\Gamma^4 \mathbb{P})_{\mu\nu}$$
(5.41)

and its inverse is

$$\mathcal{C}^{-1\mu\nu} = -\mathbf{i}|y|\mathbf{e}^{-\mathbf{i}c\lambda_3}(\mathbb{P}\Gamma^4)^{\mu\nu}.$$
(5.42)

From (5.41) we can read off the shift (5.34) of the dilaton

⁸To have a smooth limit from $AdS_d \times S^d \times S^d \times T^{10-3d}$ to $AdS_d \times S^d \times T^{10-2d}$ at $c \to 0$, we had rescale the coordinates λ_{\pm} and λ_3 of the second sphere.

$$\Delta \Phi = \frac{1}{2} \log(\det \mathcal{C}) = -(d-1) \log |y| + i(d-1)c\lambda_3$$
(5.43)

and from (5.42) we read off the change (5.35) in the RR fluxes upon the fermionic T-duality for all the considered cases

$$\Delta F = 8J_{\mu}\mathcal{C}^{-1\mu\nu}J_{\nu}\Gamma = -8i\mathbb{P}\Gamma^{4}\Gamma = -(1+i\Gamma^{0123})\not\!\!\!/ \, \mathcal{I}.$$
(5.44)

Explicit Form of the Ramond-Ramond Fluxes

In the interest of completeness, let us present more details on the form of the RR fluxes characterized by (5.44) in some of the exceptional $AdS_d \times S^d \times S^d \times T^{10-3d}$ cases:

(a) $AdS_3 \times S^3 \times S^3 \times S^1$: Consider type IIB theory with a 3-flux as in [127], for example

$$I_{3}^{\mu} = 2\left(\Gamma^{014} + \sqrt{\alpha}\Gamma^{823} + \sqrt{1-\alpha}\Gamma^{567}\right) = 4\mathcal{P}_{16}\Gamma^{014} .$$
(5.45)

In this case, $\Gamma = -\Gamma^{23}$ as in the corresponding non-exceptional $\alpha = 0$ case. Alternatively, we may T-dualise this background along the S^1 -coordinate φ^9 to get the IIA background with 4-flux only, as written in [121], and use the same \mathcal{P}_{16} with $\Gamma = \Gamma^{239}$.

(b) $AdS_2 \times S^2 \times S^2 \times T^4$: Consider type IIA theory with a 4-flux, then we may write the corresponding projector of rank 8 as a product of two rank 16 projectors, $\mathcal{P}_8 = P_1P_2$, as in [121]. Re-numbering the 4-flux components of [121] such that $0, \ldots, 3$ are the directions along which we dualise (with 2, 3, 8, 9 the T^4 directions, one sphere being parametrised by $x^7 = \lambda_3$ and $x^1 = \lambda_+$, and the other sphere directions labelled by 5, 6) this reads

with $\Gamma = \Gamma^{239}$.

In all the cases under consideration, the shifts (5.43) and (5.44) are undone by the corresponding bosonic T-dualities, as we shall show next for the $AdS_d \times S^d \times M^{10-2d}$ backgrounds.

5.4.2 Compensating Bosonic T-duality

General Case

The complete Buscher rules for bosonic T-duality are part of the O(D, D) symmetry of generalised geometry [133]. However, in the cases of interest to us here, the antisymmetric NSNS *B*-field vanishes and the metric is diagonal, simplifying the rules greatly. Let \mathcal{I} be the set of directions along which we dualise, then the new metric has the components

$$G'_{tt} = \frac{1}{G_{tt}} , \qquad t \in \mathcal{I}$$

and remains unchanged in all other directions. The shift in the dilaton is given by (minus half of the) log of the determinant of this block, that is

$$\Delta \Phi = -\frac{1}{2} \log \det G_{MN} = -\frac{1}{2} \sum_{t \in \mathcal{I}} \log G_{tt}.$$
(5.47)

Allowing for a some abuse in notation that t refers to flat directions here, we can write the change in the RR forms as

$$\mathbf{I}^{\prime\prime\prime} = \left(\prod_{t\in\mathcal{I}} c_t \,\Gamma^t\right) \mathbf{I}^{\prime\prime}, \qquad c_t := \begin{cases} -i & \text{for } t=0\\ 1 & \text{else,} \end{cases}$$
(5.48)

$$\mathbf{F}'' = -\mathrm{i}\Gamma^{0123}\mathbf{F}'.$$

The AdS_d Metric

Consider the T-dualisation of the background metric and the dilaton. With our choice of the coset element and corresponding $AdS_d \times S^d$ metric, the effect of applying T-duality along all d-1 boundary directions of AdS_d on the line element on AdS_d is

$$ds^{2} = \frac{-(\mathrm{d}x^{0})^{2} + \sum_{i=1}^{d-2} \mathrm{d}x^{i} \mathrm{d}x^{i} + \sum_{r=1}^{d+1} \mathrm{d}y^{r} \mathrm{d}y^{r}}{|y|^{2}}$$
$$\to |y|^{2} \Big[-(\mathrm{d}x^{0})^{2} + \sum_{i=1}^{d-2} \mathrm{d}x^{i} \mathrm{d}x^{i} \Big] + \frac{\sum_{r=1}^{d+1} \mathrm{d}y^{r} \mathrm{d}y^{r}}{|y|^{2}}$$

and the dilaton shift is

$$\Delta_{AdS}\Phi = (d-1)\log|y|$$

We can return the metric to its original form by defining $y'^r = y^r/|y|$ which sends $|y| = \sqrt{\sum_{r=1}^{d+1} y^r y^r} \rightarrow \frac{1}{|y|}$. Dualising along some torus directions has no effect on the metric or the dilaton.

The S^d Metric

In the exceptional cases $AdS_d \times S^d \times S^d \times T^{10-3d}$ we also dualise along some directions of one of the spheres: λ_+ for d = 2, and λ_{\pm} for d = 3. The effect on the line element on S^d , after the usual rescaling,

$$ds^{2} = \frac{1}{4} \left[(d\lambda_{3})^{2} + e^{2ic\lambda_{3}} (d\lambda_{+})^{2} + \underbrace{e^{2ic\lambda_{3}} (d\lambda_{-})^{2}}_{\text{only for } d=3} \right]$$
$$\rightarrow \frac{1}{4} \left[(d\lambda_{3})^{2} + e^{-2ic\lambda_{3}} (d\lambda_{+})^{2} + \underbrace{e^{-2ic\lambda_{3}} (d\lambda_{-})^{2}}_{\text{only for } d=3} \right]$$

and we recover the original metric by defining $\lambda'_3 = -\lambda_3$. The effect on the dilaton is

$$\Delta_S \Phi = -i(d-1)c\lambda_3 + (d-1)\log 2.$$
(5.49)

Finally, $\Delta_{AdS} \Phi + \Delta_S \Phi$ cancels the fermionic dualitys shift in the dilaton, (5.43) (modulo the constant term which can be ignored). Since the only contribution from the dilaton is as an overall factor multiplying the action in the path integral, it will not affect the classical supergravity argument. In summary, we have shown that the $AdS_d \times S^d \times M^{10-2d}$ backgrounds with (d = 2, 3, 5) are invariant under the combined fermionic-bosonic T-duality.

5.5 Summary

This chapter dealt with the less than maximally supersymmetric backgrounds in the same way that $AdS_5 \times S^5$ was treated at the end of Chapter 4. Our goal was to establish T-self-duality for such backgrounds with and without κ -symmetry gauge fixing. To this end, we were successful. We showed that the background $AdS_3 \times S^3 \times T^4$ was self-dual without fixing κ -symmetry gauge up to second order in ξ , and argued the same was true up to highest (4th order) in ξ . Furthermore, we showed that the exceptional backgrounds $AdS_d \times S^d \times S^d \times M^{10-3d}$ (for d = 2, 3) were self-dual in a gauge in which the non-supercoset fermions were fixed to zero, either by hand or using κ -symmetry.

Part IV

Summary and Conclusion

Conclusion

An integrable system is one which is exactly solvable, that is, we can obtain explicit solutions. In the context of string theory and field theory, there are an finite number of conserved charges. Here, integrability implies that there is an associated infinite dimensional algebra. Consider $AdS_5 \times S^5$ which is integrable and dual to $\mathcal{N} = 4$ super Yang-Mills. The presence of integrability makes it reasonable to expect that a complete solution for the spectrum of planar scattering amplitudes of $AdS_5 \times S^5$ may be obtained [21]. Similarly, this is the case for $\mathcal{N} = 4$ super Yang-Mills and its spectrum of dimensions of the gauge invariant generators. Moreover, integrability is expected to be connected with self-duality in the some way. This is not yet well understood [22]. Simply put, integrability places tight constraints on a theory. We would like to see the conserved charges put to use to completely fix all amplitudes. In [26], T-self-duality was used for the AdS_5 case to construct classical solutions for open strings which were related to strong coupling limits of gluon scattering amplitudes [21].

Duality is a desirable property, however, it is difficult to show in general. This is where integrability supplements duality. Integrability, as mentioned already, imposes constraints on the theory. Therefore, more integrability will impose more constraints. These constraints will be seen on both sides of the duality. This is also difficult to show. T-duality has the advantage of being algorithmic. This means that it is much easier to implement. In this thesis we have tried to connect the triangle:



Figure 5.2: Triangle of the relationships between various concepts that we have studied.

The central conjecture made in this thesis may be given by the following statement: T-self-duality \iff integrability. This 'if and only if' statement was hinted at in [22], but, is this conjecture true for every 10-dimensional AdS-superstring background? Also, in [22], self-duality is proven to all orders in α' for the integrable background $AdS_5 \times S^5$. The proof was elegant, and $AdS_5 \times S^5$ acts as the ideal laboratory. However, our tests work out too well in this laboratory. With tests of generality looming regarding our proposed conjecture, we face the question: Will our conjecture work for backgrounds with less than

maximal supersymmetry? We needed to determine whether $AdS_5 \times S^5$ was a unique case. Thus, we need to break supersymmetry, systematically. Before moving on, we note that the proof of self-duality in [22] was completed using a partially fixed κ -symmetry gauge. In this thesis, based on the work in [36], we prove that $AdS_5 \times S^5$ is exactly self-dual without fixing κ -symmetry at all.

Breaking a little supersymmetry to start, the next to maximally supersymmetric background we arrive at is $AdS_4 \times \mathbb{C}P^3$. As is the case for the maximally supersymmetric $AdS_5 \times S^5$ case, $AdS_4 \times \mathbb{C}P^3$ is integrable in the planar limit [42]. Furthermore, there is a lot of evidence available which favours the presence of dual superconformal symmetry. Thus, it seems likely that the background would be a good candidate for our conjecture. However, the results are disappointing in this regard. $AdS_4 \times \mathbb{C}P^3$ is not self-dual under a combined set of bosonic and fermionic T-duality. We can say that the background is self-dual classically, but at the quantum level we are at a loss. This means that we cannot use the classical self-duality of $AdS_4 \times \mathbb{C}P^3$ to account for the dual superconformal symmetry of ABJM [42]. This brings us to a key question and its associated conjecture: "Why is $AdS_4 \times \mathbb{C}P^3$ not self-dual, despite $AdS_5 \times S^5$ being self-dual?" It is odd that, for some reason, $AdS_4 \times \mathbb{C}P^3$ fails the conjecture despite its many other similarities to $AdS_5 \times S^5$. Now we have the following logic.

The maximal case has the property that it is self-dual and integrable. However, when considering a background with less supersymmetry like $AdS_4 \times \mathbb{C}P^3$, we find that it is not self-dual. We wanted to test whether the absence of self-duality was as a result of $AdS_4 \times \mathbb{C}P^3$ possessing less than maximal supersymmetry. The work covered in Chapters 3-5 addressed this question. We considered two classes of backgrounds: $AdS_d \times S^d \times T^{10-2d}$ (d = 2, 3) and $AdS_d \times S^d \times T^{10-3d}$ (d = 2, 3). Both backgrounds have less supersymmetry than $AdS_4 \times \mathbb{C}P^3$ and we proved that they were all self-dual from the supergravity and worldsheet perspectives. Therefore, our claim that $AdS_d \times S^d \times T^{10-2d}$ (d = 2, 3) does not work due to less supersymmetry was incorrect. Some other mechanism, which we do not yet understand, is at play.

Recently, Ó Colgain and Pittelli released a fascinating paper [42] dealing with issues surrounding the self-duality of $AdS_4 \times \mathbb{C}P^3$. Interestingly, they showed that $AdS_4 \times \mathbb{C}P^3$ could not be self-dual using either the supergravity or worldsheet approach. Irrespective of the chosen isometries, one encounters a singularity in the dilaton shift. This results because the generators of the Lie algebra do not admit a non-singular order-4 automorphism [42,45,47]. One might then decide to deform the background in such a way that the supersymmetry is preserved, resulting in a new background in the process. This is done via a TsT transformation which involves a Tduality, then a shift (or translation), then a further T-duality. However, as stated in [42], TsT transformations commute with fermionic T-duality. This means that even though we change the background through deformation and without breaking supersymmetry, the conclusion above still holds. This background is not self-dual.

These results preclude the $AdS_4 \times \mathbb{C}P^3$ geometry from being self-dual based on fermionic T-duality. However, a puzzling aspect remains despite this new knowledge. There is still a lot of evidence suggesting that there should be a self-duality transformation at work, based on perturbative observations linked to integrability. Thus, what we learn from this is that the T-duality transformations in [22] are not responsible for the self-duality of supercosets represented by quotient groups or the TsT deformations of their $AdS_5 \times S^5$ case. This, despite $AdS_5 \times S^5$ being integrable. Currently, an explanation eludes us.

Moving forward, there remain a number of outstanding questions. What is responsible for the correct self-duality, i.e. what transformations do we require given the supposition that some sort of self-duality is at play? Do we need to use complexified CP^3 coordinates? What happens when we consider non-trivial

105

NSNS 2-forms? Evidently, even after nearly three decades of intense scrutiny, the idea of T-duality in string remains as fruitful an area of research as ever. We look forward to exploring these and more ideas further.

 $\mathbf{Part}~\mathbf{V}$

Appendices

Appendix A

Conventions

A.1 Pauli Matrices

The Pauli matrices are defined as

$$\sigma_1 = \begin{bmatrix} 0 & 1 \\ 1 & 0 \end{bmatrix}, \quad \sigma_2 = \begin{bmatrix} 0 & -i \\ i & 0 \end{bmatrix}, \quad \sigma_3 = \begin{bmatrix} 1 & 0 \\ 0 & -1 \end{bmatrix},$$

with the following algebra

$$\sigma_i \sigma_j = \delta_{ij} \mathbf{1} + 2\epsilon_{ijk} \sigma_k,$$

and the SU(2) generators are given by

$$t^i = \frac{1}{\sqrt{2}}\sigma^i,$$

such that

$$\operatorname{Tr}(t^i t^j) = \delta^{ij}, \quad [t^i, t^j] = i f_k^{ij} t^k = i \sqrt{2} \epsilon_k^{ij} t^k.$$

A.2 Gamma Matrices

We follow the conventions laid out in [16, 82]. Namely, we work with the 32×32 dimensional gamma matrices which represent $\mathbb{R}^{1,9}$ with a mostly plus metric signature, (- + ... +). The Levi-Civita tensor is

defined with $\epsilon_{0...9} = 1$. The gamma matrices have the following structure

$$\Gamma^{0} = i\sigma_{2} \otimes \mathbf{1}_{16} = \begin{bmatrix} 0 & \mathbf{1} \\ -\mathbf{1} & 0 \end{bmatrix}, \quad \Gamma^{m} = \sigma_{1} \otimes \gamma^{m} = \begin{bmatrix} 0 & (\gamma^{m})_{\alpha\beta} \\ (\gamma^{m})_{\alpha\beta} & 0 \end{bmatrix}$$

with $(\Gamma^0)^2 = -1$ and $(\Gamma^m)^2 = 1$. The 16 × 16 γ matrices are as follows

$$\begin{split} \gamma^{1} &= \sigma^{2} \otimes \sigma^{2} \otimes \sigma^{2} \otimes \sigma^{2} \\ \gamma^{2} &= \sigma^{2} \otimes \mathbf{1} \otimes \sigma^{1} \otimes \sigma^{2} \\ \gamma^{3} &= \sigma^{2} \otimes \mathbf{1} \otimes \sigma^{3} \otimes \sigma^{2} \\ \gamma^{4} &= \sigma^{2} \otimes \sigma^{1} \otimes \sigma^{2} \otimes \mathbf{1} \\ \gamma^{5} &= \sigma^{2} \otimes \sigma^{3} \otimes \sigma^{2} \otimes \mathbf{1} \\ \gamma^{6} &= \sigma^{2} \otimes \sigma^{2} \otimes \mathbf{1} \otimes \sigma^{1} \\ \gamma^{7} &= \sigma^{2} \otimes \sigma^{2} \otimes \mathbf{1} \otimes \sigma^{3} \\ \gamma^{8} &= \sigma^{2} \otimes \mathbf{1} \otimes \mathbf{1} \otimes \mathbf{1} \\ \end{split}$$

and the 8-dimensional chirality operator is given by

$$\gamma^{11} = \sigma^3 \otimes \mathbf{1} \otimes \mathbf{1} \otimes \mathbf{1}.$$

These γ^m are all real and symmetric. The charge conjugation matrix is $C = \Gamma^0$. Notice that $\Gamma^{11} \equiv \prod_{m=0}^{9} \Gamma^m = \sigma^3 \otimes \mathbf{1}_{16}$, which is the 10-dimensional chirality operator. The Γ -matrices satisfy the following algebra

$$\{\Gamma^a, \Gamma^b\} = 2\eta^{ab}.$$

A.3 The Killing Spinor Equations

We arrive at the Killing spinor equations by requiring that the supersymmetry variations of the fermions vanish.

A.3.1 Type IIA String Theory

The Killing spinor variations for type IIA string theory are written in terms of the 32-component full spinor E. The gravitino variation is

and the dilatino variation is

A.3.2 Type IIB String Theory

Type IIB string theory contains fermions which are doublets of gravitini and dilatini, which have opposite chirality. In this treatment, we choose the dilatini λ and $\hat{\lambda}$ to have negative chirality, whilst the gravitini ψ and $\hat{\psi}$ have positive chirality. The Killing spinors, or supersymmetry parameters, ϵ and $\hat{\epsilon}$ also have positive chirality. Then the gravitino variations (in two-component form) are

and the dilatino variations (in two-component form) are

$$\begin{split} \delta\lambda = & \not \! \partial \phi \epsilon - \frac{1}{2} \not \! H \epsilon + \frac{e^{\phi}}{2} \left(2 \not \! F_{(1)} + \not \! F_{(3)} \right) \hat{\epsilon}, \\ \delta \hat{\lambda} = & \not \! \partial \phi \hat{\epsilon} + \frac{1}{2} \not \! H \hat{\epsilon} - \frac{e^{\phi}}{2} \left(2 \not \! F_{(1)} - \not \! F_{(3)} \right) \epsilon, \end{split}$$

where

$$\mathbf{F}_{(n)} = \frac{1}{n!} F_{m_1 \dots m_n} \Gamma^{m_1 \dots m_n}$$
$$\mathbf{H}_m = \frac{1}{2} H_{mnr} \Gamma^{nr}.$$

A.4 Hodge Duality

The Hodge duality conventions are such that on a p-form in D-dimensions

$$(\star F_p)_{\mu_{p+1}\cdot\mu_D} = \frac{1}{p!}\sqrt{|g|}\epsilon_{\mu_1\dots\mu_D}F^{\mu_1\dots\mu_p}.$$

Note also that $\star \star F_p = s(-1)^{p(D-1)}F_p$ where s is the spacetime signature.

Appendix B

$D(2,1;\alpha)$ Supercoset Currents

This appendix is taken directly from our paper [36].

Due to the structure (5.14) of the superalgebra $\mathfrak{d}(2,1;c^2)$ and the chosen coset element (5.15), the coset currents $J^{(0)}$ and J have the following components (see (5.19)):

$$J_{P}^{(0)} = \left[e^{-B} \left(dxP + d\theta^{\alpha} Q_{\alpha} + d\lambda_{+} L_{+} \right) e^{B} \right]_{P}, \quad J_{K}^{(0)} = 0, \quad J_{D}^{(0)} = \left[e^{-B} de^{B} \right]_{D},$$

$$J_{L_{+}}^{(0)} = \left[e^{-B} \left(dxP + d\theta^{\alpha} Q_{\alpha} + d\lambda_{+} L_{+} \right) e^{B} \right]_{L_{+}}, \quad J_{L_{-}}^{(0)} = 0, \quad J_{L_{3}}^{(0)} = \left[e^{-B} de^{B} \right]_{L_{3}},$$

$$J_{R^{\alpha}{}_{\beta}}^{(0)} = \left[e^{-B} de^{B} \right]_{R^{\alpha}{}_{\beta}},$$

$$J_{Q_{\alpha}}^{(0)} = \left[e^{-B} \left(dxP + d\theta^{\alpha} Q_{\alpha} + d\lambda_{+} L_{+} \right) e^{B} \right]_{Q_{\alpha}}, \quad J_{\hat{Q}_{\alpha}}^{(0)} = \left[e^{-B} de^{B} \right]_{\hat{Q}_{\alpha}},$$

$$J_{S_{\alpha}}^{(0)} = 0, \quad J_{\hat{S}_{\alpha}}^{(0)} = \left[e^{-B} de^{B} \right]_{\hat{S}_{\alpha}}$$
(B.1a)

and

$$J_{P} = J_{P}^{(0)}, \quad J_{Q_{\alpha}} = J_{Q_{\alpha}}^{(0)}, \quad J_{L_{+}} = J_{L_{+}}^{(0)},$$

$$J_{D} = J_{D}^{(0)} + \sigma_{\alpha\beta}^{2} J_{Q_{\alpha}}^{(0)} \xi^{\beta}, \quad J_{L_{3}} = J_{L_{3}}^{(0)} + ic^{2} \sigma_{\alpha\beta}^{2} J_{Q_{\alpha}}^{(0)} \xi^{\beta},$$

$$J_{R^{\alpha}{}_{\beta}} = J_{R^{\alpha}{}_{\beta}}^{(0)} - is^{2} \sigma_{\alpha\gamma}^{2} J_{Q_{\gamma}}^{(0)} \xi^{\beta} - \frac{i}{2} s^{2} \sigma_{\gamma\delta}^{2} \delta^{\alpha}{}_{\beta} J_{Q_{\gamma}}^{(0)} \xi^{\delta},$$

$$J_{\hat{Q}_{\alpha}} = J_{\hat{Q}_{\alpha}}^{(0)} - J_{P}^{(0)} \xi^{\alpha}, \quad J_{\hat{S}_{\alpha}} = J_{\hat{S}_{\alpha}}^{(0)} + J_{L_{+}}^{(0)} \xi^{\alpha},$$

$$J_{K} = -\sigma_{\alpha\beta}^{2} J_{\hat{S}_{\alpha}}^{(0)} \xi^{\beta} + \frac{i}{2} J_{L_{+}}^{(0)} \xi^{2}, \quad J_{L_{-}} = -c^{2} \sigma_{\alpha\beta}^{2} J_{\hat{Q}_{\alpha}}^{(0)} \xi^{\beta} - \frac{i}{2} c^{2} J_{P}^{(0)} \xi^{2},$$

$$J_{S_{\alpha}} = -\frac{1}{2} J_{D}^{(0)} \xi^{\alpha} - \frac{i}{2} J_{L_{3}}^{(0)} \xi^{\alpha} - i J_{R^{\beta}{}_{\alpha}}^{(0)} \xi^{\beta} + d\xi^{\alpha} - \frac{i}{2} s^{2} J_{Q_{\alpha}}^{(0)} \xi^{2}.$$
(B.1b)

In these expressions, we have made all the ξ -dependence explicit.

Dual currents

The Maurer-Cartan form $\tilde{J} = \tilde{g}^{-1} d\tilde{g}$ constructed from the dual coset representative (5.26) is of the form

$$\widetilde{J} = e^{-\sigma^{1\beta}{}_{\alpha}\xi^{\alpha}Q_{\beta}}\widetilde{J}^{(0)}e^{\sigma^{1\beta}{}_{\alpha}\xi^{\alpha}Q_{\beta}} + \sigma^{1\beta}{}_{\alpha}d\xi^{\alpha}Q_{\beta}
= \widetilde{J}^{(0)} - \sigma^{1\beta}{}_{\alpha}\xi^{\alpha}[S_{\beta},\widetilde{J}^{(0)}] - \frac{i}{4}\xi^{2}\sigma^{2\,\alpha\beta}\{S_{\alpha},[S_{\beta},\widetilde{J}^{(0)}]\} + \sigma^{1\beta}{}_{\alpha}d\xi^{\alpha}Q_{\beta},$$
(B.2)

where, as before, $\tilde{J}^{(0)}$ does not depend on the ferminic coordinate ξ^{α} , and we have set $\xi^2 := i\sigma_{\alpha\beta}^2 \xi^{\alpha} \xi^{\beta}$. A calculation similar to the one that led to (B.1) yields

$$\begin{split} \tilde{J}_{P}^{(0)} &= 0 , \quad \tilde{J}_{K}^{(0)} &= \left[e^{-B} \left(d\tilde{x}K - i\sigma^{2\,\alpha\beta} d\tilde{\theta}_{\alpha}S_{\beta} + d\tilde{\lambda}_{+}L_{-} \right) e^{B} \right]_{K} , \quad \tilde{J}_{D}^{(0)} &= \left[e^{-B} de^{B} \right]_{D} , \\ \tilde{J}_{L_{+}}^{(0)} &= 0 , \quad \tilde{J}_{L_{-}}^{(0)} &= \left[e^{-B} \left(d\tilde{x}K - i\sigma^{2\,\alpha\beta} d\tilde{\theta}_{\alpha}S_{\beta} + d\tilde{\lambda}_{+}L_{-} \right) e^{B} \right]_{L_{-}} , \quad \tilde{J}_{L_{3}}^{(0)} &= \left[e^{-B} de^{B} \right]_{L_{3}} , \\ \tilde{J}_{R^{\alpha}}^{(0)} &= \left[e^{-B} de^{B} \right]_{R^{\alpha}_{\beta}} , \\ \tilde{J}_{Q_{\alpha}}^{(0)} &= 0 , \quad \tilde{J}_{Q_{\alpha}}^{(0)} &= \left[e^{-B} de^{B} \right]_{\hat{Q}_{\alpha}} , \\ \tilde{J}_{S_{\alpha}}^{(0)} &= \left[e^{-B} \left(d\tilde{x}K - i\sigma^{2\,\alpha\beta} d\tilde{\theta}_{\alpha}S_{\beta} + d\tilde{\lambda}_{+}L_{-} \right) e^{B} \right]_{S_{\alpha}} , \quad \tilde{J}_{\hat{S}_{\alpha}}^{(0)} &= \left[e^{-B} de^{B} \right]_{\hat{S}_{\alpha}} \end{split}$$
(B.3a)

and

$$\begin{split} \tilde{J}_{K} &= \tilde{J}_{K}^{(0)} , \quad \tilde{J}_{S_{\alpha}} &= \tilde{J}_{S_{\alpha}}^{(0)} , \quad \tilde{J}_{L_{-}} &= \tilde{J}_{L_{-}}^{(0)} , \\ \tilde{J}_{D} &= \tilde{J}_{D}^{(0)} - \mathrm{i}\sigma_{\alpha\beta}^{1}\tilde{J}_{S_{\alpha}}^{(0)}\xi^{\beta} , \quad \tilde{J}_{L_{3}} &= \tilde{J}_{L_{3}}^{(0)} + c^{2}\sigma_{\alpha\beta}^{1}\tilde{J}_{S_{\alpha}}^{(0)}\xi^{\beta} , \\ \tilde{J}_{R^{\alpha}{}_{\beta}} &= \tilde{J}_{R^{\alpha}{}_{\beta}}^{(0)} + s^{2}\left(\sigma_{\alpha\gamma}^{1}\tilde{J}_{S_{\beta}}^{(0)} - \frac{1}{2}\delta^{\beta}{}_{\alpha}\sigma_{\gamma\delta}^{1}\tilde{J}_{S_{\delta}}^{(0)}\right)\xi^{\gamma} , \\ \tilde{J}_{\hat{Q}_{\alpha}} &= \tilde{J}_{\hat{Q}_{\alpha}}^{(0)} + \sigma^{1\,\alpha}{}_{\beta}\tilde{J}_{L_{-}}^{(0)}\xi^{\beta} , \quad \tilde{J}_{\hat{S}_{\alpha}} &= \tilde{J}_{\hat{S}_{\alpha}}^{(0)} - \sigma^{1\,\alpha}{}_{\beta}\tilde{J}_{K}^{(0)}\xi^{\beta} , \\ \tilde{J}_{P} &= -\mathrm{i}\sigma_{\alpha\beta}^{1}\tilde{J}_{\hat{Q}_{\alpha}}^{(0)}\xi^{\beta} - \frac{\mathrm{i}}{2}\tilde{J}_{L_{-}}^{(0)}\xi^{2} , \quad \tilde{J}_{L_{+}} &= -\mathrm{i}c^{2}\sigma_{\alpha\beta}^{1}\tilde{J}_{\hat{S}_{\alpha}}^{(0)}\xi^{\beta} + \frac{\mathrm{i}}{2}c^{2}\tilde{J}_{K}^{(0)}\xi^{2} , \\ \tilde{J}_{Q_{\alpha}} &= \sigma^{1\,\alpha}{}_{\beta}\left(\frac{1}{2}\tilde{J}_{D}^{(0)}\xi^{\beta} + \frac{\mathrm{i}}{2}\tilde{J}_{L_{3}}^{(0)}\xi^{\beta} + \mathrm{d}\xi^{\beta}\right) - \mathrm{i}\sigma^{1\,\beta}{}_{\gamma}\tilde{J}_{R^{\beta}{}_{\alpha}}^{(0)}\xi^{\gamma} + \frac{\mathrm{i}}{2}s^{2}\tilde{J}_{S_{\alpha}}^{(0)}\xi^{2} . \end{split}$$
(B.3b)

As before, in these expressions, we have made all the ξ -dependence explicit. Note that $\tilde{J}_{\hat{Q}_{\alpha}}^{(0)} = J_{\hat{Q}_{\alpha}}^{(0)}$, $\tilde{J}_{\hat{S}_{\alpha}}^{(0)} = J_{D}^{(0)}$, $\tilde{J}_{L_{3}}^{(0)} = J_{L_{3}}^{(0)}$, and $\tilde{J}_{R^{\beta}_{\alpha}}^{(0)} = J_{R^{\beta}_{\alpha}}^{(0)}$, respectively.

Bibliography

- [1] J. Polchinski, String duality: A Colloquium, Rev. Mod. Phys. 68 (1996) 1245[hep-th/9607050].
- [2] C. M. Hull and P. K. Townsend, Unity of superstring dualities, Nucl. Phys. B 438 (1995)109[hep-th/9410167].
- P. K. Townsend, The eleven-dimensional supermembrane revisited, Phys. Lett. B 350 (1995)184[hep-th/9501068].
- [4] E. Witten, String theory dynamics in various dimensions, Nucl. Phys. B 443 (1995) 85[hep-th/9503124].
- [5] C. Vafa, Lectures on strings and dualities, hep-th/9702201.
- [6] J. M. Maldacena, The Large N limit of superconformal field theories and supergravity, Int.J. Theor. Phys. 38 (1999) 1113[Adv. Theor. Math. Phys. 2 (1998) 231][hep-th/9711200].
- [7] G. t Hooft, Introduction to String Theory, Series of Lectures at the Spinoza Institute, University of Utrecht, http://www.phys.uu.nl/~thooft/.
- [8] https://upload.wikimedia.org/wikipedia/commons/c/c0/M-Theory.svg
- [9] A. V. Ramallo, Introduction to the AdS/CFT correspondence, Springer Proc. Phys. 161(2015) 411[arXiv:1310.4319].
- [10] A. Giveon, M. Porrati and E. Rabinovici, Target space duality in string theory, Phys. Rept.244 (1994) 77[hep-th/9401139].
- [11] T. H. Buscher, Quantum Corrections and Extended Supersymmetry in New Models, Phys.Lett. B 159 (1985) 127.
- [12] T. H. Buscher, A Symmetry of the String Background Field Equations, Phys. Lett. B 194(1987) 59.
- T. H. Buscher, Path Integral Derivation of Quantum Duality in Nonlinear Sigma Models, Phys. Lett. B 201 (1988) 466.
- [14] M. B. Green, J. H. Schwarz and E. Witten, Superstring Theory. Vol. 1: Introduction, Cambridge, Uk: Univ. Pr. (1987), (Cambridge Monographs On Mathematical Physics)
- [15] M. B. Green, J. H. Schwarz and E. Witten, Superstring Theory. Vol. 2: Loop Amplitudes, Anomalies And Phenomenology, Cambridge, Uk: Univ. Pr. (1987), (Cambridge Monographs On Mathematical Physics)
- [16] I. Bakhmatov, Fermionic T-duality and U-duality in type II supergravity, arXiv:1112.1983[hep-th].
- [17] K. Kikkawa and M. Yamasaki, Casimir Effects in Superstring Theories, Phys. Lett. B 149(1984) 357.

- [18] N. Sakai and I. Senda, Vacuum Energies of String Compactified on Torus, Prog. Theor. Phys. 75 (1986) 692[Prog. Theor. Phys. 77 (1987) 773].
- [19] G. Itsios, C. Nunez, K. Sfetsos and D. C. Thompson, Non-Abelian T-dualityand the AdS/CFT correspondence:new N = 1 backgrounds, Nucl. Phys. B 873 (2013)1[arXiv:1301.6755].
- [20] H. Dimov, S. Mladenov, R. C. Rashkov and T. Vetsov, Non-abelian T-duality of Pilch-Warner background, Fortsch. Phys. 64 (2016) 657[arXiv:1511.00269 [hep-th]].
- [21] N. Beisert, R. Ricci, A. A. Tseytlin and M. Wolf, *Dual Superconformal Symmetry fromAdS*₅ × S^5 Superstring Integrability, Phys. Rev. D 78 (2008) 126004[arXiv:0807.3228 [hep-th]].
- [22] N. Berkovits and J. Maldacena, Fermionic T-Duality, Dual Superconformal Symmetry, and the Amplitude/Wilson Loop Connection, JHEP 0809 (2008) 062[arXiv:0807.3196].
- [23] E. Ó Colgáin, Fermionic T-duality: A snapshot review, Int. J. Mod. Phys. A 27 (2012)1230032[arXiv:1210.5588 [hep-th]].
- [24] G. t Hooft, A Planar Diagram Theory for Strong Interactions, Nucl. Phys. B 72 (1974)461.
- [25] T. Azeyanagi, M. Fujita and M. Hanada, From the planar limit to M-theory, Phys. Rev.Lett. 110 (2013) 12, 121601[arXiv:1210.3601 [hep-th]].
- [26] L. F. Alday and J. Maldacena, Comments on gluon scattering amplitudes via AdS/CFT, JHEP 0711 (2007) 068[arXiv:0710.1060 [hep-th]].
- [27] J. M. Drummond, G. P. Korchemsky and E. Sokatchev, Conformal properties of four-gluonplanar amplitudes and Wilson loops, Nucl. Phys. B 795 (2008) 385[arXiv:0707.0243 [hep-th]].
- [28] A. Brandhuber, P. Heslop and G. Travaglini, MHV amplitudes in N = 4 super Yang-Millsand Wilson loops, Nucl. Phys. B 794 (2008) 231[arXiv:0707.1153 [hep-th]].
- [29] L. F. Alday and R. Roiban, Scattering Amplitudes, Wilson Loops and the String/GaugeTheory Correspondence, Phys. Rept. 468 (2008) 153[arXiv:0807.1889 [hep-th]].
- [30] J. M. Drummond, Review of AdS/CFT Integrability, Chapter V.2: Dual SuperconformalSymmetry, Lett. Math. Phys. 99 (2012) 481[arXiv:1012.4002 [hep-th]].
- [31] J. M. Drummond, J. Henn, G. P. Korchemsky and E. Sokatchev, *Dual superconformalsymmetry of scattering amplitudes in* N = 4 *super-Yang-Mills theory*, Nucl. Phys. B 828(2010) 317[arXiv:0807.1095 [hep-th]].
- [32] J. M. Drummond, J. Henn, V. A. Smirnov and E. Sokatchev, Magic identities for conformal four-point integrals, JHEP 0701 (2007) 064[hep-th/0607160].
- [33] N. Beisert et al., Review of AdS/CFT Integrability: An Overview, Lett. Math. Phys. 99(2012) 3[arXiv:1012.3982 [hep-th]].
- [34] A. A. Tseytlin, Duality Symmetric Formulation of String World Sheet Dynamics, Phys.Lett. B 242 (1990) 163.
- [35] A. A. Tseytlin, Duality symmetric closed string theory and interacting chiral scalars, Nucl.Phys. B 350 (1991) 395.
- [36] M. C. Abbott, J. Murugan, S. Penati, A. Pittelli, D. Sorokin, P. Sundin, J. Tarrant, M. Wolfand L. Wulff, *T*-duality of Green-Schwarz superstrings on $AdS_d \times S^d \times M^{10-2d}$, JHEP 1512(2015) 104[arXiv:1509.07678 [hep-th]].

- [37] A. Babichenko, B. Stefanski, Jr. and K. Zarembo, Integrability and the AdS₃/CFT₂ correspondence, JHEP 1003 (2010) 058[arXiv:0912.1723 [hep-th]].
- [38] D. Sorokin and L. Wulff, Evidence for the classical integrability of the complete $AdS_4 \times CP^3$ superstring, JHEP 1011 (2010) 143[arXiv:1009.3498 [hep-th]].
- [39] D. Sorokin, A. Tseytlin, L. Wulff and K. Zarembo, Superstrings in $AdS_2 \times S^2 \times T^6$, J. Phys.A 44 (2011) 275401[arXiv:1104.1793 [hep-th]].
- [40] A. Cagnazzo, D. Sorokin and L. Wulff, More on integrable structures of superstrings in $AdS_4 \times CP^3$ and $AdS_2 \times S^2 \times T^6$ superbackgrounds, JHEP 1201 (2012) 004[arXiv:1111.4197[hep-th]].
- [41] P. Sundin and L. Wulff, Classical integrability and quantum aspects of the $AdS_3 \times S^3 \times S^3 \times S^1$ superstring, JHEP 1210 (2012) 109[arXiv:1207.5531 [hep-th]].
- [42] E. Ó Colgáin and A. Pittelli, A Requiem for $AdS_4 \times CP^3$ Fermionic self-T-duality, Phys. Rev.D 94 (2016) no.10, 106006[arXiv:1609.03254 [hep-th]].
- [43] D. Bak and S. J. Rey, Integrable Spin Chain in Superconformal Chern-Simons Theory, JHEP 0810 (2008) 053[arXiv:0807.2063 [hep-th]].
- [44] J. A. Minahan and K. Zarembo, The Bethe ansatz for superconformal Chern-Simons, JHEP0809 (2008) 040[arXiv:0806.3951 [hep-th]].
- [45] G. Arutyunov and S. Frolov, Superstrings on $AdS_4 \times CP^3$ as a Coset Sigma-model, JHEP 0809 (2008) 129[arXiv:0806.4940 [hep-th]].
- [46] B. Stefanski, jr, Green-Schwarz action for Type IIA strings on $AdS_4 \times CP^3$, Nucl. Phys. B808 (2009) 80[arXiv:0806.4948 [hep-th]].
- [47] J. Gomis, D. Sorokin and L. Wulff, The Complete AdS₄×CP³ superspace for the type IIA superstring and D-branes, JHEP 0903 (2009) 015[arXiv:0811.1566 [hep-th]].
- [48] J. Maharana, T-duality of NSR superstring: The worldsheet perspective, Int. J. Mod. Phys.A 27 (2012) 1250140[arXiv:1203.3357 [hep-th]].
- [49] C. V. Johnson, *D-Branes*, Cambridge, Uk: Univ. Pr. (1987), (Cambridge Monographs On Mathematical Physics).
- [50] J. Maharana and J. H. Schwarz, Noncompact symmetries in string theory, Nucl. Phys. B390 (1993) 3[hep-th/9207016].
- [51] D. Tong, String Theory Lectures, arXiv:0908.0333 [hep-th].
- [52] S. F. Hassan, Supersymmetry and the Systematics of T-duality Rotations in Type-II Superstring Theories, Nucl. Phys. Proc. Suppl., 102 (2001) 77.
- [53] S. F. Hassan, T Duality and nonlocal supersymmetries, Nucl. Phys. B 460 (1996) 362[hepth/9504148].
- [54] E. Alvarez, L. Alvarez-Gaume and Y. Lozano, A Canonical approach to duality transformations, Phys. Lett. B 336 (1994) 183[hep-th/9406206].
- [55] A. Giveon, E. Rabinovici and G. Veneziano, *Duality in String Background Space*, Nucl.Phys. B 322 (1989) 167.
- [56] F. Rennecke, O(d, d)-Duality in String Theory, JHEP 1410 (2014) 69[arXiv:1404.0912[hep-th]].

- [57] K. S. Narain, New Heterotic String Theories in Uncompactified Dimensions, Phys.Lett. B 169 (1986)
 41.
- [58] N. T. Macpherson, Non-abelian T-duality, generalised geometry and holography, J. Phys.Conf. Ser. 490 (2014) 012122[arXiv:1309.1358].
- [59] M. Rocek and E. P. Verlinde, Duality, quotients, and currents, Nucl. Phys. B 373 (1992)630[hep-th/9110053].
- [60] A. M. Polyakov, Quantum Geometry of Bosonic Strings, Phys. Lett. B 103 (1981) 207.
- [61] E. Alvarez, L. Alvarez-Gaume, J. L. F. Barbon and Y. Lozano, Some global aspects ofduality in string theory, Nucl. Phys. B 415 (1994) 71[hep-th/9309039].
- [62] E. Plauschinn, T-duality revisited, JHEP 1401 (2014) 131[arXiv:1310.4194 [hep-th]].
- [63] E. Bergshoeff, C. M. Hull and T. Ortin, Duality in the type II superstring effective action, Nucl. Phys. B 451 (1995) 547[hep-th/9504081].
- [64] M. Cvetic, H. Lu, C. N. Pope and K. S. Stelle, T duality in the Green-Schwarz formalism, and the massless/massive IIA duality map, Nucl. Phys. B 573 (2000) 149[hep-th/9907202].
- [65] B. Kulik and R. Roiban, T-duality of the Green-Schwarz superstring, JHEP 0209 (2002)007[hep-th/0012010].
- [66] I. A. Bandos and B. Julia, Superfield T-duality rules, JHEP 0308 (2003) 032[hep-th/0303075].
- [67] R. Benichou, G. Policastro and J. Troost, *T-duality in Ramond-Ramond backgrounds*, Phys.Lett. B 661 (2008) 192[arXiv:0801.1785 [hep-th]].
- [68] B. Zwiebach, A first course in string theory, Cambridge, UK: Univ. Pr. (2009) 673 p
- [69] M. Gualtieri, *Generalized complex geometry*, math/0401221 [math-dg].
- [70] N. Hitchin, Generalized Calabi-Yau manifolds, Quart. J. Math. 54 (2003) 281[math/0209099 [mathdg]].
- [71] N. Hitchin, Lectures on generalized geometry, arXiv:1008.0973 [math.DG].
- [72] N. Hitchin, Brackets, forms and invariant functionals, math/0508618 [math-dg].
- [73] M. Grana, R. Minasian, M. Petrini and D. Waldram, *T-duality, Generalized Geometry andNon-Geometric Backgrounds*, JHEP 0904 (2009) 075[arXiv:0807.4527].
- [74] G. Aldazabal, D. Marques and C. Nunez, Double Field Theory: A Pedagogical Review, Class. Quant. Grav. 30 (2013) 163001[arXiv:1305.1907 [hep-th]].
- [75] D. S. Berman and D. C. Thompson, Duality Symmetric String and M-Theory, Phys. Rept.566 (2014) 1[arXiv:1306.2643 [hep-th]].
- [76] O. Hohm, D. Lst and B. Zwiebach, The Spacetime of Double Field Theory: Review, Remarks, and Outlook, Fortsch. Phys. 61 (2013) 926[arXiv:1309.2977 [hep-th]].
- [77] W. Siegel, Superspace duality in low-energy superstrings, Phys. Rev. D 48 (1993) 2826[hep-th/9305073].
- [78] W. Siegel, Two vierbein formalism for string inspired axionic gravity, Phys. Rev. D 47(1993) 5453[hep-th/9302036].
- [79] C. Hull and B. Zwiebach, *Double Field Theory*, JHEP 0909 (2009) 099[arXiv:0904.4664].

- [80] J. Lee, Introduction to Smooth Manifolds, Graduate Texts in Mathematics, Springer 218(2012) isbn: 9781441999825
- [81] M. Fukuma, T. Oota and H. Tanaka, Comments on T-dualities of Ramond-Ramond potentials on tori, Prog. Theor. Phys. 103 (2000) 425[hep-th/9907132].
- [82] E. O Colgáin, Self-duality of the D1 D5 near-horizon, JHEP, 1204 (2012) 047[arXiv:1202.3416 [hep-th]].
- [83] C. M. Hull, Timelike T duality, de Sitter space, large N gauge theories and topological field theory, JHEP 9807 (1998) 021[hep-th/9806146].
- [84] X. C. de la Ossa and F. Quevedo, Duality symmetries from nonAbelian isometries in stringtheory, Nucl. Phys. B 403 (1993) 377[hep-th/9210021].
- [85] B. E. Fridling and A. Jevicki, Dual Representations and Ultraviolet Divergences in Nonlinearσ Models, Phys. Lett. B 134 (1984) 70.
- [86] A. Giveon and M. Rocek, On non-Abelian duality, Nucl. Phys. B 421 (1994) 173[hep-th/9308154].
- [87] E. Alvarez, L. Alvarez-Gaume and Y. Lozano, On non-Abelian duality, Nucl. Phys. B 424(1994) 155[hep-th/9403155].
- [88] K. Sfetsos and D. C. Thompson, On non-abelian T-dual geometries with Ramond fluxes, Nucl. Phys. B 846 (2011) 21[arXiv:1012.1320 [hep-th]].
- [89] G. Itsios, C. Nunez, K. Sfetsos and D. C. Thompson, On Non-Abelian T-Duality and newN=1 backgrounds, Phys. Lett. B 721 (2013) 342[arXiv:1212.4840].
- [90] Y. Lozano, E. Ó Colgáin, K. Sfetsos and D. C. Thompson, Non-abelian T-duality, RamondFields and Coset Geometries, JHEP 1106 (2011) 106[arXiv:1104.5196 [hep-th]].
- [91] N. T. Macpherson, C. Nez, L. A. Pando Zayas, V. G. J. Rodgers and C. A. Whiting, *TypeIIB supergravity solutions with AdS₅ from Abelian and non-Abelian T dualities*, JHEP 1502(2015) 040[arXiv:1410.2650 [hep-th]].
- [92] Ö. Kelekci, Y. Lozano, N. T. Macpherson and E. Ó Colgáin, Supersymmetry and non-Abelian Tduality in type II supergravity, Class. Quant. Grav. 32 (2015) no.3, 035014[arXiv:1409.7406[hep-th]].
- [93] E. Caceres, N. T. Macpherson and C. Nez, New Type IIB Backgrounds and Aspects of TheirField Theory Duals, JHEP 1408 (2014) 107[arXiv:1402.3294 [hep-th]].
- [94] T. Curtright and C. K. Zachos, Currents, charges, and canonical structure of pseudodualchiral models, Phys. Rev. D 49 (1994) 5408[hep-th/9401006].
- [95] K. Sfetsos, Gauged WZW models and non-Abelian duality, Phys. Rev. D 50 (1994) 2784[hepth/9402031].
- [96] Y. Lozano, Non-Abelian duality and canonical transformations, Phys. Lett. B 355 (1995)165[hep-th/9503045].
- [97] K. Sfetsos, Non-Abelian duality, parafermions and supersymmetry, Phys. Rev. D 54 (1996)1682[hep-th/9602179].
- [98] S. F. Hassan, SO(d, d) transformations of Ramond-Ramond fields and space-time spinors, Nucl. Phys. B 583 (2000) 431[hep-th/9912236].
- [99] K. Sfetsos, K. Siampos and D. C. Thompson, Canonical pure spinor (Fermionic) T-duality, Class. Quant. Grav. 28 (2011) 055010[arXiv:1007.5142].

- [100] E. Bergshoeff, R. Kallosh, T. Ortin, D. Roest and A. Van Proeyen, New formulations of D = 10supersymmetry and D8 - O8 domain walls, Class. Quant. Grav. 18 (2001) 3359[hep-th/0103233].
- [101] H. Godazgar and M. J. Perry, Real fermionic symmetry in type II supergravity, JHEP1101 (2011) 032[arXiv:1008.3128].
- [102] A. P. Polychronakos and K. Sfetsos, *High spin limits and non-abelian T-duality*, Nucl.Phys. B 843 (2011) 344[arXiv:1008.3909].
- [103] G. Itsios, Y. Lozano, E. Ó Colgáin and K. Sfetsos, Non-Abelian T-duality and consistent truncations in type-II supergravity, JHEP 1208 (2012) 132[arXiv:1205.2274].
- [104] Y. Lozano, E. Ó Colgáin, D. Rodriguez-Gömez and K. Sfetsos, Supersymmetric AdS₆ via TDuality, Phys. Rev. Lett. 110 (2013) no.23, 231601[arXiv:1212.1043].
- [105] K. Sfetsos, Recent developments in non-Abelian T-duality in string theory, Fortsch. Phys.59 (2011) 1149[arXiv:1105.0537].
- [106] A. S. Schwarz and A. A. Tseytlin, Dilaton shift under duality and torsion of elliptic complex, Nucl. Phys. B 399 (1993) 691[hep-th/9210015].
- [107] I. Bakhmatov and D. S. Berman, Exploring Fermionic T-duality, Nucl. Phys. B 832 (2010)89[arXiv:0912.3657].
- [108] J. Simon, Brane Effective Actions, Kappa-Symmetry and Applications, Living Rev. Rel.15 (2012) 3[arXiv:1110.2422].
- [109] A. Neveu and J. H. Schwarz, Factorizable dual model of pions, Nucl. Phys. B 31 (1971)86.
- [110] P. Ramond, Dual Theory for Free Fermions, Phys. Rev. D 3 (1971) 2415.
- [111] M. B. Green and J. H. Schwarz, Supersymmetrical Dual String Theory, Nucl. Phys. B 181(1981) 502.
- [112] M. B. Green and J. H. Schwarz, Supersymmetrical Dual String Theory. 2. Vertices and Trees, Nucl. Phys. B 198 (1982) 252.
- [113] M. B. Green and J. H. Schwarz, Supersymmetrical String Theories, Phys. Lett. B 109(1982) 444.
- [114] D. Friedan, E. J. Martinec and S. H. Shenker, Conformal Invariance, Supersymmetry and String Theory, Nucl. Phys. B 271 (1986) 93.
- [115] M. B. Green and J. H. Schwarz, Properties of the Covariant Formulation of SuperstringTheories, Nucl. Phys. B 243 (1984) 285.
- [116] M. T. Grisaru, P. S. Howe, L. Mezincescu, B. Nilsson and P. K. Townsend, N = 2 Super-strings in a Supergravity Background, Phys. Lett. B 162 (1985) 116.
- [117] http://relativity.livingreviews.org/
- [118] N. Berkovits, M. Bershadsky, T. Hauer, S. Zhukov and B. Zwiebach, Superstring theoryon $AdS_2 \times S^2$ as a coset supermanifold, Nucl. Phys. B 567 (2000) 61[hep-th/9907200].
- [119] N. Berkovits, The Ten-dimensional Green-Schwarz superstring is a twisted Neveu-Schwarz-Ramond string, Nucl. Phys. B 420 (1994) 332[hep-th/9308129].
- [120] N. Berkovits, A New description of the superstring, In *Rio de Janeiro 1995, Particles andfields* 390-418[hep-th/9604123].
- [121] L. Wulff, Superisometries and integrability of superstrings, JHEP 1405 (2014)115[arXiv:1402.3122].

- [122] M. C. Abbott, J. Tarrant and J. Murugan, Fermionic T-Duality of $AdS_n \times S^n(\times S^n) \times T^m$ using IIA Supergravity, arXiv:1509.07872 [hep-th].
- [123] H. Lu, C. N. Pope and J. Rahmfeld, A Construction of Killing spinors on Sⁿ, J. Math.Phys. 40 (1999) 4518[hep-th/9805151].
- [124] I. Adam, A. Dekel and Y. Oz, On Integrable Backgrounds Self-dual under Fermionic T-duality, JHEP 0904 (2009) 120[arXiv:0902.3805 [hep-th]].
- [125] A. Dekel and Y. Oz, Self-Duality of Green-Schwarz Sigma-Models, JHEP 1103 (2011)117[arXiv:1101.0400 [hep-th]].
- [126] A. Cagnazzo, D. Sorokin, A. A. Tseytlin and L. Wulff, Semiclassical equivalence of Green-Schwarz and Pure-Spinor/Hybrid formulations of superstrings in AdS₅ × S⁵ and AdS₂ × S² × T⁶, J. Phys. A 46 (2013) 065401[arXiv:1211.1554 [hep-th]].
- [127] V. Forini, V. G. M. Puletti and O. Ohlsson Sax, The generalized cusp in AdS 4x CP 3 and more one-loop results from semiclassical strings, J. Phys. A 46 (2013)115402[arXiv:1204.3302 [hep-th]].
- [128] I. Bakhmatov, On $AdS_4 \times CP^3$ T-duality, Nucl. Phys. B 847 (2011) 38[arXiv:1011.0985].
- [129] J. Simon, T duality and effective D-brane actions, Phys. Rev. D 61 (2000) 047702[hep-th/9812095].
- [130] P. M. Cowdall and P. K. Townsend, Gauged supergravity vacua from intersecting branes, Phys. Lett. B 429 (1998) 281, Erratum: [Phys. Lett. B 434 (1998) 458][hep-th/9801165].
- [131] H. J. Boonstra, B. Peeters and K. Skenderis, Brane intersections, anti-de Sitter spacetimes and dual superconformal theories, Nucl. Phys. B 533 (1998) 127[hep-th/9803231].
- [132] G. W. Gibbons, Hyper-Kahler manifolds and multiply intersecting branes, Nucl. Phys.Proc. Suppl. 62 (1998) 422[hep-th/9707232].
- [133] E. Alvarez, L. Alvarez-Gaume and Y. Lozano, An Introduction to T-duality in stringtheory, Nucl. Phys. Proc. Suppl. 41 (1995) 1[hep-th/9410237].
- [134] A. Brandhuber, N. Itzhaki, J. Sonnenschein and S. Yankielowicz, *Baryons from supergravity*, JHEP 9807 (1998) 020[hep-th/9806158].
- [135] J. A. de Azcarraga and J. Lukierski, Supersymmetric Particles with Internal Symmetries and Central Charges, Phys. Lett. B 113 (1982) 170.
- [136] J. A. de Azcarraga and J. Lukierski, Supersymmetric Particles in N = 2 Superspace: Phase Space Variables and Hamiltonian Dynamics, Phys. Rev. D 28 (1983) 1337.
- [137] W. Siegel, Hidden Local Supersymmetry in the Supersymmetric Particle Action, Phys.Lett. B 128 (1983) 397.
- [138] R. R. Metsaev and A. A. Tseytlin, Type IIB superstring action in $AdS_5 \times S^5$ background, Nucl. Phys. B 533 (1998) 109[hep-th/9805028].
- [139] S. S. Gubser, I. R. Klebanov and A. M. Polyakov, Gauge theory correlators from noncritical string theory, Phys. Lett. B 428 (1998) 105[hep-th/9802109].
- [140] G. T. Horowitz and H. Ooguri, Spectrum of large N gauge theory from supergravity, Phys.Rev. Lett. 80 (1998) 4116[hep-th/9802116].
- [141] E. Witten, Anti-de Sitter space and holography, Adv. Theor. Math. Phys. 2 (1998) 253[hep-th/9802150].

- [142] S. Ferrara, C. Fronsdal and A. Zaffaroni, On N = 8 supergravity on AdS_5 and N = 4 superconformal Yang-Mills theory, Nucl. Phys. B 532 (1998) 153[hep-th/9802203].
- [143] I. Bena, J. Polchinski and R. Roiban, Hidden symmetries of the AdS₅ × S⁵ superstring, Phys. Rev. D 69 (2004) 046002[hep-th/0305116].
- [144] N. Berkovits, Quantization of the type II superstring in a curved six-dimensional background, Nucl. Phys. B 565 (2000) 333[hep-th/9908041].
- [145] R. Kallosh, Duality symmetric quantization of superstrings, Phys. Rev. D. 52 (1995) 6020 [hepth/9506113].
- [146] P. A. Grassi, G. Policastro and P. van Nieuwenhuizen, An Introduction to the covariant quantization of superstrings, Class. Quant. Grav. 20 (2003) S395[hep-th/0302147].
- [147] A. Salam and J. A. Strathdee, On Kaluza-Klein Theory, Annals Phys. 141 (1982) 316.
- [148] L. Frappat, P. Sorba and A. Sciarrino, Dictionary on Lie superalgebras, [hep-th/9607161].
- [149] K. Zarembo, Strings on Semisymmetric Superspaces, JHEP 1005 (2010) 002
- [150] R. Ricci, A. A. Tseytlin and M. Wolf, On T-Duality and Integrability for Strings on AdSBackgrounds, JHEP 0712 (2007) 082[arXiv:0711.0707 [hep-th]].
- [151] P. A. Grassi, D. Sorokin and L. Wulff, Simplifying superstring and D-brane actions inAdS₄ × CP^3 superbackground, JHEP 0908 (2009) 060[arXiv:0903.5407].
- [152] N. Rughoonauth, P. Sundin and L. Wulff, Near BMN dynamics of the $AdS_3 \times S^3 \times S^3 \times S^1$ superstring, JHEP 1207 (2012) 159[arXiv:1204.4742].
- [153] P. Claus, R. Kallosh, J. Kumar, P. K. Townsend and A. Van Proeyen, Conformal theory M2, D3, M5 and D1-branes + D5-branes, JHEP 9806 (1998) 004[hep-th/9801206].
- [154] L. Wulff, The type II superstring to order θ^4 , JHEP 1307 (2013) 123[arXiv:1304.6422[hep-th]].
- [155] L. Wulff, On integrability of strings on symmetric spaces, JHEP 1509 (2015)115[arXiv:1505.03525 [hep-th]].
- [156] J. P. Gauntlett, R. C. Myers and P. K. Townsend, Supersymmetry of Rotating Branes, Phys. Rev. D 59 (1998) 025001[hep-th/9809065].
- [157] J. de Boer, A. Pasquinucci and K. Skenderis, AdS/CFT dualities involving large 2-D N = 4 superconformal symmetry, Adv. Theor. Math. Phys. 3 (1999) 577[hep-th/9904073].
- [158] S. Gukov, E. Martinec, G. W. Moore and A. Strominger, The Search for a holographic dual to $AdS_3 \times S^3 \times S^3 \times S^1$, Adv. Theor. Math. Phys. 9 (2005) 435[hep-th/0403090].
- [159] E. Ivanov, S. Sidorov and F. Toppan, Superconformal mechanics in SU(2 1) superspace, Phys. Rev. D 91 (2015) no.8, 085032 [arXiv:1501.05622 [hep-th]].
- [160] N. Beisert, T-Duality, Dual Conformal Symmetry and Integrability for Strings on $AdS_5 \times S^5$, Fortsch. Phys. 57 (2009) 329[arXiv:0903.0609 [hep-th]].
- [161] C. Hull and B. Zwiebach, The Gauge algebra of double field theory and Courant brackets, JHEP 0909 (2009) 090[arXiv:0908.1792 [hep-th]].
- [162] D. S. Berman and N. B. Copland, The String partition function in Hulls doubled formalism, Phys. Lett. B 649 (2007) 325[hep-th/0701080].

- [163] J. A. Minahan and K. Zarembo, The Bethe ansatz for N = 4 superYang-Mills, JHEP 0303(2003) 013[hep-th/0212208].
- [164] C. M. Hull, A Geometry for non-geometric string backgrounds, JHEP 0510 (2005) 065[hep-th/0406102].
- [165] C. M. Hull, Global aspects of T-duality, gauged sigma models and T-folds, JHEP 0710(2007) 057[hep-th/0604178].
- [166] A. Dabholkar and C. Hull, Generalised T-duality and non-geometric backgrounds, JHEP0605 (2006) 009[hep-th/0512005].
- [167] C. M. Hull, Doubled Geometry and T-Folds, JHEP 0707 (2007) 080[hep-th/0605149].
- [168] D. S. Berman, N. B. Copland and D. C. Thompson, Background Field Equations for theDuality Symmetric String, Nucl. Phys. B 791 (2008) 175[arXiv:0708.2267 [hep-th]].
- [169] D. S. Berman and D. C. Thompson, Duality Symmetric Strings, Dilatons and O(d, d)Effective Actions, Phys. Lett. B 662 (2008) 279[arXiv:0712.1121 [hep-th]].
- [170] O. Hohm, C. Hull and B. Zwiebach, Background independent action for double field theory, JHEP 1007 (2010) 016[arXiv:1003.5027 [hep-th]].
- [171] N. Berkovits, *Explaining the Pure Spinor Formalism for the Superstring*, JHEP 01 (2008) 065[arXiv:0712.0324[hep-th]].